IMPLEMENTING SUSTAINABLE ENERGY IN COMPETITIVE ELECTRICITY MARKETS

by

Andrew Pape

B.A.Sc., University of Waterloo, 1994

RESEARCH PROJECT SUBMITTED IN PARTIAL FULFILLMENT OF THE REQUIREMENTS FOR THE DEGREE OF MASTER OF NATURAL RESOURCE MANAGEMENT

in the School of Resource and Environmental Management

Report No. 212

© Andrew Erich Pape 1997

SIMON FRASER UNIVERSITY

December, 1997

All rights reserved. This work may not be reproduced in whole or in part, by photocopy or other means without the permission of the author.



National Library of Canada

Acquisitions and Bibliographic Services

395 Wellington Street Ottawa ON K1A 0N4 Canada Bibliothèque nationale du Canada

Acquisitions et services bibliographiques

395, rue Wellington Ottawa ON K1A 0N4 Canada

Your file Votre rélérence

Our file Notre référence

The author has granted a nonexclusive licence allowing the National Library of Canada to reproduce, loan, distribute or sell copies of this thesis in microform, paper or electronic formats.

The author retains ownership of the copyright in this thesis. Neither the thesis nor substantial extracts from it may be printed or otherwise reproduced without the author's permission. L'auteur a accordé une licence non exclusive permettant à la Bibliothèque nationale du Canada de reproduire, prêter, distribuer ou vendre des copies de cette thèse sous la forme de microfiche/film, de reproduction sur papier ou sur format électronique.

L'auteur conserve la propriété du droit d'auteur qui protège cette thèse. Ni la thèse ni des extraits substantiels de celle-ci ne doivent être imprimés ou autrement reproduits sans son autorisation.

0-612-24221-8



ABSTRACT

The purpose of this study is to assess mechanisms to foster sustainable electricity generation technologies (Sustainables) under competitive electricity markets. Sustainables are those technologies which have environmentally desirable attributes when compared to conventional market-driven electricity generation technologies. The electricity sector is undergoing fundamental restructuring from direct government intervention and control to a more market-oriented approach. In particular, governments are introducing competition in electricity generation and, in some jurisdictions, retail supply, and imposing functional or corporate separation of vertically-integrated utilities to enhance competition and transparency. Some of the traditional mechanisms which governments have used to foster Sustainables may no longer be compatible, or may have to be supplemented, in this new market environment.

This study has identified and evaluated market-oriented policy mechanisms for fostering Sustainables under competitive electricity markets. These policies include: (1) a sustainable energy portfolio standard (SPS), which guarantees that a minimum share of electricity generated by Sustainables be included in the wholesale supply of electricity; (2) the collection of a nonbypassable System Benefits Charge (SBC) on electricity sales to establish a fund which subsidizes the development of Sustainables; and (3) the application of revenue-neutral environmental adders (EA) to generator bids into the electricity market which will affect their competitiveness.

The policy mechanisms were compared under several evaluative criteria in a simulation model called the Competitive Electricity Market Policy Analysis (CEMPA) model. Simulations were conducted of electricity dispatch and investment behavior in B.C. and Alberta from 1995 - 2025 assuming implementation of those mechanisms.

The results indicate that the SPS mechanism tends to maximize the installed capacity of Sustainables because it is designed to support the most cost-effective technologies or resources. The SBC mechanism, by design, enhances the diversity of Sustainables supported, and maximizes the cost reduction of emerging Sustainables. The SPS minimizes financial impacts due to a phasing-in of the requirement, and a continuous cost-minimization pressure on the market. The EA minimizes the CO₂ abatement cost because the policy mechanism is set-up as a CO₂ tax, thus driving the market to reduce emissions at the least cost. Finally, the SPS is the most simple to operate and administer because it relies on the market to achieve the policy objectives.

ACKNOWLEDGMENTS

I would like to express my appreciation to Mark Jaccard who provided so much guidance and support during the course of my research efforts. Dr. Jaccard not only trained me in the economics discipline, but also helped me to become a more balanced environmentalist. He provided me with several work opportunities that were highly inter-related with my research and have opened up career options for the future.

I would also like to thank the other committee members for their valuable comments on my research, and the members of the Energy Research Group who provided support, feedback and intellectual guidance to me over the last three years.

I would like to acknowledge all the individuals and organizations who have provided me with information that was critical for my study, and feedback on my work. The B.C. Utilities Commission, BC Hydro, The Alberta Power Pool, Willis Energy Services Ltd., Sigma Engineering, Canadian Hydro Developers, Vision Quest Wind Electric, Stothert Engineering, Westcoast Power, Inland Pacific Energy, Energy Alternatives, and Taylor Munro Energy Systems are some of those organizations.

I am also very grateful for the financial assistance of the B.C. Government through their B.C. Environmental Research Scholarship, Simon Fraser University for their Fellowships and Research Assistance funding, and the Energy Research Group for hiring me as a consultant on several research related contracts.

Finally, I would like to thank my parents for providing me with moral support over the last three years, and several friends that have been there for me.

TABLE OF CONTENTS

ABSTRACT	Ш
ACKNOWLEDGEMENTS	IV
TABLE OF CONTENTS	v
LIST OF TABLES	VП
LIST OF FIGURES AND EQUATIONS	VIII

1. INTRODUCTION	1
1.1 Background on the Study	I
1.2 Competitive Electricity Markets	3
1.3 Environmental Threats under Retail Competition	5
1.3.1 Changes in Regulation Over Investment in New Generation Technologies	5
1.3.2 Changes in Regulation Over the Operation of Generation Facilities	7
1.4 Sustainable Electricity Generation Technologies	7
1.4.1 Costs and Benefits of Sustainables	9
1.5 Traditional Mechanisms to foster Sustainables	11
1.5.1 Set-Asides or Purchase Requirements	11
1.5.2 Utility and Market Based Initiatives	11
1.6 Market Instruments to Foster Sustainables under Retail Competition	12
1.6.1 Tradable Portfolio Standard or Market-based Set-Aside	13
1.6.2 Non-Bypassable System Benefits Charge with Subsidy	15
1.6.3 Revenue-Neutral Environmental Adders on Power Pool Bids	17
1.7 Report Outline	17
2 MODELLING METHODOLOGY	10
2. In Operation Metric Contraction Supply	19
2.2 Introduction to the Competitive Market Policy Analysis Model (CEMPA)	20
2.2 Intoduction to the competitive market 1 oney Phillipsis Model (CLIM P)	20
2.5 I Load Duration Curve and Screening Curve Investment Algorithm	21
2.3.2 Variation in Investment Behavior	23
2.3.3 Development Limits for Particular Technologies	23
2.3.4 Investment Issues for Sustainables	25
2.3.5 Methodology for Modelling	26
2 4 Electricity Dispatch - Spot Market Sub-Model	33
2.4 1 Background on U.K. System	34
2.4.2 Dispatch Issues for Sustainables	36
2.4.3 Methodology for Modelling	37
2.5 Sustainability Policy Implementations	43
2.5.1 Introduction	43
2.5.2 Sustainables Portfolio Standard (SPS)	44
2.5.3 Non-Bynassable System Benefits Charge with Sustainability Fund	46
2.5.4 Revenue-Neutral Environmental Cost Adder	49
2 MODELLING INDUTS AND SCENADIOS	5-
J. MUDELLING INFUTS AND SULNAMIUS 2.1 Modelling Input Dommeter:	54
3.1 1 Discount Date	54 En
3.1.1 Discount Rate	52
2.1.2 Collinioully Files]]
S.I.J LOAU FOICEAS	23

3.1.4 Competing Electricity Generation Technologies	56
3.2 Policy Scenarios	63
4. SIMULATION RESULTS	64
4.1 British Columbia Model	65
4.2 Alberta Model	71
5. SIMULATION ANALYSIS AND EVALUATION	78
5.1 Analysis of Policy Mechanisms	79
5.1.1 Market Penetration of Sustainables	79
5.1.2 Diversity of Sustainables Developed	80
5.1.3 Reduction of Unit Costs of Sustainables	81
5.1.4 Financial Impacts - Total Expenditure on Electricity Sales and Impacts on Spot Price	82
5.1.5 Annual Carbon Dioxide Emissions and Abatement Costs	86
5.1.6 Administrative and Operational Simplicity	86
5.2 Cost Effectiveness of Policies for Fostering Sustainables	87
5.2.1 Analysis of Scenario Parameters Adopted	88
5.2.2 Market Penetration of Sustainables	89
5.2.3 Financial Impacts - Total Expenditure on Electricity Sales and Impacts on Spot Price	95
5.2.4 Annual Carbon Dioxide Emission Abatement Cost	104
5.3 Summary of Analysis	107
5.4 Analysis of Uncertainty and Accuracy of the CEMPA Model	108
5.4.1 Uncertainty Analysis	108
5.4.2 Accuracy of the Model	109
5.4.3 Discussion on Limitations of Model	110
6. CONCLUSIONS	112
6.1 Overview	112
6.2 Future Work - What are the Next Steps?	113
7. REFERENCES	114
8. APPENDIX A - SPECIFIC METHODOLOGICAL DETAILS	118
8.1 Screening Curve Model	118
9. APPENDIX B - SIMULATION RESULTS FROM THE B.C. MODEL	119
10. APPENDIX C - SIMULATION RESULTS FROM THE ALBERTA MODEL	132

LIST OF TABLES

.

Table 2.1 - Factors Applied for Declining Capital Costs of Emerging Sustainables	33
Table 2.2 - Loss of Load Probabilities	40
Table 2.3 - Allocation of Sustainability Fund among Sustainables	48
Table 2.4 - Carbon Taxes Applied for Technologies under the Environmental Adder Mechanism	51
Table 3.1 - Energy Price Forecasts (real \$1995 per Gigajoule)	54
Table 3.2 - Load and Energy Forecasts (MW and GWh)	56
Table 3.3 - Technology Parameters	59
Table 3.4 - Parameters for Existing Technologies	59
Table 3.5 - Development Limits for Technologies (MW)	60
Table 3.6 - Levelized Costs of New Technologies in B.C. (\$1995/kWh)	61
Table 3.7 - Levelized Costs of New Technologies in Alberta (\$1995/kWh)	62
Table 4.1 - Market Penetration (MW of canacity) of Technologies in 2025 - B.C.	65
Table 4.2 - Peaking and Baseload Wholesale Snot Prices of Electricity (PSP) (1995\$/MWh)	68
Table 4.3 - Cost of Electricity in the Year 2025 (Millon \$1995)	69
Table 4.4 - Market Penetration (MW of canacity) of Technologies in 2025 - Alberta	77
Table 4.5 - Peaking and Baseload Wholesale Snot Prices of Electricity (PSP) (1995(MWh)	75
Table 4.6 - Cost of Electricity in the Verr 2025 (Million \$1005)	76
Table 5.1 - Market Penetration of Sustainables (AND) in 2025	90
Table 5.2 - Canadity of New Sustainables Developed (AGID) in 2025	00 90
Table 5.2 - Capacity of New Sustainables Developed (NEW) III 2025 Table 5.3 - Deduction in the Unit Cost of Sustainables (\$1005(4:117b) - British Columbia	0U 01
Table 5.5 - Reduction in the Unit Cost of Sustainables (\$1555/KWII) - Diffish Columbia	01
Table 5.4 - Reduction in the Onit Cost of Sustainables (\$1995/KWII) - Alberta	02 00
Table 5.5 - Foncy Parameters Employed for Cost-Effectiveness Analysis	00
	07
Table 5.7 Cond Despine of Fontoine Succide Laboration (\$10050.6W of Succide Laboration)	DO 00
Table 5.7 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) -	B.C.98
Table 5.7 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Table 5.8 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) -	B.C.98
Table 5.7 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Table 5.8 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Alberta Table 5.9 - Cost Premium for Sustainables Development (\$1995/MW of Sustainables) -	B.C.98
Table 5.7 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Table 5.8 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Alberta Table 5.9 - Cost Premium for Sustainables Production (\$1995/kWh from Sustainables) - B.C.	B.C.98 98 99
Table 5.7 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Table 5.8 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Alberta Table 5.9 - Cost Premium for Sustainables Production (\$1995/kWh from Sustainables) - B.C. Table 5.10 - Cost Premium for Sustainables Production (\$1995/kWh of Sustainables) - Alberta	B.C.98 98 99 99
Table 5.7 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Table 5.8 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Alberta Table 5.9 - Cost Premium for Sustainables Production (\$1995/kWh from Sustainables) - B.C. Table 5.10 - Cost Premium for Sustainables Production (\$1995/kWh of Sustainables) - B.C. Table 5.11 - CO ₂ Emissions in 2025 - B.C. and Alberta (megatonnes)	B.C.98 98 99 99 104
Table 5.7 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Table 5.8 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Alberta Table 5.9 - Cost Premium for Sustainables Production (\$1995/kWh from Sustainables) - B.C. Table 5.10 - Cost Premium for Sustainables Production (\$1995/kWh of Sustainables) - B.C. Table 5.11 - CO ₂ Emissions in 2025 - B.C. and Alberta (megatonnes) Table 5.12 - Summary Matrix of Policy Analysis	B.C.98 98 99 99 104 107
Table 5.7 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Table 5.8 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Alberta Table 5.9 - Cost Premium for Sustainables Production (\$1995/kWh from Sustainables) - B.C. Table 5.10 - Cost Premium for Sustainables Production (\$1995/kWh of Sustainables) - B.C. Table 5.10 - Cost Premium for Sustainables Production (\$1995/kWh of Sustainables) - Alberta Table 5.11 - CO ₂ Emissions in 2025 - B.C. and Alberta (megatonnes) Table 5.12 - Summary Matrix of Policy Analysis Table B.1 - Total generation capacity (MW) by technology type - BAU	B.C.98 98 99 99 104 107 119
Table 5.7 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Table 5.8 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Alberta Table 5.9 - Cost Premium for Sustainables Production (\$1995/kWh from Sustainables) - B.C. Table 5.10 - Cost Premium for Sustainables Production (\$1995/kWh of Sustainables) - B.C. Table 5.10 - Cost Premium for Sustainables Production (\$1995/kWh of Sustainables) - Alberta Table 5.11 - CO ₂ Emissions in 2025 - B.C. and Alberta (megatonnes) Table 5.12 - Summary Matrix of Policy Analysis Table B.I - Total generation capacity (MW) by technology type - BAU Table B.2 - Total generation capacity (MW) by technology type - SPS	B.C.98 98 99 99 104 107 119 120
Table 5.7 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Table 5.8 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Alberta Table 5.9 - Cost Premium for Sustainables Production (\$1995/kWh from Sustainables) - B.C. Table 5.10 - Cost Premium for Sustainables Production (\$1995/kWh of Sustainables) - B.C. Table 5.10 - Cost Premium for Sustainables Production (\$1995/kWh of Sustainables) - Alberta Table 5.11 - CO ₂ Emissions in 2025 - B.C. and Alberta (megatonnes) Table 5.12 - Summary Matrix of Policy Analysis Table B.1 - Total generation capacity (MW) by technology type - BAU Table B.2 - Total generation capacity (MW) by technology type - SPS Table B.3 - Total generation capacity (MW) by technology type - SBC	B.C.98 98 99 104 107 119 120 121
Table 5.7 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Table 5.8 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Alberta Table 5.9 - Cost Premium for Sustainables Production (\$1995/kWh from Sustainables) - B.C. Table 5.10 - Cost Premium for Sustainables Production (\$1995/kWh of Sustainables) - B.C. Table 5.10 - Cost Premium for Sustainables Production (\$1995/kWh of Sustainables) - Alberta Table 5.11 - CO ₂ Emissions in 2025 - B.C. and Alberta (megatonnes) Table 5.12 - Summary Matrix of Policy Analysis Table B.1 - Total generation capacity (MW) by technology type - BAU Table B.2 - Total generation capacity (MW) by technology type - SPS Table B.3 - Total generation capacity (MW) by technology type - ADD	B.C.98 98 99 104 107 119 120 121 122
Table 5.7 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Table 5.8 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Alberta Table 5.9 - Cost Premium for Sustainables Production (\$1995/kWh from Sustainables) - B.C. Table 5.10 - Cost Premium for Sustainables Production (\$1995/kWh of Sustainables) - B.C. Table 5.10 - Cost Premium for Sustainables Production (\$1995/kWh of Sustainables) - Alberta Table 5.11 - CO ₂ Emissions in 2025 - B.C. and Alberta (megatonnes) Table 5.12 - Summary Matrix of Policy Analysis Table B.1 - Total generation capacity (MW) by technology type - BAU Table B.2 - Total generation capacity (MW) by technology type - SPS Table B.3 - Total generation capacity (MW) by technology type - SBC Table B.4 - Total generation capacity (MW) by technology type - ADD Table B.5 - Annual energy generation (GWh) by technology type - BAU	B.C.98 98 99 104 107 119 120 121 122 123
Table 5.7 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Table 5.8 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Alberta Table 5.9 - Cost Premium for Sustainables Production (\$1995/kWh from Sustainables) - B.C. Table 5.10 - Cost Premium for Sustainables Production (\$1995/kWh of Sustainables) - B.C. Table 5.10 - Cost Premium for Sustainables Production (\$1995/kWh of Sustainables) - Alberta Table 5.11 - CO ₂ Emissions in 2025 - B.C. and Alberta (megatonnes) Table 5.12 - Summary Matrix of Policy Analysis Table B.I - Total generation capacity (MW) by technology type - BAU Table B.2 - Total generation capacity (MW) by technology type - SPS Table B.3 - Total generation capacity (MW) by technology type - SBC Table B.4 - Total generation capacity (MW) by technology type - ADD Table B.5 - Annual energy generation (GWh) by technology type - BAU Table B.5 - Annual energy generation (GWh) by technology type - BAU	B.C.98 98 99 99 104 107 119 120 121 122 123 124
Table 5.7 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Table 5.8 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Alberta Table 5.9 - Cost Premium for Sustainables Production (\$1995/kWh from Sustainables) - B.C. Table 5.10 - Cost Premium for Sustainables Production (\$1995/kWh of Sustainables) - B.C. Table 5.10 - Cost Premium for Sustainables Production (\$1995/kWh of Sustainables) - Alberta Table 5.11 - CO ₂ Emissions in 2025 - B.C. and Alberta (megatonnes) Table 5.12 - Summary Matrix of Policy Analysis Table B.1 - Total generation capacity (MW) by technology type - BAU Table B.2 - Total generation capacity (MW) by technology type - SPS Table B.3 - Total generation capacity (MW) by technology type - SBC Table B.4 - Total generation capacity (MW) by technology type - ADD Table B.5 - Annual energy generation (GWh) by technology type - BAU Table B.5 - Annual energy generation (GWh) by technology type - EA	B.C.98 98 99 99 104 107 119 120 121 122 123 124 125
Table 5.7 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Table 5.8 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Alberta Table 5.9 - Cost Premium for Sustainables Production (\$1995/kWh from Sustainables) - B.C. Table 5.10 - Cost Premium for Sustainables Production (\$1995/kWh of Sustainables) - B.C. Table 5.10 - Cost Premium for Sustainables Production (\$1995/kWh of Sustainables) - Alberta Table 5.11 - CO ₂ Emissions in 2025 - B.C. and Alberta (megatonnes) Table 5.12 - Summary Matrix of Policy Analysis Table B.1 - Total generation capacity (MW) by technology type - BAU Table B.2 - Total generation capacity (MW) by technology type - SPS Table B.3 - Total generation capacity (MW) by technology type - SBC Table B.4 - Total generation capacity (MW) by technology type - ADD Table B.5 - Annual energy generation (GWh) by technology type - BAU Table B.5 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA	B.C.98 98 99 99 104 107 119 120 121 122 123 124 125 126
Table 5.7 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Table 5.8 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Alberta Table 5.9 - Cost Premium for Sustainables Production (\$1995/kWh from Sustainables) - B.C. Table 5.10 - Cost Premium for Sustainables Production (\$1995/kWh of Sustainables) - B.C. Table 5.11 - CO ₂ Emissions in 2025 - B.C. and Alberta (megatonnes) Table 5.12 - Summary Matrix of Policy Analysis Table B.1 - Total generation capacity (MW) by technology type - BAU Table B.2 - Total generation capacity (MW) by technology type - SPS Table B.3 - Total generation capacity (MW) by technology type - SBC Table B.4 - Total generation capacity (MW) by technology type - ADD Table B.5 - Annual energy generation (GWh) by technology type - BAU Table B.5 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA	B.C.98 98 99 99 104 107 119 120 121 122 123 124 125 126 132
Table 5.7 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Table 5.8 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Alberta Table 5.9 - Cost Premium for Sustainables Production (\$1995/kWh from Sustainables) - B.C. Table 5.10 - Cost Premium for Sustainables Production (\$1995/kWh of Sustainables) - Alberta Table 5.11 - CO ₂ Emissions in 2025 - B.C. and Alberta (megatonnes) Table 5.12 - Summary Matrix of Policy Analysis Table B.1 - Total generation capacity (MW) by technology type - BAU Table B.2 - Total generation capacity (MW) by technology type - SPS Table B.3 - Total generation capacity (MW) by technology type - SBC Table B.4 - Total generation capacity (MW) by technology type - ADD Table B.5 - Annual energy generation (GWh) by technology type - BAU Table B.5 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA	B.C.98 98 99 99 104 107 119 120 121 122 123 124 125 126 132 133
Table 5.7 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Table 5.8 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Alberta Table 5.9 - Cost Premium for Sustainables Production (\$1995/kWh from Sustainables) - B.C. Table 5.10 - Cost Premium for Sustainables Production (\$1995/kWh of Sustainables) - Alberta Table 5.11 - CO ₂ Emissions in 2025 - B.C. and Alberta (megatonnes) Table 5.12 - Summary Matrix of Policy Analysis Table B.1 - Total generation capacity (MW) by technology type - BAU Table B.2 - Total generation capacity (MW) by technology type - SPS Table B.3 - Total generation capacity (MW) by technology type - SBC Table B.4 - Total generation capacity (MW) by technology type - ADD Table B.5 - Annual energy generation (GWh) by technology type - BAU Table B.5 - Annual energy generation (GWh) by technology type - BAU Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - BAU Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - SPS Table C.1 - Total generation capacity (MW) by technology type - SPS Table C.3 - Total generation capacity (MW) by technology type - SPS	B.C.98 98 99 99 104 107 119 120 121 122 123 124 125 126 132 133 134
Table 5.7 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Table 5.8 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Alberta Table 5.9 - Cost Premium for Sustainables Production (\$1995/kWh from Sustainables) - B.C. Table 5.10 - Cost Premium for Sustainables Production (\$1995/kWh of Sustainables) - Alberta Table 5.11 - CO ₂ Emissions in 2025 - B.C. and Alberta (megatonnes) Table 5.12 - Summary Matrix of Policy Analysis Table B.1 - Total generation capacity (MW) by technology type - BAU Table B.2 - Total generation capacity (MW) by technology type - SBC Table B.3 - Total generation capacity (MW) by technology type - ADD Table B.4 - Total generation capacity (MW) by technology type - BAU Table B.5 - Annual energy generation (GWh) by technology type - BAU Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - BAU Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA Table C.1 - Total generation capacity (MW) by technology type - SPS Table C.3 - Total generation capacity (MW) by technology type - SPS	B.C.98 98 99 99 104 107 119 120 121 122 123 124 125 126 132 133 134 135
Table 5.7 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Table 5.8 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Alberta Table 5.9 - Cost Premium for Sustainables Production (\$1995/kWh from Sustainables) - B.C. Table 5.10 - Cost Premium for Sustainables Production (\$1995/kWh of Sustainables) - Alberta Table 5.11 - CO ₂ Emissions in 2025 - B.C. and Alberta (megatonnes) Table 5.12 - Summary Matrix of Policy Analysis Table B.1 - Total generation capacity (MW) by technology type - BAU Table B.2 - Total generation capacity (MW) by technology type - SBC Table B.3 - Total generation capacity (MW) by technology type - SBC Table B.4 - Total generation capacity (MW) by technology type - BAU Table B.5 - Annual energy generation (GWh) by technology type - BAU Table B.5 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA Table C.1 - Total generation capacity (MW) by technology type - EA Table C.3 - Total generation capacity (MW) by technology type - BAU Table C.4 - Total generation capacity (MW) by technology type - BAU Table C.3 - Total generation capacity (MW) by technology type - BAU Table C.4 - Total generation capacity (MW) by technology type - SBC Table C.4 - Total generation capacity (MW) by technology type - SBC Table C.4 - Total generation capacity (MW) by technology type - BAU	B.C.98 98 99 99 104 107 119 120 121 122 123 124 125 126 132 133 134 135 136
 Table 5.7 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Table 5.8 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Alberta Table 5.9 - Cost Premium for Sustainables Production (\$1995/kWh from Sustainables) - B.C. Table 5.10 - Cost Premium for Sustainables Production (\$1995/kWh of Sustainables) - Alberta Table 5.11 - CO₂ Emissions in 2025 - B.C. and Alberta (megatonnes) Table 5.12 - Summary Matrix of Policy Analysis Table B.1 - Total generation capacity (MW) by technology type - BAU Table B.2 - Total generation capacity (MW) by technology type - SPS Table B.3 - Total generation capacity (MW) by technology type - ADD Table B.5 - Annual energy generation (GWh) by technology type - BAU Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - BAU Table B.6 - Annual energy generation (GWh) by technology type - BAU Table B.6 - Annual energy generation (GWh) by technology type - BAU Table B.6 - Annual energy generation (GWh) by technology type - BAU Table B.6 - Annual energy generation (GWh) by technology type - BAU Table C.1 - Total generation capacity (MW) by technology type - BAU Table C.2 - Total generation capacity (MW) by technology type - BAU Table C.3 - Total generation capacity (MW) by technology type - BAU Table C.4 - Total generation capacity (MW) by technology type - BAU Table C.5 - Annual energy generation (GWh) by technology type - BAU Table C.5 - Annual energy generation (GWh) by technology type - BAU Table C.5 - Annual energy generation (GWh) by technology type - BAU Table C.5 - Annual energy generation (GWh) by technology type - BAU Table C.5 - Annual energy generation (GWh) by technology type - BAU 	B.C.98 98 99 99 104 107 119 120 121 122 123 124 125 126 132 133 134 135 136 137
Table 5.7 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Table 5.8 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Alberta Table 5.9 - Cost Premium for Sustainables Production (\$1995/kWh from Sustainables) - B.C. Table 5.10 - Cost Premium for Sustainables Production (\$1995/kWh of Sustainables) - Alberta Table 5.11 - CO ₂ Emissions in 2025 - B.C. and Alberta (megatonnes) Table 5.12 - Summary Matrix of Policy Analysis Table 5.12 - Summary Matrix of Policy Analysis Table B.1 - Total generation capacity (MW) by technology type - BAU Table B.2 - Total generation capacity (MW) by technology type - SPS Table B.3 - Total generation capacity (MW) by technology type - SBC Table B.4 - Total generation capacity (MW) by technology type - BAU Table B.5 - Annual energy generation (GWh) by technology type - BAU Table B.5 - Annual energy generation (GWh) by technology type - BAU Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - EA Table B.6 - Annual energy generation (GWh) by technology type - BAU Table C.1 - Total generation capacity (MW) by technology type - SPS Table C.3 - Total generation capacity (MW) by technology type - BAU Table C.4 - Total generation capacity (MW) by technology type - BAU Table C.3 - Total generation capacity (MW) by technology type - BAU Table C.4 - Total generation capacity (MW) by technology type - SPS Table C.3 - Total generation capacity (MW) by technology type - SPS Table C.4 - Total generation capacity (MW) by technology type - SPS Table C.5 - Annual energy generation (GWh) by technology type - BAU Table C.5 - Annual energy generation (GWh) by technology type - BAU Table C.5 - Annual energy generation (GWh) by technology type - BAU Table C.5 - Annual energy generation (GWh) by technology type - BAU Table C.5 - Annual energy generation (GWh) by technology type - BAU	B.C.98 98 99 99 104 107 119 120 121 122 123 124 125 126 132 133 134 135 136 137 138

LIST OF FIGURES AND EQUATIONS

Figure 1.1 - Costs and Benefits of a Renewable Portfolio Standard in the U.S.	14
Figure 1.2 - Allocation of AB1890 Renewables Funds in California from Wires Charge	16
Figure 2.1 - Annual Load Duration Curve Applied in CEMPA	23
Figure 2.2 - Overview of the Investment Sub-Model Methodology	27
Figure 2.3 - Overview of the Screening Curve Routine	28
Equation 2.1 - Levelized Cost Equation	30
Figure 2.4 - Overview of the Investment Variation and Development Limit Routines	31
Equation 2.2 - Market -Share Based Capital Cost Reduction of Emerging Sustainables	32
Figure 2.5 - System Marginal Price Calculation	38
Equation 2.3 - Hourly Average System Loss of Load Probability	39
Equation 2.4 - Pool Purchase Price of Electricity	41
Equation 2.5 - Pool Selling Price	42
Figure 2.6 - Overview Diagram of CEMPA Snot Market Sub-Model	43
Figure 2.7 - CFMPA Implementation of Sustainables Portfolio Standard	45
Figure 2.8 - CEMPA implementation of the Non-Bypassable System Benefits Charge	47
Figure 2.9 - Financial Calculations under a Revenue-Neutral Environmental Charge Mechanism	50
Figure 4.1. Generation Share by Sustainables (nervent), all scenarios	67
Figure 4.2 - Carbon Diovide Emissions (tonnes) - all scenarios	70
Figure 4.3 $_{\rm e}$ Generation Share by Sustainables (nercent), all scenarios	74
Figure 4.4. Carbon Diovide Emissions (tonnes) - all scenarios	77
Figure 5.1 - Policy Cost Premium over BAII (Million \$1905) - British Columbia	83
Figure 5.2 - Policy Cost Premium Over BAU (Million \$1995) - Diffusit Columbia	85
Figure 5.3 - B C Installed Canacity of Sustainables (AW)	91
Figure 5.4 - Alberta Installed Canacity of Sustainables (MW)	97
Figure 5.5 - B.C. Market Share of Generation from Sustainables (NFW)	03
Figure 5.6 - Alberta Market Share of Ceneration from Sustainables (percent)	04
Figure 5.7 - B.C. Bolicy Cost Branium over BAIL for each Bolicy (million \$1005)	96
Figure 5.8 - Alberta Dolicy Cost Premium over BAU for each Policy (million \$1995)	07
Figure 5.9 - Alberta Poncy Cost 1 remains over DAG for each 1 oney (minion \$1555) Figure 5.9 $_{\rm C}$ Wholesale Spot Price (\$1995/cWh)	100
Figure 5.10 - B.C. Peaking Spot Price (\$1005/kWh)	101
Figure 5.10 - D.C. Federing Spot Flice (\$1775/KWII)	101
Figure 5.11 - Alberta Wildebale Spot Files (\$1095/KWIL)	102
Figure 5.12 - CO. Emissions Abstament Cast Curve (\$1995/tanne)- British Columbia	105
Figure 5.15 - CO ₂ Emissions Abatement Cost Curve (\$1995/tonne)- Alberta	105
Figure 5.14 - CO2 Emissions Additional Cost Curve (\$1775) (onne)- Addita	110
Equation A.1 - Equation for the Optimization of Electricity Supply Resource investment	110
Figure B.1 - Investment in Sustainables (GWb) - all scenarios	127
Figure D.2 - Ceneration by Sustainables (G will) - an scenarios	140
Figure D.5 - Feak wholesale electricity rates (\$1995/kwil) - all sectiatios	127
Figure D.4 - Daseload wholesale electricity faits (31995/KWII) - all scenarios	130
Figure B.5 - Total expenditures on electricity (million \$1995) - all scenarios	131
Figure C.1 - Investment in Sustainables Capacity (MW) - all scenarios	140
rigure U.2 - Generation by Sustainables (GWD) - all scenarios	141
rigure U.3 - reak wholesale electricity rates (\$1995/kWh) - all scenarios	142
rigure U.4 - Daseload wholesale electricity rates (\$1995/KWh) - all scenarios	145
rigure U.S - 1 otal expenditures on electricity (million \$1995) - all scenarios	144

1. INTRODUCTION

The purpose of this study is to assess mechanisms to foster environmentally desirable electricity generation technologies under competitive electricity markets. This study quantitatively analyzes three energy policy mechanisms which foster those technologies under competitive markets.

1.1 Background on the Study

This section includes a description of a current energy policy issue, framed as a problem statement for the study, a description of the study purpose which is an attempt to address the policy issue, and a brief overview of the methodology undertaken to demonstrate the application of policy mechanisms to respond to the problem/issue.

Problem Statement

Sustainable electricity generation technologies, referred to as *Sustainables* throughout this study, are those technologies which have environmentally desirable attributes when compared to conventional market-driven electricity generation technologies. The term *renewables* is often used to describe these technologies because they are mostly powered by renewable resources such as flowing water, solar radiation, wind, or biomass matter.

The electricity sector is undergoing fundamental restructuring from direct government intervention and control to a more market-oriented approach. In particular, governments are introducing competition in electricity generation and, in some jurisdictions, retail supply, and imposing functional or corporate separation of vertically-integrated utilities to enhance competition and transparency.

Some of the traditional mechanisms which governments have used to foster *Sustainables* may no longer be compatible, or may have to be supplemented, in this new market environment. These mechanisms include government regulation of electricity resource acquisition by distributors or direct government or public investment in sustainable energy.

Purpose of Study

This study has identified and evaluated market-oriented policy mechanisms for fostering Sustainables under competitive electricity markets. Several evaluative criteria were applied for comparing the policy mechanisms, including cost, market development, and environmental criteria.

Methodology of Study

The hypothesis of the study is that environmentally desirable electricity generation technologies (*Sustainables*) can be effectively fostered in competitive electricity markets. The hypothesis is demonstrated and tested through the following steps:

- Simulations were conducted of electricity dispatch and investment behavior in B.C. and Alberta from 1995 - 2025 assuming implementation of different policy mechanisms for fostering Sustainables.
- For the purposes of the study, *Sustainables* were defined as: wind generation systems, solar photovoltaic cells, small run-of-river hydroelectric and tidal generators, biomass and wastefuel thermal generators, and biomass cogeneration systems.
- Three market-oriented policy mechanisms were tested, including a tradable Sustainables portfolio standard or market purchase requirement, a wire charge with subsidized support for Sustainables, and the establishment of a revenue-neutral environmental cost adder which gives Sustainables a competitive advantage over less environmentally desirable resources.
- The policy mechanisms were compared according to the following evaluative criteria:
 - 1. Market share of Sustainables (MW capacity and GWh production).
 - 2. Degree of diversity of Sustainables (number of technologies fostered).
 - 3. Impact on the reduction of the unit cost of Sustainables (\$/kWh).
 - 4. Total expenditure on electricity production (\$), and wholesale spot prices (\$/GWh).
 - 5. Annual carbon dioxide emission abatement cost (\$ / tonne CO₂ reduced).
 - 6. Administrative and operational simplicity.

The first criterion was selected to demonstrate the achievement of the primary objective of the study, which is to analyze policy mechanisms to foster Sustainables. The capacity development and energy production are the most obvious ways of demonstrating that. The second and third criteria illustrate specific impacts of policies on the sustainable energy industry itself, the diversity of technologies developed, and the cost reduction achieved through economies of scale and manufacture. Technological diversity has a benefit of minimizing resource risk - so that if one resource (i.e. natural gas) is short in supply or high in price, then other resources will have a sufficient market share to partially replace thier loads. With sufficient cost reduction, the industry may be able to compete without any preferential policies

driving it. The fourth criterion demonstrates the impacts of the sustainability policies on the financial health of the electricity sector, both in terms of price impacts (spot price) and the total cost of operating the electrical system. The fifth is a measure of environmental impacts of the electricity and the financial performance of various policies to meet an environmental objective $(CO_2 \text{ reduction})$. The sixth is a qualitative assessment of the relative administrative burden of the various policies.

1.2 Competitive Electricity Markets

1

2

The electricity industry in Canada is facing a transformation that will foster greater competition for electricity supply and potentially modify the structure of current regulated vertically-integrated utilities. This transformation is often called *electricity market reform* or *electricity market restructuring*. Policymakers around the world are investigating or undertaking electricity market reform initiatives.

One restructuring outcome is the adoption of a *wholesale competition* market which maintains regulated, natural monopoly control over the transmission and distribution of power and retail sales to consumers, but the supply of electricity is opened to competition. The short-term wholesale price of electricity is determined like a commodity, according to the supply and wholesale demand for electricity in a spot market¹, rather than through regulatory mechanisms. The toll for regulated transmission and distribution (T&D) services is determined through rate hearings, encompassing the cost of providing those services and allowing the T&D utility a fair rate of return. In some cases, T&D tolls vary by the location of the generator and/or consumer, to provide an incentive for generators to locate in areas with high marginal transmission or distribution costs. Also, the cost of ancillary services² is often bundled in T&D tolls, although in some jurisdictions a separate competitive ancillary services market has been created.

Beyond wholesale competition, retail competition allows end-use consumers to contract directly with producers or marketer intermediaries for the electricity commodity, which is delivered by a

A dynamic commodity market structure whereby the "spot price" of electricity is determined at regular intervals based on available supply and instantaneous demand. The "spot price" is set from the highest bid of the selected least-cost generators required to meet demand for that period.

Transmission support services necessary to ensure integrity of the transmission network, including voltage control, reactive power, load following, loss compensation, energy imbalance, scheduling and dispatch services, and system protection (operating reserves).

regulated natural monopoly through T&D lines. In some jurisdictions, those customers who turn to alternative suppliers may have to pay an "exit fee" to pay for their share of any *stranded costs* in generation facilities, those costs which result from the devaluation of utility generation assets as the market is opened to competition.

The main objective of fostering *wholesale competition* in the electricity generation market is to maximize the economic efficiency of investment and operation decisions so that electricity is provided at the least cost to consumers. With *retail competition*, a further objective is to provide greater choice to consumers in the types of pricing and services they receive. Additional objectives for both models may include streamlining or eliminating redundant regulatory mechanisms and diversifying the electricity sector by involving more players, such as independent power producers on the supply side and power marketers and brokers on the retail side.

Several factors are contributing to the momentum of electricity market restructuring in British Columbia and elsewhere (BCUC, 1995).

- U.S. trade requirements which dictate market access reciprocity ~ if Canadian producers want market access into the U.S., then U.S. producers have to have the same rights in Canada.
- Technological change and limitations of large-scale centralized generation technologies have contributed to the erosion of the rationale for maintaining vertically-integrated utilities, which control the generation, transmission and distribution of electricity within one organization.
- Increased global competition is forcing industrial electricity consumers to cut costs to the level that their competitors pay, hence pushing for a North-America wide competitive market.
- Public policy is emphasizing economic efficiency more than before also favoring free markets and private ownership.

Alberta was the first jurisdiction in North America to restructure its electricity market by establishing a mandatory *power pool* in January, 1996, and appointing an independent *transmission administrator* (TA) in late-1997. The former is an independent, non-profit entity which coordinates all electricity trading in the province; and determines the short-term market price of electricity. An Irish company has been selected as the TA, as it has no financial interest in the Alberta electricity market as a buyer or seller. The role of the TA is to determine how transmission facilities are operated and developed. Eventually, when pre-restructuring contracts between

generators and purchasers are phased-out, a full wholesale competition market will develop and the power pool will act as a spot market for short-term electricity trading in the province.

A small number of jurisdictions have adopted the *retail competition* model, including the United Kingdom, New Zealand, Chile, and parts of Australia. Several U.S. states will adopt a *retail competition* market model in 1998. It is expected that many other jurisdictions will adopt this model over the next decade. The *retail competition* market model is the one that is considered in this study.

1.3 Environmental Threats under Retail Competition

The development and utilization of electricity generation technologies and resources can have a significant impact on environmental quality. The investment behavior of electricity generators affects environmental quality through their choice of energy resources. The mode of operation of existing generation facilities can also enhance or reduce the environmental impacts of the sector.

Existing environmental regulations or regulatory mechanisms which foster Sustainables, or cause generators to operate their facilities in a manner that is environmentally desirable, may be affected by electricity market reforms, as outlined in the following sections.

1.3.1 Changes in Regulation Over Investment in New Generation Technologies

In certain jurisdictions with regulated electricity markets, generators have been required by regulators or government to include environmental considerations in their investment decisions through such mechanisms as Integrated Resource Planning (IRP) and Social Costing. IRP is a utility planning process which requires the consideration of all known resources for meeting the demand for a utility's product, including those which focus on traditional supply sources, and those which focus on the conservation and management of demand, referred to as Demand-Side Management (DSM) (BCUC, 1993). In some cases, regulators have required generators to quantify non-monetary environmental and social impacts of electricity generation through Social Costing policies, attaching a monetary or index value to impacts such as air emissions, job creation potential, or changes in land-use. Both IRP and Social Costing have potential environmental benefits through the promotion of DSM which reduces the consumption of electricity or delays the construction of new capacity plants, or by fostering new environmentally desirable electricity

generation technologies that would not have been competitive without the inclusion of environmental considerations.

These types of mechanisms require regulatory oversight by governments or utility commissions over the sales of electricity from generators to consumers as has been the case in many electricity markets composed of vertically-integrated utilities. Under electricity market reforms which establish *wholesale competition*, whereby the electricity generation sector is deregulated, regulatory oversight is still maintained, as resource acquisition decisions of electricity distribution companies are scrutinized. However, under market reforms which establish *retail competition*, whereby a consumer has the ability to contract directly with a generator for power acquisition under a deregulated market, the ability of the regulator to require IRP or Social Costing ceases (BCUC, 1995). Thus, in order to maintain the inclusion of environmental considerations in resource acquisition decisions of generators under *retail competition*, current IRP or Social Costing mechanisms designed for regulated monopoly markets will need to be replaced by other policy measures which foster Sustainables.

The reform of electricity markets towards greater competition in North America is expected to increase investment in natural gas generating technologies, because of their low capital costs, forecasted low resource costs, and short development lead-times relative to other technologies (Jaccard, 1996). Natural gas combustion has several environmental impacts including greenhouse gas emissions, emissions of nitrous oxides and volatile organic compounds which can contribute to the creation of ground level ozone (smog), emissions of particulate matter which affects human health, and land and water impacts associated with the extraction and transportation of natural gas.

The impact of increased natural gas utilization on the environment can be positive if it is replacing coal-thermal facilities as they retire, resulting in smaller levels of greenhouse gas and smog causing emissions. However, in those jurisdictions that currently generate a large proportion of their electricity from non carbon based resources such as hydroelectricity, increased natural gas consumption can exacerbate environmental impacts, particularly with respect to air quality and greenhouse gases emissions.

In summary, under *retail competition* electricity markets, electricity regulators are unable to require generators to incorporate environmental considerations into their technology investment decisions, resulting in potential negative environmental impacts, particularly in those jurisdictions

where a large proportion of the existing electricity supply is hydro based, and where natural gas technologies are gaining market share through new investments in supply.

1.3.2 Changes in Regulation Over the Operation of Generation Facilities

Electric utilities are currently subject to environmental regulation. The operation of any power plant in British Columbia requires an air emission permit as set out under the provincial environmental assessment process and the Waste Management Act, in addition to any municipal requirements. The Burrard Thermal (natural gas) plant in Vancouver has strict guidelines to follow with respect to local air quality impacting pollutants. These requirements will not change under competitive markets.

However, greenhouse gas emissions are currently not regulated, and they are directly related to the operation of fossil fuel facilities. Greenhouse gas emissions from existing facilities are expected to increase in some jurisdictions due to the introduction of competition into the market. For example, excess capacity will be exported on the open market throughout North America, and those competitive facilities that currently operate at less than their full capacity due to limitations in domestic demand will eventually find new markets for that power, resulting in increased utilization. This argument is particularly persuasive with respect to coal facilities, which have low operating costs, and are currently operating at less than full capacity in many U.S. jurisdictions, and in Alberta and Ontario. The 1991 average capacity factor for coal facilities in the U.S. was 44% (NARUC, 1995), and under competitive markets it is expected that their utilization could almost double, depending on other environmental constraints.

1.4 Sustainable Electricity Generation Technologies

Sustainable electricity generation technologies (Sustainables) have environmentally-desirable characteristics when compared to conventional market-driven electricity generation technologies. The focalpoint of this research project is the assess the costs and benefits of fostering Sustainables in *retail competition* electricity markets. It is assumed that Sustainables are not sufficiently cost-effective to be competitive with conventional technologies (i.e. combined cycle natural gas turbines), and as such an explicit policy mechanism is required to facilitate their market penetration.

The definition of what constitutes Sustainables is a value-based judgment, although certain technologies clearly fit into that category and others clearly do not. The definition also changes over time as different environmental and social priorities dictate the bounds of the definition. Current environmental priorities in B.C. and elsewhere point towards those technologies which have low net greenhouse gas emissions, minimal locally-impacting air emissions, minor impacts on watersheds or landscapes, and no contribution to toxic waste buildup. Those technologies are generally based on renewable resources, and are relatively small in scale.

The definition of Sustainables used throughout this study assumes those attributes. The specific technologies which fit into those bounds include: wind generation systems (including wind farms), solar thermal collectors and photovoltaic cells, small run-of-river hydroelectric and tidal generators, biomass and waste-fuel thermal generators, and waste fuel cogeneration systems.

Biomass technologies produce greenhouse gas and other emissions, but are assumed to be connected with a forestry or agricultural operation which grows an equivalent amount of biomass resources as it consumes, hence mitigating the CO_2 emissions. Also, it is assumed that the location of those facilities is outside of urban areas due to the nature of the resource, such that local air quality impacting emissions such as particulates, NO_X , and volatile organic compounds (VOCs) do not have a significant environmental impact on communities³.

Waste fuel cogeneration systems have minimal net CO_2 emissions because the heat rate⁴ of producing both steam and electricity from a cogenerator is similar to the heat rate of producing steam from a boiler (for water, space, or process heating), hence providing electricity with a minimal increment of fuel consumption. As a result, the conversion of an existing boiler in a commercial or industrial facility is environmentally desirable.

3

4

Although the Lower Mainland region (Greater Vancouver) has a substantial agricultural resource, local air quality considerations would preclude the development of new biomass plants.

The amount of fuel required to produce a GJ of heat, or a kWh of electricity, indicative of the first-law efficiency of energy conversion from chemical energy to thermal, electrical or mechanical energy.

1.4.1 Costs and Benefits of Sustainables

The private cost of generating electricity from Sustainables is typically higher than from conventional technologies (i.e. natural gas turbines), despite the fact that their social cost⁵ may be lower. Sustainables are capital intensive, with relatively low operation, maintenance, and fuel costs.

However, Sustainables offer several environmental, technical and economic benefits which have often been ignored within competitive electricity markets despite their recognition by energy policymakers and regulators.

Sustainables are environmentally benign, a characteristic which may become significant in the future if environmental legislation requires energy companies to pay for environmental externalities⁶ through taxes, emissions caps or marketable emissions permits. Although the perceived threat of future environmental regulations is causing some electricity companies to develop Sustainables capacity, the majority still do not recognize their environmental benefits.

Conventional energy systems are generally based on large-scale centralized technologies characterized by long development lead times. The small scale of Sustainables also makes the time required from design to operation short, so that needed improvements can be identified by field testing and quickly incorporated into modified designs. Sustainables can advance at a faster pace than conventional technologies, and it is easier to apply modern manufacturing techniques that facilitate capital cost reduction. The dynamics of Sustainables development will have more in common with the rapid technological process and sharp price reductions that are characteristic of microprocessor-based technologies, pharmaceuticals, than with the experience for conventional energy technologies (Williams, 1993). The fact that many of these technologies are relatively new means that the potential for capital cost reduction is still great relative to many conventional technologies which have already exhausted many cost reduction opportunities.

⁵

Social cost is defined as the private financial cost plus the monetized value of environmental and social externalities.

Impacts that occur whenever a third party receives, benefits or bears costs arising from an economic transaction in which she or he is not a direct participant (McGuigan and Moyer, 1989).
 An example would be the effects of pollution from a coal plant on area residents.

Sustainables are increasingly marketed as technologies which have low financial risk because they are based on abundant renewable resources, resulting in low or zero fuel costs, and not subject to global price volatility. In contrast, a large proportion of the cost of gas turbines is based directly on fuel prices, and resource price volatility substantially affects the operating costs of those technologies. An important caveat to this argument is that volatility in interest rates will affect capital intensive technologies such as Sustainables greater than gas turbines.

Finally, Sustainables can increase the reliability of electrical systems, leading to technical and financial benefits. Some of these benefits have been recognized by utilities to a limited extent, although they are often not sufficient to offset the high capital costs of Sustainables. Many of these benefits also apply for other distributed technologies.

- Distributed Sustainables can help to provide power to a region that has been cut-off from the main grid in time of failure (due to weather, vandalism, etc.), giving transmission and distribution (T&D) control systems more options to route power through distributed generation paths (Ibid.).
- Distributed Sustainables have cost advantages over centralized technologies due to avoided T&D capital and maintenance costs. A Pacific Gas & Electric company study priced the value of distributed technologies at \$89US per kilowatt-year of avoided T&D development (Ibid.).
- Supply of electricity from Sustainables is sometimes correlated with peaks in customer electricity demand at certain times of the day or year. For example, PV generation peaks at a time when air conditioning loads are greatest. This simplifies system control functions, and can help to reduce the marginal costs of meeting increasing capacity requirements by "shaving" peaks in electricity demand.
- The excess capacity needed to maintain a given level of electricity supply reliability declines as size of individual units declines. A system made-up of plants which have a capacity of 100MW or less, require only an 11% safety margin capacity to maintain system reliability, while one with 1000MW plants needs a margin of 25% 33% (Johannson et.al., 1993). However, this argument has become partly outmoded because of increased electricity trading between utilities, which inherently gives more options to system controllers during system failure periods.

1.5 Traditional Mechanisms to foster Sustainables

This section describes several traditional mechanisms to support Sustainables.

1.5.1 Set-Asides or Purchase Requirements

In some jurisdictions, the government has provided explicit support for Sustainables through legislation that requires electricity distributors to purchase Sustainables' generated electricity.

The Alberta Small Power Research and Development Act (1988) established a renewable setaside of up to 125MW of capacity through the promotion of a variety of technologies, including: wind, biomass, small-hydro, each with a capacity of less than 2.5MW. The Act required that investor-owned utilities purchase this electricity at a guaranteed rate of 5.2 cents / kWh, and recover those revenues from their ratepayers. The initiative resulted in the development of 22MW of wind capacity, the only significant wind generation facilities in Canada today, in addition to several small-hydro and biomass plants.

In recent years, Germany has been a very active supporter of wind power and other Sustainables with its 1990 *Electricity Feed Law* (Stomeinspeisungsgesetz) which demands that any electricity generated from wind, solar, hydro, waste fuels or biomass be bought by the public electricity utility at fixed rates equivalent to 80% (90% for wind and solar) of utility rates (Groscurth, 1996). This mechanism is designed to foster the development of Sustainables without seriously impacting on consumer rates, although utilities in one State have been allowed to increase rates by up to 1% as a result of this law. Germany recently surpassed the U.S. for total installed wind energy capacity at 1,675 MW (Wind Power Monthly, 1997).

1.5.2 Utility and Market Based Initiatives

A variety of electric utility and market-based initiatives have supported Sustainables.

In some jurisdictions, government owned electric utilities have undertaken initiatives to support Sustainables, leveraging funds from ratepayers to cover any cost premium required. For example, Hydro Québec is planning to develop up to 150MW of wind capacity over the next decade under this type of arrangement. A more common approach has been to initiate green marketing initiatives where a portion of the customer base voluntarily pays a cost premium to sustain a small amount Sustainables development and operation.

Reverse metering⁷ is the practice of using a single meter to measure the difference between the total generation and total consumption of electricity by customers with on-site generation facilities by allowing the meter to run backwards. Reverse metering can increase the economic value of investing in Sustainables. It allows the customers to use the utility grid to "bank" self-generated electricity for later consumption. By using the existing meter, the utility is in essence buying Sustainables-generated electricity at full retail rates. Several utilities also buy power in excess of the total consumption of the customer in a billing period at the utility avoided cost. Currently 17 U.S. States have reverse metering programs and Toronto Hydro has a small trial program. These initiatives have not had many subscribers to date, possibly due to the high capital costs of many Sustainables.

Other mechanisms include tax rebates, such as those established under the U.S. Energy Policy Act of 1992 for wind power, or accelerated capital cost allowances for Sustainables, as was recently implemented in the Canadian taxation system, including the special CCA Class 34.1, and the Canadian Renewable Energy and Conservation Expenses (CRCE) write-off category. Advocates of sustainable energy in Canada have welcomed these recent changes in the Canadian taxation laws, but have cautioned policymakers that tax incentives are not sufficient to foster additional investment in the majority of Sustainables, as current electricity rates are too low to make sufficient profits to gain the taxation benefits.

1.6 Market Instruments to Foster Sustainables under Retail Competition

Some of the policy mechanisms presented in the previous section are only suitable under monopolistic electricity markets, particularly the ones that entail government intervention in the electricity market. They would not be appropriate for application under competitive markets, as traditional monopolies will be broken and government intervention in the electricity generation sector will be kept to a minimum to avoid stifling the free market. The green marketing, reverse metering and tax rebate mechanisms are compatible with competitive electricity markets, but are

⁷

Sometimes referred to as Net Metering or Net Billing.

not expected to have a significant impact on increasing the market share of Sustainables. On green marketing programs, Nancy Rader, a renewable energy advocate in the U.S., stated that, "... utility-sponsored programs have not been very successful, supporting very small installations that would not sustain existing renewable energy producers, even if repeated by many utilities" (Rader, 1996). As such, this study excludes an analysis of green marketing initiatives because they will not satisfy the policy evaluative criterion of "enhancing the market share of Sustainables".

Several policy mechanisms have been proposed and implemented which provide substantial support for Sustainables under competitive electricity markets. Three of these mechanisms form the foundation of the analysis for this study. They are presented below.

1.6.1 Tradable Portfolio Standard or Market-based Set-Aside

A Renewables Portfolio Standard (RPS), advocated in several jurisdictions in the U.S., requires that every retail power supplier acquire renewable energy *credits* equivalent to some percentage of its total annual energy sales (Rader et.al., 1996). The *credits* are created for kWhs of electricity generated from renewable energy resources or Sustainables. Power retailers could generate Sustainables power through their own facilities or purchase it from separate companies to then resell. The government or regulator must pre-determine the level of portfolio requirement in the market based on a variety of economic, environmental and social criteria. The RPS mechanism is ideally suited for competitive market conditions, as it is sufficiently flexible to enable investment in the most cost-effective Sustainables. If the mechanism is applied across several jurisdictions and trading of *credits* is enabled, then further cost-effectiveness gains may be possible.

The U.S. Congressman Schaefer's proposed "Consumers Power to Choose Act" of 1996 would establish a national Renewables Portfolio Standard of 4% of all GWh sold by the year 2010, excluding large hydro. A recent Tellus Institute study (Bernow et.al., 1997) estimates the electricity rate impact of the Schaefer proposal to be about 0.03 ¢/kWh in 2010, with an increase in generation from Sustainables of 56 TWh/yr, and resultant CO_2 emissions reductions of about 9 million tonnes/year. The data are outlined in Figure 1.1.

Two U.S. Senate bills, "The Electric Consumers' Protection Act of 1997" (Sen. Bumper), and the "Electric System Public Benefits Protection Act of 1997" (Sen. Jeffords), include larger portfolio standards (the latter has a requirement of 20% by 2020), and the former includes large hydro as an eligible source to meet the requirement.

Figure 1.1 - Costs and Benefits of a Renewable Portfolio Standard in the U.S.

(Source: Bernow et.al., Quantifying the Impacts of a National, Tradable Renewables Portfolio Standard. In The Electricity Journal, May, 1997, pp.42-52)

	2000		2005		2010	
	Base Case	RPS Case	Base Case	RPS Case	Base Case	RPS C
Renewable Generation (TWh)	62	64	72	102	87	144
Renewable Percentage of Total Generation	1.9 %	2.0 %	2.1 %	3.0 %	2.4 %	4.0 %
Additional Renewable Generation from MRGR (TWh)	N/A	2	N/A	30	N/A	56
Change in Electricity Price (1995 c/kWh)	N/A	0.00	N/A	0.01	N/A	0.03
Credit Trading Price (1995 c/kWh)	N/A	1.03	N/A	1.20	N/A	1.03
Carbon Emissions Saved from Electric Sector (MMT)	N/A	0	N/A	3	N/A	9



Figure 1: Breakdown of Total Non-Hydro Renewables with and without the RPS

1.6.2 Non-Bypassable System Benefits Charge with Subsidy

Another mechanism for fostering Sustainables under competitive markets is to subsidize their costs through the collection of a non-bypassable *wires charge* on all sales of electricity in the jurisdiction, also referred to as a "System Benefits Charge". The *wires charge* is collected over a period of time to build a pool of funds used to support the development of Sustainables. The funds could be distributed as direct subsidies to existing or new Sustainables facilities on a per-kWh basis (contracts-for-differences) or through capital grants via a competitive bidding system for new facilities.

The State of California, in its restructuring bill (AB1890), directs retailers to collect funds through wires charges for a variety of "public purpose programs", including a minimum of \$540 million between 1998 and 2001 for Sustainables. The *wires charge* for energy efficiency, renewable energy development, and low-income services is estimated to be about 3% of revenues for investor owned utilities. The allocation of the \$540 million for Sustainables was determined by the California Energy Commission through a public consultation process, resulting in a decision in March, 1997 to distribute 45% of the fund among existing Sustainables developed under the PURPA program, 30% to new technologies through production credits, 15% for customer rebates for green power purchased and public education programs, and 10% for "emerging technologies". See Figure 1.2 for a more detailed description of the allocation of AB1890 sustainable energy funds in California.

A similar proposal has been made in the Final Report of the "Comprehensive Review of the Northwest Energy System" calling for a 3% non-bypassable *wires charge* to facilitate the development of cost-effective conservation and appropriate renewable resource options, and sustaining appropriate low-income energy services.

Figure 1.2 - Allocation of AB1890 Renewables Funds in California from Wires Charge

(Source: California Energy Commission. Policy Report on AB1890 Renewables Funding -

Report to the Legislature. March, 1997. pg.13)



In the U.K., the restructuring of the electricity sector in 1990 included a minimum 1.3% wires charge on the retail rate under the Non-Fossil Fuel Obligation (NFFO) to subsidize Sustainables (Wind Power Monthly, 1997). A series of five "Renewables Orders" have been made by the government whereby the local distribution companies signed fixed-term contracts with competitive Sustainables' companies to provide electricity from a variety of technologies, including wind power facilities, small hydro, municipal wastes and biomass. The program has a development target of 1500MW of Sustainables capacity by the year 2000, and has contributed to the development of a robust Sustainables industry that was virtually non-existent prior to restructuring. In addition, Sustainables producers in the U.K. are planning to establish a "green power pool" to market their non-NFFO power.

1.6.3 Revenue-Neutral Environmental Adders on Power Pool Bids

The Environmental Adder (EA) mechanism affects the dispatch order of technologies as directed by the electricity system operator, by adding a charge to the bid/offer price of generation facilities into the market that is reflective of environmental impacts. Environmentally-undesirable technologies will be at a competitive disadvantage, and the mechanism may lower their dispatch ranking⁸ in the market. The mechanism favors Sustainables by potentially improving their dispatch ranking and by increasing the spot price of electricity. In order to minimize consumer rate impacts of the mechanism, it could be designed to be "revenue-neutral", meaning the funds equivalent to the total value of the EAs are transferred to consumers through a reduction in the wholesale price of electricity.

The actual value of the EA is determined by a policymaker or regulator, reflective of the environmental impacts of the operation of a particular technology.

1.7 Report Outline

This chapter has outlined the study objectives and methodology, provided background information on competitive electricity markets, potential environmental implications of adopting competitive structures, opportunities and constraints for sustainable electricity generation technologies

The dispatch ranking is used to determine which technologies will operate within a given time period (usually one hour), with the least cost resources operating first, and gradually more expensive technologies being dispatched as the cheaper ones reach capacity.

(Sustainables), and policy mechanisms to foster Sustainables. Chapter two provides detail on the methodology for the quantitative analysis (modelling). Chapter three outlines the modelling inputs and information on the scenarios. Chapter four outlines the modelling results, and Chapter five provides an overview analysis and evaluation of the policy alternatives tested within the modelling framework. Finally, Chapter six concludes on the effectiveness of the study and on the preferred policy alternatives for application in British Columbia and Alberta.

All financial figures in this report are in real 1995 dollars.

2. MODELLING METHODOLOGY

This chapter outlines the methodology for the development and execution of the simulation model which is used to meet the research objective outlined in Chapter 1. Section 2.1 provides an overview of simulation models which have been used for electricity market forecasting in the U.S., and compares them with the approach taken in this study. Section 2.2 introduces the model used for this study. Section 2.3 provides detail on the technology investment sub-model. Section 2.4 outlines the technology operation sub-model. Finally, Section 2.5 outlines the methodology for simulating three sustainability policy mechanisms.

Assumptions associated with each sub-model are outlined in this chapter, along with a description of parameters that are specific to the actual model. Chapter 4 outlines the assumptions and parameters that are exogenous to the model (provided as inputs): that is fuel prices, energy demand forecasts, discount rates, and details on competing technologies. Many of these assumptions have implications on the analysis of the modelling results provided in Chapter 6.

2.1 Overview of Simulation Models for Electricity Supply

Simulation modelling has been used as a tool by electricity planners and policymakers to forecast future market conditions and predict outcomes of policy reforms, allowing them to make informed policy decisions based on a variety of technical, economic, social or environmental considerations. A variety of simulation models have been developed for electricity markets, including specific "resource planning models" that are used by electric utilities. For example, BC Hydro uses a "Stochastic Resource Planning Model" (Drannan and Spafford, 1987) which incorporates uncertainty on a variety of levels in order to plan on generation expansion, DSM, and optimization of in-service dates for new projects. A probabilistic model of this type is useful for representing a variety of hydro reservoir streamflow conditions and variations in demand and fuel prices.

The United States Department of Energy, through its Energy Information Administration, has developed an electricity market model (EMM) for its National Energy Modelling System (NEMS), incorporating a demand forecast sub-model, an electricity capacity planning sub-module, a electricity fuel dispatch sub-model, and finally an electricity finance and pricing sub-model. The electricity capacity planning sub-module determines how best to meet expected growth in electricity demand, given available resources, expected load curves, demands, fuel prices, environmental constraints, and costs for utility and non-utility technologies (EIA, 1994). This type of modelling approach has been adapted into the technology "investment" methodology described in Section 2.3.

Simulation models of the dispatch of generation facilities and pricing of power are typically based on either optimizing the operation of existing generation facilities based on minimum system cost criteria, or optimizing the power flow through transmission lines, based on technical and financial criteria. The former are called "Unit Commitment Models" which decide which generation units to turn on and off based on least system cost criteria, and when to do it, considering significant time lags in start-up and shut-down and future values of electricity (Kahn et. al., 1996). The latter are called "Optimal Power Flow" models which solve for the least cost set of power injections on a transmission system, given a fixed demand and network parameters. Both of these types of models are computationally complex, and require a significant amount of data, far beyond the scope of this study.

In contrast, "Single Area Production Cost" (SAPC) models treat all facilities of a certain type as having the same financial characteristics, and neglect explicit representation of the electricity network. If they treat unit commitment, it is through a heuristic approach, rather than an explicit optimization. SAPC models are useful for analyzing economic and policy problems over long time frames, and are computation simple. This type of model has been applied for this study. A shortcoming of SAPC models which rely on the *load duration curve* model of demand (as in this study - Section 2.3.1) is that significant generation facility operational economies and constraints are not represented, and regional representation is lacking. One response to this has been to adapt the SAPC model to include transmission network limits, which has important implication on electricity trade, transmission congestion costs, and potential market power. However, this detailed level of analysis is beyond the scope of this study.

2.2 Introduction to the Competitive Market Policy Analysis Model (CEMPA)

The Competitive Electricity Market Policy Analysis model (CEMPA) used in this study was designed to simulate a competitive electricity market in Alberta and B.C., assuming that both jurisdictions have adopted a U.K. style *retail competition* market. The ultimate objective of CEMPA is to assess the potential for fostering sustainable electricity generation technologies (Sustainables) under *retail competition* electricity markets, and to identify the benefits and

shortcomings of alternative market policies which achieve this. These policies include: (1) a sustainable energy portfolio standard, which guarantees that a minimum share of electricity generated by Sustainables be included in the wholesale supply of electricity; (2) the collection of a non-bypassable *wires charge* on electricity sales to establish a fund which subsidizes the development of Sustainables; and (3) the application of revenue-neutral environmental adders to generator bids into the *power pool* which will affect their dispatch rank order.

CEMPA assumes that all power is dispatched through a *power pool*, a hypothetical point of delivery where all electricity trades occur in a jurisdiction, and the short-term market price of electricity (spot price) is adjusted according to the instantaneous electricity supply and demand. Under a *retail competition* market, consumers can contract with any power supplier or marketer to guarantee a long-term electricity price, but they receive all their power through the *power pool*. Formally, purchasers and suppliers enter into *contracts for differences*, which provide a financial hedge against short-term variations in the spot electricity price from the *power pool*. Consumers can enter into such contracts directly with producers, or deal with a broker, marketer, or local distribution company (existing utility).

The CEMPA model is applied separately to the jurisdictions of Alberta and British Columbia, both markets which currently have some form of competition proposed or adopted, but neither of which have adopted full *retail competition*. The simulation period is from 1995 to 2025.

CEMPA is implemented in several spreadsheets with all of the market parameters and data included, and a series of Visual Basic macros that implement the market structure and policies associated with it. CEMPA is comprised of several sub-models, an *Investment Sub-Model* which determines the annual investment in generating capacity in each of those jurisdictions, a *Spot Market Sub-Model* which determines the wholesale price of electricity and the dispatch / operation of each technology, and three *Sustainability Policy Sub-Models* which simulate each of the previously mentioned sustainable energy policies on the market. The latter sub-models are actually applications of the first two, but with additional features (macros and parameters) which mimic the policies being simulated.

2.3 - Investment Sub-Model

Investment in new generating technologies occurs on an annual basis in CEMPA, following standard principles of economic supply and demand. New generation facilities are developed when

the capacity of all existing generating units, including a reserve margin, falls below the expected peak capacity demand for the year. The selection of new generating technologies is based on the annualized capital and other fixed costs, operating and maintenance costs, and incremental fuel costs associated with that technology over its expected life.

2.3.1 Load Duration Curve and Screening Curve Investment Algorithm

An annual *load-duration curve* (LDC) is utilized by CEMPA to represent the expected electricity demand, as illustrated in Figure 2.1, showing the number of operating hours required during the year (x-axis) for each proportion of peak capacity demand (y-axis). The peak capacity changes each year in CEMPA, but the shape of the LDC does not. The area under the LDC curve represents the annual energy demand (MWh).

The LDC data were provided from BC Hydro, and it is assumed that those data hold in non-BC Hydro service areas and in Alberta. This type of data was not available from Alberta due to confidentiality constraints.

The time segments on the far left portion of the LDC curve exemplify *peak* demand periods of the year, typically corresponding to winter evenings in B.C. and Alberta, while those on the right half represent *off-peak* (or *baseload*) periods such as summer nights, with *shoulder* periods in between. A *baseload* electricity plant provides power on a continuous basis, but cannot significantly modify its power output on short notice. Hence, it is best suited to the *off-peak* demand segments. A *peaking* plant provides additional electricity most economically for those hours associated with *peak* demand hours on the left side of the LDC.

The screening curve method of selecting generation technologies for meeting new capacity demand minimizes the total cost of generation in the system by optimizing generation resources according to the characteristics of the LDC. Technologies which have low operating costs, and often higher fixed costs, are most appropriate for meeting *baseload* capacity, while technologies with low capital costs, but higher operating and fuel costs, are appropriate for meeting *peak* demand. The CEMPA investment sub-model utilizes the screening curve method to select the least capital cost option for the *peak* operating hours, and successively higher capital cost technologies with lower operational costs for shoulder and baseload conditions. This process is detailed in Section 2.3.5.





Source: BC Hydro

2.3.2 Variation in Investment Behavior

The sub-model allows for a pre-determined amount of investment in *baseload* generating technologies which are slightly more expensive than the least-cost technology, provided their *levelized-costs*⁹ are within a threshold cost premium over the cheapest technology. This is intended to mimic the variation in the installed costs of the technologies due to geographic or technical constraints, and the variation in investment behavior that occurs due to other factors. Specific factors which could contribute to this variation in investment include the following.

9

The cost per kWh of electricity from a technology, including operating and amortized capital costs.

- companies selecting electricity purchases based on an established relationship with a supplier that may not offer the cheapest resource.
- specific needs at a consumer site where an IPP project is being developed, or preference for or against specific technologies (e.g. woodwaste cogeneration facilities at forestry operations).
- preference towards technologies which have transmission or distribution benefits to the utility (e.g. small scale distributed technologies which help to avoid transmission bottlenecks).
- development of generation technologies at sites that are not integrated on the utility network, and where the least cost technology is not available (remote locations with no gas).
- geographically caused variation in operation expenses (e.g. different gas prices, or variation in reliability associated with climatic conditions).

An assumption is made that 20% of the new capacity developed each year will be met by those technologies that are within the specified price threshold. This market share is allocated equally among those technologies to keep the modelling simple. The 20% market share figure was selected to be representative of the combined effects of all the factors above that would induce an investor to purchase technologies that on-average are more expensive than the cheapest technology.

CEMPA assumes that the variation in investment occurs for those technologies which have costs no greater than 20% over the *levelized-cost* of the least-cost technology. This figure is representative of the willingness-to-pay for the criteria listed above, some of which have financial implications, and others which do not. The ISTUM modelling framework utilized at SFU applies a similar investment variation rule, whereby industrial end-use technologies (e.g. boilers) that have a life-cycle cost within 15% of the least cost technology will take up to 20% of the market share (ISTUM manual, 1996). Non-financial criteria are treated separately in ISTUM through minimum or maximum market shares for certain technologies. Based on this information, and assuming that non-price criteria are worth up to a 5% financial cost premium, it was decided to select a 20% price threshold in CEMPA as the basis for variation in investment behavior.

2.3.3 Development Limits for Particular Technologies

CEMPA applies a development limit for certain technologies, given that the resources required for their operation may be limited (e.g. wood waste), or due to political or social conditions that dictate a limitation on their development (e.g. large hydro developments). The development limits applied are specified in Chapter 3.

2.3.4 Investment Issues for Sustainables

Sustainable electricity generation technologies (Sustainables) are treated equally with other technologies in the CEMPA investment model. The economic characteristics of these technologies dictate that they be utilized for *baseload* operation, due to their high capital cost and low operating costs (no fuel costs).

2.3.4.1 Dispatchability

CEMPA applies development limits to certain Sustainables, reflective of their resource limitations within the jurisdiction, and for the purpose of maintaining the power quality of the electric grid. The latter objective is achieved by maintaining a minimum proportion of *dispatchable* technologies on the system grid. Those technologies are available for operation on command (short notice), allowing the grid operator to track variations in electricity demand. In contrast, *non-dispatchable* generation technologies, such as wind, tidal, and solar, are limited in their ability to produce electricity on short notice. They are powered by renewable resources which are intermittent, typically dependent on weather conditions. Research indicates that in order to maintain electricity system integrity, the total development of *non-dispatchable* technologies should not exceed 30-50% (Johansson et.al., 1993) of the total electrical system peak capacity. If several types of *non-dispatchable* technologies are developed, and there is diversity in both resource type and geographic location, then their total market share could be brought up to the 50% level without affecting power quality. In contrast, if the development occurs in one location with a single resource, then the maximum market share of that class of technology should be reduced to the 30% level.

2.3.4.2 Economies of Manufacture and Capital Cost Reduction of Emerging Technologies

CEMPA incorporates expected reductions in the capital cost of certain emerging Sustainables. Capital cost reduction results from the achievement of economies of manufacture in the factory production and innovations in the design of the technology. The capital cost of a new technology in a competitive marketplace often decreases as the sales of the technology increase because industry realizes efficiencies through increased experience with the technology and with mass production. Eventually, the cost of the technology stabilizes, due to a stasis in technological innovation, similar to what has happened over the last century with steam turbines and hydraulic generators.

Robert Williams applies an "experience curve" in his assessment of the cost reduction potential of photovoltaic solar technologies. His thesis is that, ".. per unit production costs decline as a direct, estimable proportion of cumulative production" (Williams, 1993). This type of approach is adopted for certain Sustainables in CEMPA, including solar technologies, wind generators, tidal generators and fuel cells.

An assumption is made that the cumulative investment in Sustainables in B.C. or Alberta reflects investment patterns across North America, and cost reductions occur in the model that are reflective of a market-wide cost reduction. For example, if 10MW of photovoltaic capacity is developed in B.C., it is assumed that a proportionate amount of capacity is developed across North America, resulting in approximately 100 times the B.C. capacity. Thus, the anticipated capital cost reduction associated with 1000MW of cumulative capacity investment across North America is applied into the model.

The capital cost of these emerging technologies is determined by an exponential formula which is a function of the starting cost of the technology in year one of the simulation, its cumulative capacity in the jurisdiction, and a specific exponent which is reflective of the rate of cost reduction and the specified limit of the cost reduction. This formula is calibrated against predicted cost reductions specified in the BC Hydro *Integrated Electricity Plan*, in Appendix E, *Resource Options* (1995). Specific details on the formula are presented in Section 2.3.5.4.

Also, CEMPA simulates some autonomous capital cost reduction as a function of time, not dependent on the cumulative capacity in the jurisdiction. It is assumed that investment in other jurisdictions will support this, as well as innovations in research and development in laboratories and universities which is not a function of market share.

2.3.5 Methodology for Modelling

The CEMPA investment sub-model determines the investment in new generation technologies on an annual basis. Investment is determined in megawatts of dependable capacity. The sub-model methodology is illustrated in Figure 2.2. First, the *Load Duration Curve* (LDC) is assembled from the forecast demand for electricity and the daily load profile of consumers. Second, CEMPA calculates the requirement for new capacity to meet that demand and determines the economically optimal mix of technologies based on the characteristics of the LDC. Third, the mix is converted to megawatts of installed new capacity. Fourth, the investment model allocates a small amount of capacity to investment in technologies which are more expensive than the most competitive *baseload* technology, but within a threshold price premium, intended to mimic variation in investment behavior. Finally, the model determines if the capacity development limit has been exceeded for any technologies, and adjusts their investment level accordingly.





2.3.5.1 Load Duration Curve and Screening Curve

10

The Load Duration Curve (LDC), as outlined in Figure 2.1, is used to determine investment in new generation technologies. The Screening Curve routine selects the technologies which are least cost for meeting the LDC. See Figure 2.3 for an illustration. The total annual cost¹⁰ for each

The annualized capital cost, as well as the accumulation of all variable costs is represented.

technology is represented on the y-axis of the *screening curve*, and the range of annual operating hours along the x-axis (0 to 8760 hours). Each technology is represented by a line on the graph. The least cost technologies for all operating hours of the year are selected as those that minimize the total annual cost, switching technologies at line intersection points. In the example illustrated below, all three technologies are selected for a segment of the market, because all of them are the least cost technology for a particular operating hour range. However, in the CEMPA model with up to 20 competing technologies, not all are selected, as several technologies have significantly higher capital costs.

The screening curve routine is executed for every year of the simulation (30 years). The specific model employed is illustrated in Appendix A.1. The fixed and variable costs of technologies change over the course of the simulation with the shape of the screening curve changing as well. The capital costs of certain emerging technologies declines over time as explained previously. In other investment routines, CEMPA further accelerates the decline of capital costs of technologies when their market penetration increases (See Sections 2.3.4.2 and 2.3.5.4 for details).

The result of this routine is an indication of the number of operating hours of each new technology for each year of the simulation. With the example in Figure 2.3, the first technology would operate for 360 hours of the year, corresponding with *peak* hours only (e.g. simple cycle gas), the second for 2800 (e.g. combined cycle gas), and the third for a full 8760 hours (e.g. hydroelectricity or cogeneration), the latter being *baseload*.

Figure 2.3 - Overview of the Screening Curve Routine




2.3.5.2 Installed Capacity for Each new Technology

The CEMPA investment sub-model, in a separate routine, next allocates a specific capacity to each technology that has been selected by the *screening curve*, based on their number of operating hours. It utilizes the *load duration curve* to determine the specific capacities associated with each operating hour. For example, if the three technologies selected in the *screening curve* in Figure 2.3 were applied to the LDC on Figure 2.1, and the 100% mark on the LDC was 1000MW, then about 150MW of technology 1 would be developed (corresponding with *peak* loads), 150MW of technology 2 (*shoulder*), and about 700MW of technology 3 (*baseload*).

It is assumed that the new capacity will come on line every year with no lead times required. In other words, if a certain technology is required in a particular year, CEMPA assumes the construction of that facility would have commenced a number of years earlier, as required.

2.3.5.3 Calculation of Levelized Cost, Other CEMPA investment functions

The CEMPA investment sub-model next simulates the variation in investment behavior for reasons specified in Section 2.3.2. It applies only to *baseload* technologies. The key parameter for this routine is the *levelized-cost* function, which is the expected average cost of electricity production from a technology per kWh when all investment and operating costs are included. This *levelized-cost* calculation only applies to *baseload* technologies because it assumes full operation of the technology for the entire year, unlike *peaking* plants which operate a lot less than their capability.

In calculating the *levelized-cost*, it is assumed that the financing conditions for all technologies is identical, that the discount rate is the same, and that the amortization period is 25 years. Under regulated utility markets (such as B.C.), projects are financed on terms linked to the anticipated life of the project, with debt guarantees in place from provincial governments or purchasing contracts. Under competitive markets, financing terms for new generation facilities will vary according to the policies of the investor, although discount rates will probably increase as a result of a higher cost of capital and reduced risk tolerance compared with regulated markets (see Section 3.1.1). It is assumed there that the higher cost of capital will lead generators to seek a financing life of 25 years or more in order to minimize their debt payments.

CEMPA calculates the *levelized-cost* for each technology for each year of the simulation period. This is the cost in real 1995\$ of building and operating a technology as if it were built in that year and the operating, maintenance and fuel costs over a 25 year period are included. Those technologies with longer than 25 year lives are treated in a similar fashion, as the costs/benefits associated with those years beyond the 25th are sufficiently small when discounted to 1995 dollars. The equation used for *levelized cost* is outlined below. The *capacity factor* is specified exogenously, equivalent to the proportion of year that a technology is operating at its maximum output, assuming that it is operating as a *baseload* technology.

Equation 2.1 - Levelized Cost Equation

Levelized Cost_{perx} =
$$\frac{Annuity of [CapCost_{yerx}]}{8760 * CapacityFactor} + Annuity of [NPV(O& M + Fuel)]$$

units = $\frac{\frac{1}{MW} \cdot year}{hours/year} + \frac{1}{MWh} \Rightarrow \frac{cents}{kWh}$

CapCost = capital cost for year X of the simulation Capacity Factor (η) = the proportion of kWH produced per kW capacity Annuity of [...] = an annual payment for an investment NPV(O& M + Fuel) = Net Present Value of operating and maintenance costs

In order to determine variations in market investment behavior (Figure 2.4), CEMPA first identifies the least cost technology for each simulation year. It then assesses technologies to determine if their *levelized costs* are within the price "threshold" which investors would accept - a 20% *levelized-cost* premium over the least cost technology. The proportion of the market share of new developments that is met by those technologies is pre-set at 20%, and is allocated equally among them, adjusting other technologies' market share to avoid overcapacity. The result is a greater diversity in new technology developments than would be the case if only the strict *screening curve* method is followed.

CEMPA also limits the capacity development of technologies according to their resource limitations or other reasons. Specific details of those development limits is listed in Chapter 4.

Figure 2.4 - Overview of the Investment Variation and Development Limit Routines



2.3.5.4 Reduction of the Capital Costs of Emerging Technologies

As outlined in Section 2.3.4.2, CEMPA simulates declining capital costs of emerging technologies according to a time-based function and as a function of cumulative market penetration. There are two levels of implementation of this function in CEMPA, with the latter applied by both the routine that determines the variation in investment behavior in the market (Section 2.3.5.3) and the Sustainability sub-model mechanisms (Section 2.5).

 A time-based declining capital cost function is applied into the screening curve function, as well as into the calculation of *levelized-cost*. This time-based function calculates the annual capital cost reduction based on a linear function that bridges the current *levelized costs* with the BC Hydro estimate of *levelized costs* in the year 2015 (BCH IEP, Appendix E). Those BC Hydro estimates assume a certain market penetration of the technology, and hence declining capital costs resulting from economies of manufacture.

The specific cost reduction formula applied in CEMPA is based on BC Hydro forecasts of unit production cost reduction for emerging technologies (BCH IEP, 1995) and assumes that about half of the cost reduction occurs due to innovations through technology R&D. The specific reductions applied are listed in the first column of Table 2.1

2. A declining capital cost function that is based on the market penetration of Sustainables, applied into the calculation of the *levelized-cost* only (utilized by several CEMPA routines).

The specific cost reduction formula is based on a simple exponential function outlined in Equation 2.2. The capital cost at year zero is the starting point of the equation (assuming zero market share in the first year). These values were given in the BC Hydro IEP (Appendix E). The cumulative market share is the key variable driving the capital cost reduction. That variable is raised to the power of a negative exponent that scales the impact on capital cost reduction according to the extent of the market penetration of the technology. This causes steep capital cost reductions in early phases of the technological implementation, and gradually less impact as the share is proportionately increased.

The key parameters for this methodology are listed in Table 2.1. The capital costs in year zero are listed in the second column of Table 2.1 The exponent, listed in the third column of Table 2.1, was determined through a calibration process that utilized a target capital cost, and a market penetration level that would result in that target capital cost being reached. The target capital costs listed in the fourth column of the table, were determined based on the figures in the BC Hydro IEP. The required market penetration levels also made use of the BC Hydro source, but scaled down by a factor of 100 to convert those North America market penetration levels to a scale appropriate to B.C. or Alberta.

Equation 2.2 - Market -Share Based Capital Cost Reduction of Emerging Sustainables

 $Cost = Cost_{year o} \bullet (CMS)^{-E}$

 $Cost_{year 0}$ = the capital cost in \$/kW at the start of the simulation (listed in Table 2.1)

CMS = cumulative market share

E = exponent (listed in Table 2.1)

	Time Based Annual Reduction	Market Share Based Reduction Year zero capital cost (real \$1995)	Exponent value	Target Capital Cost (real \$1995)	Required M Share to Re Target Reduction I	
Wind Generators	0.7% / year	\$557 / kW	0.09	\$368 / kW	100MW	
Fuel Cell	0.8% / year	\$ 672 / kW	0.11	\$405 / kW	100 MW	
Cogenerators						
Solar PV	1. 7% / year	\$2621 / kW	0.46	\$909 / kW	10 MW	
Solar Thermal -	1.4% / year	\$2 036 / kW	0.23	\$828 / kW	50 MW	
Parabolic Trough						
Tidal Power	1.4 % / year	\$ 850 / kW	0.19	\$354 / kW	100 MW	

Table 2.1 - Factors Applied for Declining Capital Costs of Emerging Sustainables

2.4 Electricity Dispatch - Spot Market Sub-Model

The purpose of the CEMPA spot market sub-model is to illustrate the patterns of technology operation in the market, to forecast the wholesale price of electricity over the 30 year simulation period, and to keep track of the revenue that generators receive through market transactions.

The business functions within a competitive electricity market are typically coordinated through an unlimited-access *power pool*, which is a clearinghouse for all power trading, and the location of an electricity spot market. The spot price of electricity varies according to the instantaneous electricity supply and demand, typically peaking during periods of high demand or due to electricity supply shortages or transmission overloading periods. The spot market is closely tied to a *forward market* for electricity which is described in the next section. Also, many buyers and

sellers establish long-term contracts to guarantee the price of electricity for their transactions and hedge against uncertainty in the spot market. A *contract for differences* between two parties typically includes an agreed-upon price for electricity (the *strike price*) and a provision that the buyer or seller of power compensates the other party for the difference between the contracted price and the hourly spot price of electricity.

The CEMPA model mimics the *power pool* structure that is established in the United Kingdom. Although the Alberta electricity market also has a *power pool* structure, it is more difficult to model because of the existence of both demand and supply-side bids, and a variety of *forward market* options.

This section begins by describing the electricity market structure in the United Kingdom, then discusses some of the issues associated with running Sustainables under a *power pool* structure, and finally outlines the methodology that is followed by CEMPA in simulating the dispatch and business aspects of the competitive electricity market in Alberta and a hypothetical competitive market in British Columbia.

2.4.1 Background on U.K. System

In the U.K., three parallel markets operate for trading electricity - the forward market, options market and spot market. Only the forward and spot markets are relevant for the CEMPA model. The price of electricity that is published in newspapers for half-hour intervals of every day is based on the day-ahead forward market, while the price that is actually paid to generators is the spot price.

The U.K. market is one of the few in the world with *retail access*, meaning that some customers can contract directly with outside utilities or electricity marketers, and their local distribution utility is required to wheel/transmit power to them. Currently, only industrial customers (100kW+ transmission customers) have *retail access*, but in 1998 the whole market will be opened up.

2.4.1.1 Forward Market - Generation Supply Bidding

In the U.K., electricity generators make offers to the *Electricity Pool* (synonymous with *power pool*) one day in advance through a forward market, indicating their ability to produce power for each half-hour of the next day and the price they would be willing to produce it for. The *Electricity Pool* authority selects the least-cost electricity generation units sufficient to meet the

forecast demand for each period, and establishes a U-schedule or "unconstrained schedule" for the dispatch of those units (Hunt & Shuttleworth, 1996). The most expensive unit that has been selected to run for each half-hour of the next day (marginal unit) sets the system marginal price (SMP) for the market. In the U.K., the SMP in peak periods has exceeded that in off-peak periods by as much as 20 times over the past six years (Hamrin, 1994).

The SMP is reflective of the marginal cost of generation and does not encompass the amortized capital costs of generating technologies, but rather only the variable operating, maintenance and fuel costs. The variable costs of operating a generation facility are directly proportional to the energy output of the facility (kWh), while the fixed or capital costs of the facility are often proportional to its power capacity (kW). As such, the SMP rewards generators for the energy they produce, but not for their production capacity. In the U.K., a capacity element is included in the price paid to generators, the pool purchase price (PPP), allowing utilities to recover some of their fixed costs. The PPP is a function of the SMP, the loss-of-load probability (LOLP), and the value of lost load (VOLL). The LOLP is defined as, "the probability that demand will exceed capacity" (Ibid.), which would occur on failure of a large generation technology which has been scheduled to operate. Alternatively, this could occur when the demand for electricity suddenly increases, but the system reserve margin is designed to protect against short-term changes in demand. The VOLL is reflective of customers' willingness to pay for un-interruptible or firm electricity. In the U.K., the average VOLL has been about £2,500/MWh (Dunn & Rossi, 1996), equivalent to about \$5/kWh. The LOLP / VOLL component of the PPP is reflective of the reliability of a unit to provide firm capacity into the system. The PPP has risen to £1,000/MWh during periods when the reserve margin on the system is low (Ibid.), giving generators a premium for reliable power production. The average PPP in the 1995/96 year was equivalent to \$0.0477/kWh (Canadian), consisting of an average SMP of \$0.0388/kWh and a capacity adder of \$0.009/kWh (U.K. Stats).

2.4.1.2 Spot Market and Uplift Fees

The spot market is characterized by immediate delivery of the product, with the price varying to equalize supply and demand at half-hour intervals. The spot market is also used by generators and purchasers to cancel or modify their contractual arrangements - if a generator isn't able to produce what it offered in the forward market due to physical or economic constraints, then it has to buy back an amount equivalent to the missing component through the spot market at the spot price of electricity which fluctuates throughout the day.

The pool selling price (PSP) that wholesale customers or retailers pay for electricity bought through the *electricity pool* is equivalent to the PPP plus a component called *uplift* which is the difference between the *U-schedule* price of electricity and the spot price during the day, reflecting physical constraints on the system and changes in demand (Scott, 1996). The *uplift* includes the transmission related costs that the *electric pool* administrator has to incur to maintain system integrity such as, "reactive power, frequency response and reserve generation capacity" (Ibid.). *Uplift* amounts have doubled between 1991 and 1995, particularly due to transmission constraints, but an initiative in 1994 worked towards reducing those expenses by giving the transmission company an incentive to improve transmission outage management, to encourage competition from the demand side for reserve services and to install new transmission equipment (Ibid.). The average PSP in 1995/96 was \$0.0512/kWh, including an average *uplift* of \$0.004/kWh (U.K. Stats).

2.4.2 Dispatch Issues for Sustainables

Typically, a Sustainables facility will bid into the market at a zero or near-zero price, as the marginal cost of operating such facilities is virtually zero. The renewable resource driving the facility is a free good (wind, sun, water), and many of the technologies have low operating and maintenance expenses. This is a significant advantage for Sustainables under a competitive market because they are always going to be dispatched when they are able to produce power.

However, *non-dispatchable* Sustainables could suffer a significant market disadvantage under competitive electricity markets if the *power pool* authority required them to provide firm electricity for the full period in which they bid into the market (half-hour). They are only able to provide electricity when the resource is available, unless they contract with other *dispatchable* generation facilities to back them up which will affect their cost-effectiveness. In practice, electricity supply into a spot market frequently fluctuates, and the *power pool* authority is charged with the task of providing backup for those fluctuations. Dispatchable facilities, with an ability to load follow or provide backup, will earn a return that reflects the value they provide to the market.

In Alberta, the *power pool* authority recognizes non-dispatchable Sustainables as a unique resource type, and grants them a *standing bid* which means that they don't have to make financial offers into the *power pool* on a daily basis as other facilities do. When Sustainables facilities are producing power, they are automatically included in the *power pool* and the power is utilized.

For the purposes of modelling Sustainables in the CEMPA power pool, it is assumed that nondispatchable Sustainables will not be penalized if they don't produce.

2.4.3 Methodology for Modelling

The purpose of the CEMPA spot-market sub-model is to: (1) compute the market price of electricity for each year of the simulation and under a variety of market conditions; (2) simulate the patterns of technology dispatch, including which technologies operate and for how many hours; and finally (3) calculate the GWh's produced and revenue received by generators under each simulation condition and for each year.

2.4.3.1 Energy Demand

The CEMPA spot-market sub-model simulates the market outcome under 24 different levels of energy demand within a year, each representative of a <u>single hour</u> of electricity system operation. The energy demands are specified by 24 different positions along the annual load duration curve (see Figure 2.1). The LDC indicates the time duration for various levels of capacity demand (GW), but this can easily be converted to energy consumption (GWh) by multiplying the capacity by the number of hours required. For example, according to the LDC in Figure 2.1, if the total system peak capacity demand is 10GW, then the one hour energy demand would range from 10 GWh under peak operating hours to about 4 GWh for the baseload hours.

By dividing the LDC into 24 segments, determining the one hour energy demand for each of those segments, multiplying each of those demands by 365 hours (equivalent to about 15.2 days of the year), and finally summing those values together, an estimate of annual energy demand can be determined. In fact, the area under the LDC times the annual peak capacity, is equal to the annual system energy requirement, and CEMPA computes that area by breaking the LDC into 24 rectangular segments with the top left corner of each rectangle intersecting the LDC.

2.4.3.2 System Marginal Price (SMP) of Electricity

CEMPA calculates the electricity System Marginal Price (SMP) of electricity for each of the 24 segments along the annual load duration curve, and for each year of the simulation, amounting to a total of 720 different SMPs for the entire simulation. The SMP, provided in units of dollars per GWh, is defined as the operating cost of the most expensive technology among all those required to

meet the electricity demand for one hour. In periods of peak demand, the SMP is higher than in periods of average demand because peaking technologies typically have higher operating, maintenance and fuel costs than baseload technologies.

Figure 2.5 illustrates how the SMP is calculated in CEMPA: four technologies are required to operate within that hour, and the SMP is equivalent to the variable cost (in \$/GWh) of technology D. In that hour, technology E is not required and will not operate.





CEMPA calculates the SMP by sorting the variable costs of the new and existing technologies operating in the jurisdiction, and then dispatching them in order of their cost (from least to most expensive) until the demand for the simulation period is met. Up to 23 technologies are included in the competition, including hydro, oil, natural gas, coal, and biomass for existing technologies, and up to 18 new technologies (see Chapter 4).

2.4.3.3 The Price Received by Generators (Pool Purchase Price - PPP)

The model assumes that a capacity value is added to the SMP for the price paid to generators, similar to that in the U.K. Pool Purchase Price (PPP) (Section 2.4.1.1). The capacity adder is a function of the Loss of Load Probability (LOLP) of the system which is a measure of the rate of unplanned system failure, and the Value of Lost Load (VOLL) or the expected losses by consumers is the power were to be cut without notice. The combination of LOLP and VOLL results in a capacity adder that reflects the importance of firm or reliable capacity on the system - when the LOLP is high, then firm capacity should be valued greater than when the LOLP is low.

The capacity adder has varied in the U.K. between £0.0001/kWh and £0.0105/kWh between 1990 and 1996, translating to about \$0.0002/kWh to \$0.021/kWh in Canadian currency.

Loss of Load Probability (LOLP)

Accurate determination of the system LOLP would require a specific knowledge of the number of generation units operating within an hour, the probability of failure for each one, and the availability of *spinning reserve* on the system which provides immediate backup. The LOLP would be equivalent to the probability that total electricity plant failures would exceed the capacity of the *spinning reserve*, and that could occur by any combination of individual plant failures. For example, if 10 equal-sized plants were operating, and the failure of any 2 of those plants would cause the system capacity to fall below demand, then the LOLP would be equal to the maximum of the resulting probability of multiplying individual plant LOLPs in groups of 2 units. In the U.K., the LOLP has varied between 0.002% and 0.21% over the past 5 years (U.K. Stats, 1996).

CEMPA applies a simplified routine for determining the hourly LOLP, making the assumption that the capacity value added to the SMP is reflective of the probability of <u>any one plant</u> failing within one hour of the simulation year. If the failure is longer than an hour, then the *power pool* authority can arrange to purchase electricity from other generators with only one hour notice. It is assumed that a sudden increase in demand is met by the *spinning reserve*, but a failure in supply <u>of any</u> <u>magnitude</u> will be accounted for in the value of the LOLP.

Each hour, the LOLP is calculated by calculating an energy-weighted average LOLP from all the generating technologies required within that hour. CEMPA utilizes the following formula:

Equation 2.3 - Hourly Average System Loss of Load Probability

Average
$$LOLP = \frac{\sum_{\text{all technologies}} LOLP \bullet Energy Supplied by each Tech.}{Total Hourly Energy Supply}$$

The Canadian Electrical Association, in their annual Canadian electricity market report (1995), publishes data on the "number of forced outages" of electricity generation units defined as, "... the occurrence of a component failure or other condition which requires that the generating unit be removed from service immediately or up to and including the very next weekend". (CEA, 1995). CEMPA utilizes those numbers, divides them by the number of units included in those statistics, and then divides by 8760 hours/year, because only sub-one hour failures are important for the spot market. The resulting LOLPs for generating units is provided below:

Table 2.2 - Loss of Load Probabilities

Technology Name	LOLP
Combustion Turbine Units	0.0444%
Fossil Generating Units	0.1002%
Hydraulic Generating Units	0.0393%
Wind Generators	1%
Solar Converters	1%
Tidal Generators	0.0393%
Fuel Cells	0.01%

The numbers for the final four technologies are estimates. It is assumed that the LOLP of tidal generators is the same as hydraulic turbines, as tide tables can be read in advance, and bidding behavior will reflect that. Fuel Cells are very reliable due to few moving parts, so a 0.01% LOLP is applied as an estimate.

Solar and wind conversion units have high LOLPs because of the inherent variation in their energy resources - the sun and wind. On an annual basis, solar collectors produce an equivalent amount of energy as if their peak generation capacity was being reached 20% of the time (higher in some regions), while wind generators are at the 30-40% level (BC Hydro, 1995 and NRCan, 1996). Solar technologies only produce power during the daytime, and their output varies according to the amount of cloud cover as well. Wind generators are even less predictable. The major constraint for including large quantities of solar or wind energy in the *power pool* is that sudden changes in weather could cause them to lose a significant portion of their production capacity, and possibly put the entire electricity system at risk of falling below the required electricity supply. The LOLP should represent that risk, but determining an accurate LOLP would require specific site information and installation sizes, neither of which is easily included in the CEMPA spot-market model. The LOLP should reflect the variation of solar collectors or wind generators within one hour, which is much less than the annual variation in output. The assumed LOLP value for solar

and wind technologies was set at 1% through a sensitivity analysis of spot electricity price impacts resulting from raising their market shares to the development limits.

Value of Lost Load (VOLL)

The VOLL is a value selected by the *Power Pool* authority which represents the cost of an electricity brownout or blackout to customers. This value is somewhat arbitrary in that it is not based on detailed empirical analysis. It is in the interest of the *Power Pool* authority to ensure that reliable power is always provided, however a market value for lost load is necessary for calculating the capacity value which contributes to the PPP. CEMPA applies the VOLL that is typically applied in the U.K., valued at £2,000/MWh in 1991 (Newbery, 1995) or about \$4.3/kWh in 1995 dollars.

Pool Purchase Price (PPP)

Equation 2.4 illustrates the formula that is applied by CEMPA for determining the electricity *pool purchase price* (PPP), the price that generators receive for the electricity they actually produce, including an adder for the value of firm capacity.

Equation 2.4 - Pool Purchase Price of Electricity

 $PPP = [(1 - LOLP) \times SMP] + [LOLP \times VOLL]$

LOLP = Average(technology LOLPs) SMP = Offer price of the marginal operating unit VOLL = \$4 / kWh (1991 dollars)

The revenue that electricity generators receive on a short-term basis is equivalent to the product of the PPP and the amount of energy they actually produce and sell into the spot market. Some generators operate exclusively in the spot market, receiving only the PPP, and do not sign any long-term agreements with buyers for the electricity they produce. However, the majority of generators will sign long-term contracts with electricity purchasers - retailers, distribution utilities, or brokers - to provide some security for their investments in generation technologies, and to hedge the volatile spot price of electricity (PPP).

2.4.3.4 The Wholesale Cost of Electricity (Pool Selling Price - PSP)

In the U.K. and Alberta, there are separate forward markets which determine a day-ahead forecast SMP based on the forecast demand, but the actual SMP and PPP (in the U.K.) are based on the instantaneous supply and demand both of which vary. In the U.K., the difference between the day-ahead forecast SMP and the actual SMP contributes to the *uplift* charge which also includes the costs of operating the *power pool* and transmission facilities (Dunn, 1996). CEMPA also applies an *uplift* to determine the *pool selling price* (PSP), which is the price that wholesale customers pay for electricity purchases through the *power pool*, including *power pool* expenses and some transmission costs.

CEMPA does not determine a specific value for the *uplift* in each period, rather an average system value is determined based on transmission expenses, typical *power pool* operating expenses, and anticipated variations between day-ahead forecast and actual SMP based on U.K. data. A single *uplift* value is applied for both B.C. and Alberta. In B.C., the transmission charges under the BC Hydro *Wholesale Transmission Services Application* (1996) to the B.C. Utilities Commission varied between \$0.001 and \$0.00753 per kWh (BCH, 1996). The average is in the \$0.002 - \$0.004/kWh range. The charges for operating the *power pool* in Alberta are currently set at \$0.00007/kWh (Power Pool of Alberta), including all necessary system stability and backup services. The CEMPA *uplift* of \$0.004/kWh attempts to reflect those costs.

The equation applied by CEMPA for calculating the PSP, or the wholesale price of electricity in the jurisdiction, is detailed below:

Equation 2.5 - Pool Selling Price

Pool Selling Price = Spot Price + Uplift

Uplift =\$0.004/kWh

2.4.3.5 Annual Electricity Production, Revenue Generated and CO₂ Emissions

The annual electricity production from each technology is tracked by CEMPA. This value is compared with the CEA/NRCan electricity forecast to determine how closely the *load duration*

curve electricity demand estimate is to the actual forecast. Also, the annual revenue generated for each technology from electricity sales is determined, as well as the annual CO₂ emissions.

2.4.3.6 Summary of Spot Market Sub-Model

The CEMPA spot market sub-model achieves the tasks illustrated in Figure 2.6.

Figure 2.6 - Overview Diagram of CEMPA Spot Market Sub-Model



2.5 Sustainability Policy Implementations

2.5.1 Introduction

The purpose of the sustainability policy components in CEMPA is to forecast the economic, market, and environmental impacts of three alternative policies which are designed to enhance the market penetration of Sustainables.

The three sustainability policies are those outlined in Chapter 1 and in Section 2.2. Each is intended to be compatible with retail competition electricity markets and *power pool* market trading mechanisms. The implementation of these policies in CEMPA is achieved by modifying the Investment and Spot Market Sub-Models that were outlined in Sections 2.3 and 2.4.

2.5.2 Sustainables Portfolio Standard (SPS)

CEMPA adopts an SPS of 10% of annual electricity sales (kWh) by 2025, phased in annually over the 30 year simulation, with a 1/3% increase in the share each year. That means that by 2025, 10% of the electricity (in kWh) generated and sold must be derived from sustainable electricity generation technologies. It assumes that the market share specified by the SPS is met during all periods of the year, whether the demand is close to the system *peak* capacity or if the system is operating at *baseload* capacity.

The SPS mechanism, as defined, controls the way electricity is dispatched to meet retail <u>energy</u> demand, hence it is a spot market function. However, CEMPA controls the mechanism from the investment sub-model, as sufficient investment in new Sustainables <u>capacity</u> is necessary to meet retail sales standards. Modifications are made to the investment sub-model methodology described in Section 2.3. An assumption is made that once generating units are developed, the *power pool* structure is conducive to the full dispatch of Sustainables facilities. As such, no modification of the CEMPA spot-market model is necessary to ensure that the SPS energy requirement is met.

Figure 2.7 provides an overview of the implementation of the SPS mechanism in CEMPA, and a description follows.

Figure 2.7 - CEMPA Implementation of Sustainables Portfolio Standard



Investment Sub-Model Issues and Modifications

The challenge of implementing an SPS mechanism is ensuring that adequate investment in Sustainables takes place so that the requirement can be met. Therefore the CEMPA SPS routine focuses exclusively on investment, rather than operating requirements. CEMPA determines technology investment under the SPS mechanism in a similar manner to the methodology described in Section 2.3, but based exclusively on the *levelized-costs*.

The SPS routine applies the following methodology.

- 1. run the regular CEMPA investment sub-model (without SPS requirement).
- 2. determine the SPS requirement in GWh.
- check if any Sustainables capacity was developed based on financial merit (in step #1), and convert that to annual GWh production with the capacity factor.
- 4. select the least *levelized cost* Sustainable, determine its maximum capacity, convert that capacity to annual GWh production.

5. allocates the remaining GWh in a similar fashion among other Sustainables if necessary.

Spot Market Sub-Model Issues and Modifications

The marginal cost of electricity production from Sustainables is equivalent to their operating and maintenance costs and additional expenses such as water license fees for small hydro or land-use fees. Fuel costs are typically zero or negative for biomass cogenerators (due to a financial credit for steam generation). In order to simulate the SPS policy mechanism, CEMPA applies the spot-market sub-model described in Section 2.4 to determine the dispatch of technologies and the market price of electricity. It is assumed that the SPS is met automatically by electricity retailers provided sufficient capacity is available, because the Sustainables are often the cheapest technologies to operate on a marginal kWh basis.

2.5.3 Non-Bypassable System Benefits Charge with Sustainability Fund

CEMPA assumes that a System Benefits Charge (SBC) is collected on all electricity sales equivalent to 3% of the Pool Selling Price (PSP). This charge is deposited into a *Sustainability Fund* which is used to subsidize Sustainables by an amount equivalent to the difference between their levelized-cost of production and the Pool Purchase Price (PPP) of electricity, which is paid to generators. In British Columbia, the wholesale price of electricity is equivalent to about one-half of the retail price for residential consumers. As such, the SBC is on the order of 1.5% of the retail price of electricity.

The mechanism relies upon separate electricity price and demand forecasts (from the "business-asusual" scenario) to determine in advance the amount of money available for investment in Sustainables on an annual basis. In year one, collection of the SBC commences. In that year, the fund administrator signs *contracts-for-differences* with producers of sustainable electricity to guarantee the price they will be paid for their electricity production over the financing terms of the plant (25 years). Once they are developed, the Sustainables compete with all existing technologies in the *Power Pool*. In years 10 and 20, the value of the *Sustainability Fund* is re-assessed to determine if there is a surplus of funds available to fund additional Sustainables development. This surplus could be created by rising electricity prices (relative to the "business-as-usual" forecast) in CEMPA which reduces the gap between the market price of electricity and their levelized costs, or due to an over-assessment of the magnitude of the subsidy required by Sustainables in year 0. The funds are distributed to a variety of technology types, in order to promote diversity of supply options. This causes the cost of the mechanism to be higher than others (SPS, EA), although it was designed in this manner to reflect what is being done in other jurisdictions such as the U.K. and California.

Figure 2.8 - CEMPA implementation of the Non-Bypassable System Benefits Charge



Investment Sub-Model Issues and Modifications

The approach undertaken for implementing the SBC is as follows.

- 1. the CEMPA Investment and Spot Market Sub-models are run to forecast the PPP and PSP of electricity over the simulation period.
- the present value (\$1995) of the Sustainability Fund is determined, and then is annualized to determine the annual subsidy to Sustainables.
- 3. the *Fund* is allocated among various classes of technologies (See Table 2.3) to foster investment in each of those technologies.
- 4. the level of investment in new Sustainables is calculated by taking the available *Fund* dollars for each technology class, and determining how many megawatts can be developed based on the annualized capital costs. An additional 20% of that capacity is developed to reflect the fact that electricity sales into the spot market will partially compensate the capital costs given the low marginal costs of operating Sustainables facilities relative to the PPP.
- 5. the CEMPA Spot Market Sub-Model is re-run to determine the operation of Sustainables and the change in PSP (to see if the amount of fund collected is different).
- 6. the surplus of *Fund* dollars is calculated by re-assessing the value of the *Fund*, allocating the subsidy to existing Sustainables, and calculating the remainder.
- 7. in years 10 and 20, the surplus Fund dollars are allocated to Sustainables.

 Table 2.3 - Allocation of Sustainability Fund among Sustainables

	British Columbia	Alberta
Wind Power	25%	50%
Solar PV	15%	20%
Tidal Power	20%	0%
Small Hydro (lower grade sites)	20%	0%
Biomass Gasifier	10%	15%
Waste Fuels Cogenerator	10%	15%

The CEMPA spot-market sub-model is executed as described in Section 2.3, but the PSP includes the System Benefits Charge, and additional Sustainables technologies are included as a result of the investment via the *Sustainability Fund* outlined above.

2.5.4 Revenue-Neutral Environmental Cost Adder

2.5.4.1 Introduction

The EA mechanism subtracts a monetary value from the price that is paid to generators (PPP), reflective of the environmental impacts of the utilization of that technology. In order for electricity generators to adequately cover their costs, they would treat the charge as an "Environmental Adder" to their "offer" price. The mechanism is designed to be "revenue-neutral", meaning that the total value of the "adders" is subtracted from the wholesale price of electricity (PSP).

Figure 2.9 illustrates the mechanics of an EA mechanism, showing that the wholesale price of electricity (PSP) is only slightly higher than if the mechanism were not in place. Without the EA mechanism in place, the PSP would have been approximately $5\not \epsilon$ / kWh (SMP= $3\not \epsilon$ for natural gas, PPP= $4\not \epsilon$, PSP= $5\not \epsilon$). With the EA mechanism in place the PSP is $5.6\not \epsilon$ /kWh in the example illustrated below. In this case, wind gets $3\not \epsilon$ / kWh more, a gas generator $1\not \epsilon$ /kWh more, and a coal generator $1\not \epsilon$ /kWh less than if the EA mechanism were not in place.

Figure 2.9 - Financial Calculations for a Revenue-Neutral Environmental Charge Mechanism

(real 1995 dollars and cents)



The EC mechanism favors sustainable electricity generation technologies, translating into longterm revenue gains for their producers. There is a financial disincentive for investors to fund the development of new facilities which incur large EA expenses (e.g. coal power), and an incentive for investment in Sustainables because the PPP is higher, and Sustainables are able to receive the full extent of that increase.

2.5.4.2 Methodology for Modelling

This incentive is difficult to represent in CEMPA because the magnitude of the financial benefit is only measurable after several years of system operation. As such it was decided to treat the EA mechanism in CEMPA like a carbon dioxide tax of \$20/tonne. It is assumed that carbon dioxide emissions are generally indicative of other pollutants such as SO_2 , NO_x , CH_4 , CO, Particulates, VOCs, and N₂O. The specific carbon taxes applied for various technologies are as follows:

Technology Name	Carbon Tax (\$/kWh) resulting from tax of \$20/tonne of CO ₂ emissions
Simple Cycle Gas	\$ 0.0106
Combined Cycle Gas	\$0.0079
Coal - Pressurized Fluidized Bed Combustion	\$ 0.0201
Coal Gasifier Turbine	\$0.0173
Simple Cycle Oil	\$ 0.0176
Combined Cycle Oil	\$0.0125
Gas Cogeneration Retrofits (with steam credit)	\$0.002

Table 2.4 - Carbon Taxes Applied for Technologies under the Environmental Adder Mechanism

Investment and Spot Market Sub-Model Issues and Modifications

The CEMPA parameters are adjusted to reflect the Environmental Charge in the operating costs of each technology. The CEMPA Investment and Spot Market Sub-models are then executed to determine investment and operation of new technologies based on the inclusion of the additional charge. Following that, an EC refund routine is executed to subtract the value of the EC collected from the PSP.

3. MODELLING INPUTS AND SCENARIOS

3.1 Modelling Input Parameters

3.1.1 Discount Rate

The real discount rate applied in the simulation is 12%, attempting to balance a public utility cost of capital of 6-9%, and a private power producer's costs of capital of 5-12%. The real discount rate used by BC Hydro in its economic analysis is set by the Crown Corporations Secretariat typically at 8%. In contrast, IPP developers that were surveyed indicated a much wider range of discount rates. A developer of small-hydro projects in the province of B.C. applied a real discount rate of 12% in their assessment of small hydro resources in the province (Sigma Engineering), although that study was completed over 10 years ago. A combined-cycle gas turbine developer suggested applying an 80-20 debt-equity ratio (Westcoast Power), with discount rates as low as 5% real, given the current low lending rates. Another developer suggested using a long term bond rate for a typical industrial customer (e.g. Timber West), and adding 20% to that rate to reflect the equity component (Inland Pacific). This leads to rates in the 9-11% range, depending on the company. Corporate taxation should be reflected in the discount rate, something that Crownowned are not required to pay, but IPPs and private utilities do, pointing towards the application of a discount rate on the upper end of the spectrum mentioned above.

It was decided to go on the conservative end for the CEMPA analysis, applying a 12% real discount rate, to ensure that those parties assessing the results from a cost-effectiveness standpoint would be satisfied with the representation of costs of Sustainables. Had a 5% discount rate been applied, the costs of certain Sustainables would have been significantly lower because they are capital intensive. Also, there are no guarantees that the current low lending rates will be maintained. Applying a low discount rate would work against the conservative approach undertaken. This issue is briefly investigated in Chapter 6 through a sensitivity analysis.

The CEMPA model utilizes an inflation rate of 2%, which was used to adjust all historical costs to Real 1995 dollars.

3.1.2 Commodity Prices

The energy resource price inputs that are utilized by CEMPA include.

- natural gas price forecasts.
- coal price forecasts.
- oil price forecasts.
- biomass resource prices.
- hydroelectricity water usage fees (water license).
- steam output values for cogeneration facilities.

Natural gas prices are sourced from the recent NRCan document, "Canada's Energy Outlook 1996-2020" which indicates the anticipated "Domestic Price at the Alberta Border" in real \$1995 per million cubic feet. Those values are converted to \$/GJ by multiplying the costs by a factor of 1.055. In the province of B.C., the Alberta border price is utilized, but in Alberta, a 5% reduction is applied to reflect reduced transportation costs. The values are listed in Table 3.1, with the lower price being the Alberta price.

Coal prices are derived from Liu (1995). See Table 3.1.

Oil prices are similarly derived from Liu (1995), but inflated by 30% to ensure that existing oil facilities are not operated under CEMPA except for exceptional peaking purposes. This decision was made to reflect the fact that most oil facilities (diesel) are in non-integrated areas, and would not compete on the open market anyway. As far as new technologies are concerned, there is a general consensus in the industry that the *levelized costs* of oil based technologies are not cost competitive with natural gas ones. See Table 3.1.

Biomass resource prices reflect the opportunity cost of those resources, given that they could be used to offset natural gas in existing pulp and paper operations, or for other uses. The numbers illustrated in Table 3.1 are directly proportional to natural gas prices, but are scaled to reflect the fact that a biomass boiler is less efficient than a natural gas one. The specific scaling factor of 0.89 is derived from the ISTUM model for the pulp and paper sector by taking the ratio of steam production from a biomass cogenerator versus a natural gas cogenerator. The Alberta biomass price is raised by 25% to reflect transportation costs, so its value is actually higher than the natural gas price. Fuel costs for hydroelectricity production depend on the water license fee of the provincial water or environment ministry. In B.C., large hydroelectric operations pay slightly more than small hydro producers (conversation with Bob Mathews). See Table 3.1.

Cogeneration technologies jointly produce electricity and steam from an input fuel such as biomass or natural gas, and the steam can be used for other uses, which adds additional value to the technology. As such a "steam credit" is granted to those technologies, proportional to the fuel price of natural gas, and based on the efficiency of the technology. Data from ISTUM are used to come up with that scaling factors that lead to the net fuel costs listed in Table 3.1 (Simon Fraser University, Energy Research Group, 1997). CEMPA applies a negative fuel cost for cogeneration technologies, which helps offset the higher capital costs.

Fuel Type	1995 Price	2000 Forecast	2005	2010	2015	2020	
Natural Gas	\$1.67,	\$1.96,	\$ 2.03,	\$2.09,	\$2 .07,	\$2.06,	
(AB, BC)	\$1.76	\$2.07	\$ 2.13	\$2 .19	\$2.18	\$2 .17	
Coal	\$ 0.97	\$0.99	\$ 1.02	\$1.04	\$1.15	\$1.27	
Oil	\$2.30	\$3.61	\$4 .62	\$4.98	\$5.50	\$6.07	
Biomass	\$1.58,	\$1.85,	\$ 1.91,	\$ 1.96,	\$ 1.95,	\$ 1.94,	
(BC, AB)	\$ 1.87	\$ 2.20	\$ 2.26	\$2.33	\$ 2.32	\$2 .31	
Hydro water rental (¢/kWh)	0.46, 0.48	0.46, 0.48	0.46, 0.48	0.46, 0.48	0.46, 0.48	0.46, 0	
Gas Cogen Net	-0.157,	-0.184,	-0.190,	-0.196,	-0.195,	-0.194,	
Fuel Cost (AB,	-0.165	-0.194	-0.200	-0.206	-0.205	-0.204	
BC, ¢/kWh)							
Biomass Cogen.	-0.235,	-0.276,	-0.285,	-0.293,	-0.292,	-0.290,	
Net Fuel Cost	-0.248	-0.291	-0.300	-0.309	-0.307	-0.306	
(AB,BC,¢/kWh)							

3.1.3 Load Forecast

The Canadian Electricity Association / Natural Resources Canada annual report entitled, "Electric Power in Canada" (1996) contains a load and energy forecast for each province. Table 3.2 outlines those forecasts. The capacity demand values do not include the *reserve margin*. The energy demand values include a 7% transmission line loss factor.

The forecasted capacity demand is scaled-up in CEMPA by a *reserve margin* to ensure that peak demand and annual energy requirements are met, given that most generation technologies have limited capacity factors. CEMPA applies *reserve margins* that are reflective of the capacity factors of the dominant existing generation technologies and transmission losses. The capacity factor for B.C. hydroelectric resources assumes an average water reservoir condition. In Alberta a 15% margin is added to reflect the energy shortfall of fossil fuel technologies with capacity factors of 85%. In B.C. where hydroelectricity is the dominant generation source and the systems are "energy critical", a large *energy reserve margin* is required to meet annual energy requirements, although this can also be met by trading with other jurisdictions. CEMPA assumes a 70% capacity factor for 70% of the year, requiring a *reserve margin* of 30%. This margin is decreased in proportion to the retirement rate of existing hydroelectric facilities, about 1% / year. In summary, a 22% reserve margin is applied to Alberta (consistent with CEA stat's), and a 30% margin is applied in B.C., reduced to about 22% by the year 2025, assuming that trading or DSM during energy critical periods will make up for the 7% transmission losses.

For energy demand, CEMPA utilizes the Load Duration Curve (LDC) illustrated in Figure 2.1, and converts the demand segments to energy demand. The LDC is divided into 24 segments, each representing a characteristic point of demand for the year, and the area of each segment (under the curve) represents the demand. The sum of those quantities specifies the annual GWh demand. Table 3.2 outlines these values, including 7% line losses.

The difference between the CEA/NRCan forecasted demands, and the CEMPA LDC forecasts, range from 18-21% in Alberta (with CEMPA underpredicting demand), and 0-7% in B.C. The explanation for the large divergence in the Alberta model is that a B.C. *Load Duration Curve* was applied in the Alberta model due to the unavailability of that data. Other explanations include differences in the assessment of line losses (CEMPA assumes 7%), the lack of precision by which CEMPA determines the annual demand, or a different assumption about the shape of the LDC.

Forecast	1995	2000	2005	2010	2015	2020
BC Demand (MW)	10,504	11,906	12,684	13,936	15,311	16,822
AB Demand (MW)	6,924	7,644	8,409	9,284	10,250	11,317
BC Energy (CEA, GWh)	64,660	78,017	84,063	89,987	99,841	110,77
AB Energy (CEA, GWh)	54,931	60,687	66,752	73,335	80,177	87.657
BC Energy (CEMPA, GWh)	64,712	73,464	78,312	86,100	94,633	104,00
AB Energy (CEMPA, GWh)	43,104	47,465	52,224	57,667	63,678	70,317

Table 3.2 - Load and Energy Forecasts (MW and GWh)

3.1.4 Competing Electricity Generation Technologies

The technologies selected for this analysis include the following.

- Combustion turbine technologies Simple and combined cycle gas and oil technologies. The combined cycle technologies operate at a higher efficiency but with a higher capital cost. The data source for these technologies is from Liu (1995), and the BC Hydro IEP.
- Coal pressurized fluidized bed, and coal gasifier plants, are the latest generation of coal combustion technologies, offering advantages of higher efficiency and lower local-airshed emissions (SO_x, NO_x, Particulates), but at a higher cost than the conventional coal stearn plants used in Alberta. Data source: BC Hydro IEP and Liu (1995).
- Biomass steam and gasifier turbines utilize a variety of biomass or waste products to produce electricity through a thermal steam cycle. The gasifier converts biomass into methanol or ethanol before combustion. Data source: Liu (1995).
- Small hydro typically under 20MW in capacity, they operate as run-of-river plants, without storage reservoirs. The "Optimal" technologies are those that are cost-competitive with the

cheapest conventional technologies (natural gas), while those that are "lower grade" are more expensive due to higher capital costs, lower capacity factors, or long distance from transmission corridors. Data source: Canadian Hydro Developers and Sigma Engineering.

- Waste Fuel Cogen this technology co-generates steam and electricity from waste fuels, typically at industrial facilities such as pulp & paper plants (using mill waste). This technology was characterized with information from Stothert Engineering, the developers of the proposed Purcell Power project in the East Kootenays. Because it is a cogenerator, it is given a steam credit of about 0.2 cent/kWh which offsets its variable costs. The fuel cost of the waste fuel waste fuel used (wood waste) has an *opportunity cost* proportional to the value of natural gas, as waste fuels are often consumed in the place of natural gas. The resource potential information was derived from Jaccard and Makinen (1993).
- Gas Cogen Retrofits this technology involves retrofitting or replacing an existing natural gas boiler in an industrial, commercial or institutional facility with a cogenerator. CEMPA also applies a steam credit equivalent to the technology fuel costs. It turns out that the steam credit exceeds the cost of the natural gas fuel, because cogenerators are more efficient at producing steam than a steam boiler is, and therefore it has a negative net fuel cost. Data source: BC Government study and Willis Energy.
- Fuel Cell cogenerators convert natural gas, methanol or hydrogen into electricity and heat without combustion, at a higher efficiency than an equivalent combustion turbine. Data source: BC Hydro IEP.
- Large hydro are those plants with capacities greater than 100MW, usually in remote areas, requiring long transmission line extensions. The technology represented by CEMPA mimics the financial characteristics of the Peace Site C proposal (Data source: BC Hydro IEP).
- Solar technologies, PV and Parabolic-through both convert solar radiation into electricity.
 PV cells produce electricity directly from radiation through a semiconductor device, and the
 Parabolic collector produces steam which drives a turbine. Data source: BC Hydro IEP and
 Johansson et.al. (1993).
- Wind Wind generators produce electricity directly from the wind using a airfoil and turbine.
 Data source: BC Hydro IEP, and VisionQuest Wind Electric.
- Tidal Power Tidal generators use a vertical-axis hydraulic turbine to generate electricity from tidal flows. Data source: BC Hydro IEP and BC Government Energy Resources Map.

The development of new technologies is simulated by the CEMPA Investment Sub-Model, and competition between new and existing technologies is simulated in the Spot Market Sub-Model. Table 3.3 outlines several technical and financial characteristics of each technology, and Table 3.4 outlines those characteristics for existing technologies. Tables 3.3, and 3.4 only present technology parameters from the CEMPA B.C. model, because they are virtually identical for the Alberta model, and any differences are explained in the notes below the tables. The parameters include the following.

- fixed costs (capital and fixed operating costs) in year 1 of the simulation, annualized over a 25 year period at 12% discount rate (dollars per kW).
- non-fuel variable costs operating and maintenance costs (dollars per kWh).
- fuel-costs in the first year of the simulation in cents per kWh.
- capacity factors of each technology (%).
- operating life of each technology (years).
- the loss of load probability of each technology (% used in CEMPA Spot Market Sub-Model).
- the CO₂ emissions from the technology (tonnes per GWh).
- the annual retirement rate of existing technologies (%/year).

The sources of data for individual technologies is listed above with the technologies. The loss of load probability figures came from the Canadian Electricity Association annual report on reliability statistics (See Table 2.2). CO_2 emissions came from BC Hydro and Liu (1995).

Table 3.3 - Technology Parameters

Technology Parameters - British Columbia													
Ann	ualized Costs		Fuel Costs	Capacity	Operating	ous of Load	on Dioxide						
Name	Fixed Costs	OLM	In year 1	Factor	Life	Probability	Emissions						
	S/KW	\$/kWh	¢/ki//h	percent	Years	percent	es per GWh						
Simple cycle gas	70	0.60	1.88	85.0%	25	0.0444%	530.00						
Combined cycle gas	103	0.59	1.41	85.0%	25	0.0444%	396.94						
Coal - Pressurized Fluidized Bed	228	0.43	1.13	85.0%	25	0.1002%	1,007.00						
Coal gasifier turbine	297	0.54	0.97	85.0%	25	0.0444%	865.00						
Simple cycle oil (b)	91	0.57	2.73	85.0%	25	0.1002%	878.38						
Combined cycle oil	132	0.32	1.94	85.0%	25	0.0444%	624.56						
Biomass steam	207	0.21	1.48	85.0%	25	0.1002%	-						
Biomass gasifier turbine	235	0.14	1.25	85.0%	25	0.0444%	-						
Small hydro - Optimal	206	0.66	0.00	65.0%	50	0.0393%	-						
Waste fuels cogen	443	0.64	-0.25	90.0%	25	0.1002%	-						
Gas cogen retrofits	251	0.64	-0.17	76.5%	25	0.0444%	100.00						
Fuel Cell Cogenerator	769	0.01	1.01	65.0%	25	0.0100%	100.00						
Large Hydro	285	0.69	0.00	62.5%	50	0.0393%	•						
Solar PV	3033	0.00	0.00	99.0%	40	1.0000%	-						
Parabolic-Trough Solar-Power Plan	2350	0.00	0.00	95.0%	40	1.0000%	•						
Wind Generators	641	0.01	0.00	90.0%	25	1.0000%							
Tidal Power	978	0.01	0.00	90.0%	40	0.0393%	-						
Small Hydro - lower grade	384	0.58	0.00	70.1%	50	0.0393%	-						

Notes: In the Alberta model, all the fuel costs in year one are lower for those technologies based on natural gas (with the negative fuel cost for cogeneration technologies closer to zero, indicating less benefit), and higher for biomass technologies. Solar PV technologies are \$2424/kW because of better solar resources in Alberta, Parabolic-Trough Solar is \$1846/kW (same justification), and wind technologies are cheaper at \$531/kW because of better wind conditions in S-W Alberta.

Technology Parameters - Existing Technologies - B.C.												
<u>Name</u>	Variable Costs <u>in year 1</u> ¢/kWh	Capacity <u>Factor</u> percent	Annual Retire- <u>ment Rate</u> %/year	Loss of Load Probability percent	Carbon Dioxide <u>Emissions</u> tonnes per GWh							
Natural Gas	2.48	85.0%	1%	0.1002%	530.00							
Coal	1.61	85.0%	1%	0.1002%	1,007.00							
l Oil	3.44	85.0%	5%	0.1002%	878.38							
Hydro	1.00	70.0%	1%	0.0393%	-							
Nuclear	2.20	85.0%	0%	0.0488%	-							
Biomass	3.02	85.0%	1%	0.1002%								

Table 3.4 - Parameters for Existing Technologies

Note: In the Alberta model, the variable costs for natural gas are lower, and those for biomass are higher. Also the annual retirement rates are higher in Alberta, 2.5% / year for all thermal technologies.

The annual retirement rate of existing technologies was selected based on information from BC Hydro, with the selection of the retirement rate of oil facilities arbitrarily set high to phase them out and ensure that they don't overly contribute to peak demand in CEMPA. The Alberta figures (2.5%/year) are representative of the typical life of generation plants and information provided from the Alberta Power Pool. It should be noted that such a straight-line retirement rate is unrealistic, as plants are typically retired entirely at one time, and the capacity of exiting technologies is reduced in a stepwise fashion. However, it is beyond the scope of this study to undertake a detailed assessment of individual generation facilities.

Table 3.5 outlines the development limits for particular technologies, specified by province, given that some technologies have limited resources or limited available sites for certain technologies. Those technologies not listed have no specified development limits.

Technology Name	B.C. Development	Alberta Development			
	Limit	Limit			
Biomass steam	500 MW	200 MW			
Biomass gasifier	500 MW	200 MW			
Large hydro	500 MW	0 MW			
Optimal small hydro	300 MW	200 MW			
Waste-fuel cogenerator	444 MW	100 MW			
Less-Optimal small hydro	1,026 MW	0 MW			
Gas cogen retrofits	1000 MW	500 MW			
Wind Farm	3,257 MW	2,509- 18,561 MW			
Tidal Generator	3,186 MW	0 MW			
Solar technologies	1,000 MW	1,000 MW			

Table 3.5 - Development Limits for Technologies (MW)

CEMPA calculates the *levelized-costs* of all the technologies on an annual basis, integrating the annualized capital cost data with variable costs and expected generation of energy. See Equation 2.1 for a description of that calculation. The *levelized cost* information for the B.C. simulation model is listed in Table 3.6, and that information for Alberta is listed in Table 3.7.

Technology Name	1995	1096	1997	1998	1999	2000	2001	2002	2003	2004	2005	2005	2007	2008	2009	201
Simple cycle pas	\$0.0369	\$0.0373	\$0.0376	\$0.0379	\$0.0381	\$0.0383	\$0.0384	\$0.0385	\$0.0386	\$0.0387	\$0.0388	\$0.0390	\$0.0391	\$0.0392	\$0.0393	\$0.039
Simple cycle oil (b)	\$0.0620	\$0.0644	\$0.0666	\$0.0688	\$0.0708	\$0.0726	\$0.0743	\$0.0759	\$0.0774	\$0.0788	\$0.0800	\$0.0811	\$0.0823	\$0.0835	\$0.0847	\$0.086
Combined cycle gas	\$0,0358	\$0,0361	\$0,0363	\$0,0365	\$0.0367	\$0.0368	\$0.0369	\$0.0370	\$0.0371	\$0.0372	\$0.0372	\$0.0373	\$0.0374	\$0.0375	\$0.0376	\$0.037
Combined cycle oil	\$0.0522	\$0.0539	\$0.0555	\$0.0570	\$0.0584	\$0.0597	\$0.0609	\$0.0620	\$0.0631	\$0.0641	\$0.0650	\$0.0658	\$0.0666	\$0.0675	\$0.0684	\$0,069
Small hydro - Optimal	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.042
Biomass steam	\$0.0465	\$0.0468	\$0.0470	\$0.0472	\$0.0474	\$0.0475	\$0.0476	\$0.0477	\$0.0478	\$0.0479	\$0.0480	\$0.0481	\$0.0482	\$0.0483	\$0.0484	\$0.048
Coat - Pressurized Fluidi	\$0,0467	\$0,0468	\$0.0469	\$0.0470	\$0.0471	\$0.0472	\$0.0473	\$0.0474	\$0.0476	\$0.0477	\$0.0479	\$0.0480	\$0.0482	\$0.0484	\$0.0487	\$0.048
Biomass pasifier turbine	\$0.0472	\$0.0474	\$0.0476	\$0.0478	\$0.0480	\$0.0481	\$0.0481	\$0.0482	\$0.0483	\$0,0484	\$0.0485	\$0.0485	\$D.0486	\$0.0487	\$0.0488	\$0.048
Gas copen retrofits	\$0,0419	\$0.0419	\$0.0418	\$0.0418	\$0.0418	\$0.0418	\$0.0418	\$0.0418	\$0.0418	\$0.0417	\$0.0417	\$0.0417	\$0.0417	\$0.0417	\$0.0417	\$0.041
arge Hydro	\$0,0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0.059
Coal gasifier turbine	\$0,0555	\$0,0555	\$0.0556	\$0.0557	\$0.0558	\$0.0559	\$0.0560	\$0.0561	\$0.0562	\$0.0563	\$0.0565	\$0.0566	\$0.0568	\$0.0570	\$0.0572	\$0.057
Small Hydro - lower grad	\$0,0682	\$0.0682	\$0.0682	\$0.0682	\$0.0682	\$0.0682	\$0.0682	\$0.0682	\$0.0682	\$0.0682	\$0.0582	\$0.0682	\$0.0682	\$0.0682	\$0.0682	\$0.068
Waste fuels copen	\$0.0598	\$0.0597	\$0.0597	\$0.0597	\$0.0596	\$0.0596	\$0 0596	\$0 0596	\$0.0596	\$0.0596	\$0.0595	\$0.0595	\$0.0595	\$0,0595	\$0.0595	\$0,059
Wind Generators	\$0.0814	\$0.0809	\$0,0803	\$0.0798	\$0.0793	\$0.0787	\$0.0782	\$0.0777	\$0.0772	\$0.0767	\$0.0761	\$0.0756	\$0.0751	\$0.0746	\$0.0741	\$0.073
Fuel Cell Cogenerator	\$0,1149	\$0.1143	\$0.1136	\$0.1130	\$0,1123	\$0,1116	\$0.1109	\$0.1102	\$0,1095	\$0,1068	\$0.1081	\$0.1074	\$0,1068	\$0,1061	\$0.1054	\$0.104
Tidal Power	\$0,1241	\$0.1224	\$0.1207	\$0.1190	\$0.1173	\$0.1157	\$0.1140	\$0.1124	\$0.1109	\$0.1093	\$0.1078	\$0.1063	\$0.1048	\$0.1033	\$0.1018	\$0.100
Parabolic-Trough Solar-	\$0,2824	\$0.2783	\$0.2744	\$0 2705	\$0.2666	\$0.2628	\$0.2590	\$0.2553	\$0.2517	\$0.2481	\$0.2445	\$0.2410	\$0.2376	\$0.2342	\$0.2309	\$0.227
Solar PV	\$0,3498	\$0.3438	\$0.3380	\$0.3322	\$0.3265	\$0.3210	\$0.3155	\$0 3101	\$0.3048	\$0.2996	\$0.2945	\$0.2895	\$0.2846	\$0.2797	\$0.2750	\$0.270
Technology Name	2011	2012	2013	2014	2015	2016	2017	2018	2019	2020	2021	2022	2023	2024	2025	
Simple cycle gas	\$0.0396	\$0.0397	\$0.0399	\$0.0401	\$0 0403	\$0 0406	\$0.0409	\$0.0413	\$0.0417	\$0.0422	\$0.0427	\$0.0433	\$0.0438	\$0.0444	\$0.0450	
Simple cycle oil (b)	\$0.0874	\$0.0688	\$0.0902	\$0.0917	\$0.0931	\$0.0946	\$0.0962	\$0.0977	\$0.0993	\$0.1010	\$0.1026	\$0.1043	\$0.1060	\$0.1078	\$0.1096	
Combined cycle gas	\$0.0378	\$0.0379	\$0.0380	\$0.0382	\$0.0384	\$0.0386	\$0.0388	\$ 0 0391	\$0.0394	\$0.0397	\$0.0401	\$0.0406	\$0.0410	\$0.0414	\$0.0418	
Combined cycle oil	\$0.0702	\$0.0712	\$0.0722	\$0.0733	\$0.0743	\$0.0754	\$0.0765	\$0.0776	\$0.0787	\$0.0799	\$0.0811	\$0.0823	\$0.0835	\$0.0647	\$0.0860	
Small hydro - Optimal	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0 0427	\$0.0427	\$0 0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	
Biomass steam	\$0.0486	\$0.0487	\$0.0488	\$0.0490	\$0.0492	\$0.0494	\$0.0496	\$0 0499	\$0.0502	\$0.0506	\$0.0510	\$0.0514	\$0.0519	\$0.0523	\$0.0527	
Coal - Pressurized Fluidi	\$0.0492	\$0.0495	\$0.0498	\$0.0501	\$0.0504	\$0.0507	\$0.0510	\$0 0513	\$0.0517	\$0.0520	\$0.0523	\$0.0527	\$0.0530	\$0.0534	\$0.0538	
Biomass gasifier turbine	\$0,0489	\$0.0490	\$0.0492	\$0.0493	\$0.0495	\$0.0496	\$0.0499	\$0 0501	\$0.0504	\$0.0507	\$0.0510	\$0.0514	\$0.0518	\$0.0521	\$0.0525	
Gas cogen retrofils	\$0.0417	\$0.0417	\$0.0416	\$0.0416	\$0.0416	\$0.0416	\$0.0415	\$0.0415	\$0.0415	\$0.0414	\$0.0414	\$0.0413	\$0.0413	\$0.0412	\$0.0412	
Large Hydro	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0 0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	
Coal gasifier turbine	\$0.0576	\$0.0579	\$0.0581	\$0.0584	\$0.0586	\$0 0589	\$0.0592	\$0.0594	\$0.0597	\$0.0600	\$0.0603	\$0.0606	\$0.0609	\$0.0612	\$0.0615	
Small Hydro - lower grad	\$0.0682	\$0.0682	\$0.0682	\$0.0682	\$0.0682	\$0 0682	\$0.0682	\$0.0682	\$0.0682	\$0.0682	\$0.0682	\$0.0682	\$0.0692	\$0.0682	\$0.0682	
Waste fuels cogen	\$0.0594	\$0.0594	\$0.0594	\$0.0594	\$0 0593	\$0.0593	\$0.0593	\$0 0592	\$0.0592	\$0.0591	\$0.0590	\$0.0590	\$0.0589	\$0.0588	\$0.0587	
Wind Generators	\$0.0731	\$0.0727	\$0.0722	\$0.0717	\$0.0712	\$0.0707	\$0.0703	20 0698	\$0.0693	\$0.0689	\$0.0684	\$0.0679	\$0.0675	\$0.0670	\$0.0666	
Fuel Cell Cogenerator	\$0.1041	\$0.1035	\$0.1029	\$0.1023	\$0.1017	\$0.1012	\$0.1006	\$0 1002	\$0.0997	\$0.0993	\$0 0989	\$0.0985	\$0.0982	\$0.0978	\$0.0975	
Tidal Power	\$0.0990	\$0.0976	\$0.0962	\$0.0949	\$0.0935	\$0.0922	\$0.0909	\$0 0897	\$0 0884	\$0.0872	\$0.0859	\$0.0847	\$0.0835	\$0.0824	\$0.0812	
Parabolic-Trough Solar-	\$0.2243	\$0.2211	\$0.2180	\$0.2148	\$0.2118	\$0.2087	\$0.2058	\$0 2028	\$0,1999	\$0.1971	\$0.1943	\$ 0.1915	\$0.1687	\$0.1860	\$0.1834	
· · · · · · · · · · · · · · · · · · ·																



Table 3.6 - Levelized Costs of New Technologies in B.C. (\$1995/kWh)

Annual Levelize	d Costs	- Alberta	Model -	BAU												
Technology Name	1995	1996	1907	1990	1999	2000	2001	2002	2003	2004	2005	2006	2007	2006	2008	2010
Simple cycle gas	\$0.0359	\$0.0362	\$0 0365	\$0.0367	\$0 0370	\$0.0371	\$0.0372	\$0.0373	\$0.0375	\$0.0376	\$0.0377	\$0.0378	\$0.0379	\$0.0380	\$0.0381	\$0.0383
Simple cycle oll (b)	\$0.0620	\$0.0644	\$0.0666	\$0.0668	\$0.0706	\$0.0726	\$0.0743	\$0.0759	\$0.0774	\$0.0788	\$0.0500	\$0.0811	\$0.0623	\$0.0635	\$0.0647	\$0.0860
Combined cycle gas	\$0 0350	\$0.0353	\$0.0355	\$0.0357	\$0.0358	\$0.0360	\$0.0360	\$0.0361	\$0.0362	\$0.0363	\$0.0364	\$0.0365	\$0.0365	\$0.0366	\$0.0367	\$0.0368
Combined cycle oli	\$0.0522	\$0 0539	\$0.0555	\$0.0570	\$0.0584	\$0.0597	\$0.0609	\$0.0620	\$0.0631	\$0.0541	\$0.0650	\$0.0658	\$0.0666	\$0.0675	\$0.0684	\$0.0093
Smell hydro	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427
Biomess steem	\$0.0520	\$0.0524	\$0.0527	\$0.0530	\$0.0532	\$0.0534	\$0.0535	\$0.0536	\$0.0537	\$0.0539	\$0.0540	\$0.0541	\$0.0542	\$0.0544	\$0.0545	\$0.0546
Coel - Pressurized Fluidiz	\$0.0467	\$0.0468	\$0 0469	\$0.0470	\$0.0471	\$0.0472	\$0.0473	\$0.0474	\$0.0476	\$0.0477	\$0.0479	\$0.0480	\$0.0482	\$0.0454	\$0.0487	\$0.0489
Biomess gesiller tabine	\$0.0499	\$0.0501	\$0.0504	\$0.0506	\$0.0508	\$0.0509	\$0 0510	\$0.0511	\$0.0512	\$0.0513	\$0.0514	\$0.0515	\$0.0516	\$0.0517	\$0.0518	\$0.0519
Ges cogen refroits	\$0.0420	\$0.0420	\$0.0419	\$0.0419	\$0.0419	\$0.0419	\$0.0419	\$0.0419	\$0.0419	\$0.0418	\$0.0418	\$0.0418	\$0.0418	\$0.0418	\$0.0418	\$0.0418
Large Hydro	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0 0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590
Cosi gesifier Labine	\$0.0544	\$0.0544	\$0.0545	\$0.0545	\$0.0547	\$0.0548	\$0.0549	\$0.0550	\$0.0551	\$0.0552	\$0.0553	\$0.0554	\$0.0556	\$0.0557	\$0.0559	\$0.0561
Weste fuels cogen	\$0.0599	\$0.0599	\$0.0598	\$0.0598	\$0.0596	\$0.0598	\$0.0597	\$0.0597	\$0.0597	\$0.0597	\$0.0597	\$0.0597	\$0.0597	\$0.0595	\$0.0596	\$0.0596
Wind Generators	\$0.0702	\$0.0897	\$0 0693	\$0.0688	\$0.0683	\$0.0679	\$0.0674	\$0.0670	\$0.0665	\$0.0661	\$0.0656	\$0.0652	\$0.0548	\$0.0643	\$0.0639	\$0.0535
Fuel Cell Cogenerator	\$0.1143	\$0.1137	\$0.1131	\$0 1124	\$0.1117	\$0.1110	\$0.1103	\$0.1096	\$0.1089	\$0.1062	\$0.1075	\$0.1068	\$0.1061	\$0.1054	\$0,1048	\$0.1041
Tidal Power	\$0.1241	\$0.1224	\$0 1207	\$0.1190	\$0.1173	\$0.1157	\$0.1140	\$0.1124	\$0.1109	\$0.1093	\$0.1078	\$0.1063	\$0.1048	\$0.1033	\$0.1018	\$0.1004
Perebolic-Trough Soler-P	\$0.2269	\$0.2237	\$0 2205	\$0.2173	\$0.2142	\$0.2112	\$0 2082	\$0.2052	\$0.2023	\$0.1994	\$0.1965	\$0.1937	\$0.1909	\$0.1852	\$0.1855	\$0.1829
Solar PV	\$0.2799	\$0.2751	\$0.2704	\$0 2658	\$0.2613	\$0 2568	\$0.2525	\$0.2482	\$0.2439	\$0.2398	\$0.2357	\$0.2317	\$0.2277	\$0.2238	\$0.2200	\$0.2163
Technology Name	2011	2012	2013	2014	2015	2016	2017	2018	2019	2020	2021	2022	2023	2024	2025	
Simple cycle gas	\$0.0384	\$0.0385	\$0.0387	\$0 0389	\$0.0391	\$0.0394	\$0.0397	\$0.0400	\$0.0404	\$0 0408	\$0.0413	\$0.0419	\$0.0424	\$0.0429	\$0.0435	
Simple cycle oil (b)	\$0.0674	\$0 0888	\$0.0902	\$0 0917	\$0.0931	\$0.0946	\$0.0962	\$0.0977	\$0.0993	\$0.1010	\$0.1026	\$0.1043	\$0.1060	\$9.1078	\$0.1096	
Combined cycle ges	\$0.0369	\$0.0370	\$0.0371	\$0 0373	\$0 0374	\$0.0376	\$0.0379	\$0.0381	\$0.0384	\$0.0387	\$0.0391	\$0.0395	\$0.0399	\$0.0403	\$0.0407	
Combined cycle all	\$0.0702	\$0 0712	\$0.0722	\$0 0733	\$0 0743	\$0.0754	\$0.0765	\$0.0776	\$0.0787	\$0.0799	\$0.0811	\$0.0623	\$0.0835	\$0.0847	\$0.0860	
Smel hydro	\$0.0427	\$0.0427	\$0.0427	\$0 0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	\$0.0427	
Biomess steam	\$0.0547	\$0.0549	\$0.0551	\$0.0553	\$0.0555	\$0.0558	\$0.0561	\$0.0565	\$0.0569	\$0.0574	\$0.0580	\$0.0585	\$0.0591	\$0.0597	\$0.0603	
Coel - Pressurized Fluidiz	\$0.0492	\$0.0495	\$0.0496	\$0.0501	\$0.0504	\$0.0507	\$0.0510	\$0.0513	\$0.0517	\$0.0520	\$0.0523	\$0.0527	\$0.0530	\$0.0534	\$0 0538	
Biomess gasifier turbine	\$0.0520	\$0.0521	\$0.0522	\$0.0524	\$0.0526	\$0.0528	\$0.0530	\$0.0533	\$0.0536	\$0.0540	\$0.0544	\$0.0549	\$0.0553	\$0.0558	\$0.0562	
Gas cogen retrofits	\$0.0418	\$0.0418	\$0.0417	\$0.0417	\$0 0417	\$0.0417	\$0.0417	\$0.0416	\$0.0416	\$0.0416	\$0.0415	\$0.0415	\$0.0414	\$0.0414	\$0.0413	
Large Hydro	\$0.0590	\$0.0590	\$0 0590	\$0 0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	\$0 0590	\$0.0590	\$0.0590	\$0.0590	\$0.0590	
Coal gasifier turbing	\$0.0563	\$0.0565	\$0.0568	\$0 0570	\$0.0572	\$0.0575	\$0 0577	\$0.0579	\$0.0582	\$0.0585	\$0 0587	\$0.0590	\$0.0593	\$0.0595	\$0.0598	
Waste fuels cogen	\$0.0596	\$0.0596	\$0.0596	\$0 0595	\$0.0595	\$0 0595	\$0.0594	\$0.0594	\$0.0593	\$0.0593	\$0.0592	\$0.0591	\$0.0591	\$0.0590	\$0.0589	
Wind Generators	\$0.0631	\$0.0626	\$0.0622	\$0.0618	\$0.0614	\$0.0610	\$0.0606	\$0.0602	\$0.0598	\$0.0594	\$0 0590	\$0.0586	\$0.0582	\$0.0578	\$0.0574	
Fuel Cell Cogenerator	\$0.1035	\$0.1028	\$0 1022	\$0 1016	\$0.1010	\$0.1005	\$0.1000	\$0 0995	\$0 0990	\$0.0966	\$0 0982	\$0.0978	\$0.0974	\$0.0971	\$0.0967	
Tidel Power	\$0.0990	\$0 0976	\$0.0962	\$0 0949	\$0.0935	\$0.0922	\$0.0909	\$0 0897	\$0 0884	\$0.0672	\$0.0859	\$0.0847	\$0.0835	\$0.0824	\$0.0612	
Perebolic-Trough Soler-P	\$0.1803	\$0.1777	\$0 1752	\$0 1726	\$0.1702	\$0 1678	\$0 1654	\$0.1630	\$0.1607	\$0.1584	\$0.1561	\$0.1539	\$0.1517	\$0.1495	\$0.1474	
Solar PV	\$0.2126	\$0 2090	\$0 2054	\$0 2019	\$0,1985	\$0.1951	\$0.1918	\$0.1885	\$0.1853	\$0.1821	\$0.1790	\$0.1760	\$0.1730	\$0.1700	\$0.1671	

3.2 Policy Scenarios

There are four modelling scenarios that were applied under CEMPA for this analysis.

- Business-as-usual (BAU). This scenario is the base case under a competitive electricity
 market, that is the outcome that could be expected if no explicit policies to promote
 Sustainables are implemented. Under this scenario, the CEMPA Investment and Spot-Market
 Sub-Models are executed under the methodology outlined in Sections 2.3 and 2.4.
- Sustainables Portfolio Standard (SPS). This scenario illustrates the forecasted outcome of a 10% Sustainables portfolio requirement, phased in over the next 30 years. Under this scenario, the CEMPA SPS Sub-Model is executed under the methodology outlined in Section 2.5.2.
- 3. A Non-Bypassable System Benefits Charge (SBC) with Sustainability Fund. This scenario illustrates the outcome of subsidizing Sustainables with a 3% charge on electricity sales, with the subsidy being divided among different classes of Sustainables according to a predetermined allocation. Under this scenario, the CEMPA SBC Sub-Model is executed under the methodology outlined in Section 2.5.3.
- 4. A Revenue-neutral Environmental Adder (EA) Scenario, whereby a \$20/tonne CO₂ charge is added to the variable costs of all technologies (representative of a broad range of environmental externalities), and the extra funds that are collected are redistributed to customers through reduced rates. Under this scenario, the CEMPA EA Sub-Model is executed under the methodology outlined in Section 2.5.4.

4. SIMULATION RESULTS

Based on the methodology presented in Chapter 2, and the model parameters described in Chapter 3, the results from the CEMPA model are presented in this chapter in a sequential fashion for British Columbia and Alberta over a simulation period of 30 years, from 1995 to 2025. The results which are presented include the following outputs.

Electricity Generation

- Total generation capacity (MW) by technology type.
- Annual energy generation (GWh) by technology type (in appendix).
- Investment in Sustainables Capacity.
- Generation by Sustainables.
- Generation Share by Sustainables.

Rates and Cost

- Peak wholesale electricity rates (\$/GWh).
- Baseload wholesale electricity rates (\$/GWh).
- Total expenditures on electricity (million \$).

Emissions

• Carbon Dioxide Emissions (tonnes).

For each of the results, the specific scenario associated with the output is also specified. The scenarios are distinguished by an acronym. Details on the scenarios are included in Section 3.2 and Chapter 2.

- Business-as-usual BAU.
- Sustainables Portfolio Standard (Sustain. Port. Stand.) SPS.
- Non-Bypassable System Benefits Charge with Sustainability Fund (Sys. Benefits Charge) -SBC.
- Revenue-neutral Environmental Adder (Environmental Adders) EA.
4.1 British Columbia Model

Table 4.1 outlines the market penetration of technologies under the four scenarios. The full data is contained in Appendix B in Tables B.1 through B.4.

Under the BAU scenario, other than 300MW of the most cost-effective small hydro, the entire remainder of new technologies is met by natural gas technologies, with combined-cycle turbines dominating, followed by simple-cycle turbines for peaking purposes and gas cogeneration retrofits for baseload operation. Under the SPS scenario, Sustainables take a greater market share, with less investment in natural gas technologies than under the BAU scenario. Biomass technologies are the predominant Sustainables developed as they are the most cost-effective after Optimal Small Hydro resources are used-up. Under the SBC scenario, a diversity of Sustainables are developed, including wind, tidal, solar, hydro and biomass technologies. The EA scenario shows an outcome similar to that of the SPS scenario, again only the most cost-effective Sustainables being developed.

Tech. Name	BAU	SPS	SBC	EA
Small Hydro	300	300	353	300
S.C. Gas	4583	4517	4583	3682
C.C. Gas	6552	5171	6355	6336
Gas Cogen.	1000	1000	1000	1000
Biomass	0	1444	67	1038
Other Sustain.	0	0	77	0
Existing Hydro	8301	8301	8301	8301
Existing Gas	763	763	763	763
Existing	394	394	394	394
Biomass		·····		

Table 4.1 - Market Penetration (MW of capacity) of Technologies in 2025 - B.C.

The expected energy production in GWh of various technologies under the BAU scenario is outlined in Table B.5 of Appendix B. This is also reflective of operations in other scenarios except for the EA which maximizes the operation of non-carbon based resources. Hydroelectricity dominates the low cost generation with variable costs at 1cent/kWh, and other technologies are used during higher demand periods. Combined-cycle gas plants dominate new technologies, producing 43,817 GWh in 2025. Gas cogeneration technologies operate at nearly full capacity (6700 GWh in 2025) because of their low operating costs. Simple-cycle gas turbines, are used predominantly as peaking plants (9057 GWh in 2025), as they have higher fuel costs than both combined cycle plants and cogeneration units. These results also indicate that existing natural gas facilities (Burrard Thermal) are run at well below capacity under a competitive market. The annual production capability of those technologies is about 5681 GWh in 2025, and yet the plants only produced 355 GWh in that year (6% of the capability). Their variable costs are higher than those of hydroelectricity or new technologies, so they are used predominantly as peaking plants under competitive markets.

The EA scenario generating statistics, listed in Table B.6 (Appendix B), illustrate a 36% reduction in the operation of simple cycle gas turbines, a smaller reduction in the operation of combined cycle gas turbines, with all of the shortfall being met by biomass technologies. This is due to the fact that net carbon dioxide emissions from biomass are zero, and therefore their operating costs remain unaffected by the "environmental adder".

Figure 4.1 illustrates the percentage share of generation met from Sustainables. Figure B.1 in the Appendix illustrates the market penetration of Sustainables in MW and Figure B.2 illustrates the annual electricity generation from Sustainables in GWh's. Under the BAU case, the level of investment in Sustainables is very limited, with an exception of a small amount of small hydro development (300 MW). This is brought on line all at once, enabled by the introduction of competitive markets. The investment in Sustainables is highest under the SPS scenario, at 1750 MW in 2025. In-turn, their generation accounts for about 11% of the total generation in the year 2025. The EA mechanism fosters the development of 1340 MW of Sustainables capacity by 2025, or about 8% of the generation. Finally, the SBC mechanism fosters the development of about 497 MW of Sustainables capacity, resulting in a generation share of only 3%, compared with about 1.5% under the BAU scenario. The factors leading to these differences are discussed in Chapter 5.



Figure 4.1 - Generation Share by Sustainables (percent)- all scenarios

Table 4.2 illustrates the wholesale spot price of electricity in the years 1995 and 2025 for each of the scenarios. The wholesale spot price (the PSP) includes capacity adders, transmission expenses, and administrative charges. CEMPA characterizes the spot electricity price for 24 different positions on the *load duration curve*, with the first 5-10 representing *peaking* periods, and the remainder being *shoulder* and *baseload* conditions. The "peaking" spot price listed is representative of those hours of the year where the demand for electricity is between 75% and 77% of the peak demand for the year. This is representative of 365 non-consecutive hours of the year. The "baseload" spot price is representative of those periods were demand is between 53 % and 56% of the peak. The data are also included in Appendix B, Figures B.3 and B.4.

Scenario	1995 Peak	2025 Peak	1995 Baseload	2025 Baseload
BAU	\$ 29	\$ 36	\$ 16	\$29
SPS	\$29	\$ 36	\$ 16	\$29
SBC	\$30	\$37	\$17	\$30
EA	\$ 39	\$ 43	\$ 16	\$34

Table 4.2 - Peaking and Baseload Wholesale Spot Prices of Electricity (PSP) (1995\$/MWh)

Under all conditions, the spot price under the BAU scenario is the same as that under the SPS. The SBC scenario includes a \$1 / MWh price premium to fund the subsidy for Sustainables. The EA spot prices are higher because of the inclusion of an "environmental adder" on all spot market transactions, minus the refund to customers. However, the refund is not sufficient to completely counteract the cost of the adder to each generator's operation. Thus, a price premium of up to \$10 is paid. Figure B.3 illustrates a flat spot price profile over the simulation period. In contrast, Figure B.4 indicates a sudden jump in the spot price around 2005, due to the retirement of existing low cost generation facilities (see assumptions in Chapter 3, Table 3.4) that are met with more expensive combined-cycle gas turbines. Most consumers will not feel this price jump directly, as their purchase price would reflect an average of spot prices.

Table 4.3 lists the forecasted cost of electricity in 2025, equivalent to the *levelized-cost* of production from each technology, times the actual energy it produces over the year. This information is also illustrated in Figure B.5 in Appendix B. The wires charge associated with the

SBC scenario is not added to this cost figure because that money is directed towards the capital costs of Sustainables which is reflected in the *levelized-cost*. Also, the environmental charge associated with the EA scenario is not added, because it is refunded to customers. Chapter 5 includes an illustration of the incremental costs of adopting a policy measure over and above the costs of the BAU.

Scenario	Cost (Millions of Dollars)
BAU	4,217
SPS	4,348
SBC	4,240
EA	4,307

Table 4.3 - Cost of Electricity in the Year 2025 (Millon \$1995)

Carbon dioxide emissions in B.C. are illustrated in Figure 4.2. They rise sharply under the BAU scenario, from about 1Mt in 1995 to 23Mt in 2025. This is due to significant investment in new natural gas technologies and retirement of existing hydroelectric facilities. Emissions are 3.5 MT less (in 2010) with the EA scenario, and about 1.7MT and 0.5MT less under the SPS and SBC scenarios respectively. In the long term, the SPS is the optimal scenario for minimizing CO_2 emissions, with 4.4MT less emissions in 2025 than under the BAU.



Figure 4.2 - Carbon Dioxide Emissions (tonnes) - all scenarios

4.2 Alberta Model

Table 4.4 outlines the market penetration of technologies under the four scenarios. The full data is contained in Appendix C in Tables C.1 through C.4.

Under the BAU scenario, other than 200MW of the most cost-effective small hydro, the entire remainder of new technologies is met by natural gas technologies, with combined-cycle turbines dominating, followed by simple-cycle and gas cogeneration retrofit technologies. Under the SPS scenario, Sustainables take a greater market share. Biomass technologies are the predominant Sustainables developed as they are the most cost-effective after Small Hydro resources are used-up. However, in 2022, wind technologies are developed, reaching 149 MW by 2025. Under the SBC scenario, a diversity of Sustainables are developed, including wind, biomass, solar and small hydro. The EA scenario shows an outcome similar to that of the SPS scenario, again only the most cost-effective Sustainables being developed.

Existing technologies include coal, natural gas, hydro and biomass. The coal resources, equivalent to about 5850 MW in 1996, are retired at a rate of 2.5% / year, thus reducing the capacity significantly by the end of the simulation.

Tech. Name	BAU	SPS	SBC	EA
Small Hydro	200	200	200	200
S.C. Gas	4034	4034	4034	2931
C.C. Gas	6001	4870	5851	6205
Gas Cogen.	500	500	500	500
Biomass	0	1000	69	867
Other Sustain.	0	149 (wind)	81 (wind and PV)	30 (wind)
Existing Hydro	609	609	609	609
Existing Gas	859	859	859	859
Existing	95	95	95	95
Biomass				
Existing Coal	2935	2935	2935	2935

Table 4.4 - Market Penetration (MW of capacity) of Technologies in 2025 - Alberta.

The expected GWh production of various technologies under the BAU scenario is illustrated in Table C.5 (Appendix C, two parts). Coal dominates the low cost generation (21,857 GWh in 2025) with variable costs at 1.5 cents/kWh. New combined cycle gas turbines make up the remaining *baseload* and *shoulder* loads (40,859 GWh in 2025), and new simple-cycle gas technologies supplement the supply for *peak* demand (6561 GWh in 2025). Gas cogeneration technologies operate at nearly full capacity (3351 GWh in 2025) because of their low operating costs. Existing gas and biomass technologies operate below capacity because of high operating costs relative to new combined-cycle gas turbines. Existing hydroelectric resources operate at full capacity because of their low operating 200 GWh in 2025.

The EA scenario generating statistics, listed in Table C.6 (Appendix C), illustrate a 53% reduction in the operation of simple cycle gas turbines, a 38% reduction in the operation of coal facilities, and a 10% reduction in the operation of combined cycle gas turbines. The shortfall is mostly met by biomass technologies. This is due to the fact that net carbon dioxide emissions from biomass, and therefore their operating costs remain unaffected by the "environmental adder".

Figure 4.3 illustrates the percentage share of generation met from Sustainables. Figure C.1 in the Appendix illustrates the market penetration of Sustainables in MW and Figure C.2 illustrates the annual electricity generation from Sustainables in GWh's. Under the BAU case, the level of investment in Sustainables is very limited, with an exception of a small amount of small hydro development (200 MW). The investment in Sustainables is highest under the SPS scenario, at 1349 MW in 2025. In-turn, their generation accounts for about 11% of the total generation in the year 2025. The EA mechanism fosters the development of 1098 MW of Sustainables capacity by 2025, about 10% of the generation. Finally, the SBC mechanism fosters the development of about 350 MW of Sustainables capacity, resulting in a generation share of only 3%, compared with about 1.5% under the BAU scenario.



Figure 4.3 - Generation Share by Sustainables (percent)- all scenarios

Alberta - Market Share of Sustainables

Table 4.5 illustrates the wholesale spot price of electricity in the years 1995 and 2025 for each of the scenarios. The data are also included in Appendix C, Figures C.3 and C.4.

Scenario	1995 Peak	2025 Peak	1995 Baseload	2025 Baseload
BAU	\$28	\$34	\$ 22	\$28
SPS	\$28	\$34	\$ 22	\$28
SBC	\$29	\$35	\$ 23	\$ 29
EA	\$24	\$39	\$ 24	\$ 36

Table 4.5 - Peaking and Baseload Wholesale Spot Prices of Electricity (PSP) (1995\$/MWh)

Under all conditions, the spot price under the BAU scenario is the same as that under the SPS. The SBC scenario includes a \$1 / MWh price premium to fund the subsidy for Sustainables. The EA spot prices are higher because of the inclusion of an "environmental adder" on all spot market transactions, minus the refund to customers. This price premium increases over time as the demand for electricity increases, and more carbon intensive resources are required to meet the demand. In 2025, the spot price under the EA scenario exceeds the spot price under BAU by \$5 or \$6 per megawatt-hour.

An anomaly is evident for the peak wholesale price in 1995 for the EA scenario, indicating that the EA is cheaper than the BAU, extending through to the year 2003 (see Figure C.3). In this case, the consumer rebate against the PSP, equivalent to the total financial value of the "environmental adders", exceeds the impact of the policy on the SMP, or the cost of operating the marginal resource.

Table 4.6 lists the forecasted cost of electricity in 2025, equivalent to the *levelized-cost* of production from each technology, times the actual energy it produces over the year. This information is also illustrated in Figure C.5 in Appendix C. The wires charge associated with the SBC scenario is not added to this cost figure because that money is directed towards the capital costs of Sustainables which is reflected in the *levelized-cost*. Also, the environmental charge associated with the EA scenario is not added, because it is refunded to customers. Chapter 5

includes an illustration of the incremental costs of adopting a policy measure compared with the BAU.

Scenario	Cost (Millions of Dollars)
BAU	2,919
SPS	3,027
SBC	2,931
EA	3,119

Table 4.6 - Cost of Electricity in the Year 2025 (Million \$1995)

Carbon dioxide emissions in B.C. are illustrated in Figure 4.4. They rise moderately under the BAU scenario in early years, are flat in mid-years, and rise in later years, from about 39Mt in 1995 to 40.5MT in 2010 and 43Mt in 2025. This is due to significant investment in new natural gas technologies and retirement of existing coal facilities. Emissions are 0.5 Mt less with the SBC scenario, 3Mt less with the SPS, and almost 9Mt less with the EA scenario. In the long term, the SPS stabilizes emissions, while the EA scenario significantly reduces them.





Figure 4.4 - Carbon Dioxide Emissions (tonnes) - all scenarios

5. SIMULATION ANALYSIS AND EVALUATION

The purpose of this study is to compare the level of development of Sustainables under different electricity market policy scenarios, each including full *retail competition*. The scenarios include: Business as Usual (BAU), Sustainables Portfolio Standard (SPS), System Benefits Charge (SBC), and Revenue-neutral Environmental Adders (EA). The first scenario is the expected outcome without any explicit policy to foster Sustainables, which is currently the case in both Alberta and B.C. The other three scenarios encompass specific policy objectives to foster Sustainables.

This chapter expands on the results provided in Chapter 4 by analyzing them according to specific evaluative criteria or policy objectives. The criteria allow for an analysis of the merits of the three sustainability policies relative to the BAU scenario. The results of both the Alberta and B.C. models are merged in this chapter.

- Maximization of the market penetration of Sustainables in terms of investment and operation (MW installed and GWh per year operation).
- 2. Maximization of a diversity of Sustainables (number of technologies fostered).
- 3. Reduction of unit costs of Sustainables (\$/kWh).
- 4. Minimization of financial impacts peak and baseload wholesale electricity rates (\$/GWh) and total expenditure on electricity (million \$).
- 5. Minimization of the annual carbon dioxide emission abatement cost (\$ / tonne CO₂ reduced).
- 6. Administrative and operational simplicity.

The significance of each evaluative crieterion is outlined in Section 1.1.

An important distinction in this analysis is to separate those impacts related to the choice of policy parameters from those impacts which result from specific merits of each policy. For example, the market penetration of Sustainables can be influenced by the level of portfolio requirement under SPS, the size of the environmental charge under EA, or the quantity of the wires charge under the SBC. In order to assess the cost-effectiveness of the policies in their support for Sustainables, those parameters should be adjusted so that the market penetration of Sustainables is approximately the same for all policies, and then a relative financial comparison is possible. On the other hand, if the analysis focuses on issues such as the timing of development of Sustainables, administrative and political feasibility, the diversity of Sustainables developed, or impacts on the reduction of Sustainables' unit cost, then this is not necessary.

In order to capture the different analysis priorities, this chapter is divided into three sections.

First, in Section 5.1 the policies presented in Chapters 2 through 4 are analyzed according to the six criteria listed above. These policies were designed in such a manner as to be consistent with what other jurisdictions are proposing, so as to be political feasible.

Second, section 5.2 provides an analysis of the cost-effectiveness of the policies to foster Sustainables. In this case, each policy was adjusted to result in approximately the same market penetration of Sustainables. The size of the SBC and the EA were increased from those levels presented in previous chapters so that each results in a similar market share of Sustainables as was the case for the SPS scenario. The results from these simulation runs are not presented elsewhere in this study because they diverge from the magnitude of mechanisms that have been proposed or implemented in other jurisdictions.

Section 5.3 provides an overview of the policy analysis in the form of a summary matrix on the extent to which each of the policy mechanisms satisfy the six evaluative criteria.

Section 5.4 includes a general discussion on uncertainty and the accuracy of the CEMPA model.

5.1 Analysis of Policy Mechanisms

In this section, each of the policy mechanisms presented in Chapters 2 through 4 is analyzed according to the six evaluative criteria listed above. These policies were designed in a manner that is comparable with policies implemented or proposed in other jurisdictions.

The size of the SPS portfolio standard is set at 10% by 2025, the magnitude of the SBC is 3% of the wholesale price (PSP), and the size of the carbon tax for the EA is \$20/tonne. This results in different levels of financial support for Sustainables and hence a different market penetration for Sustainables.

5.1.1 Market Penetration of Sustainables

Under the business-as-usual (BAU) scenario, the market penetration of Sustainables remains flat in both B.C. and Alberta after all of the low-cost small hydro resources are tapped. The market penetration of Sustainables under all scenarios for both provinces is presented in Table 5.1. The SPS policy results in the highest market penetration of Sustainables, reaching 1750 MW and 1350 MW by the year 2025 in B.C. and Alberta respectively. The other policies (SBC and EA) could

also be designed in a manner which maximizes market penetration per unit cost and those results are presented in Section 5.2.

Scenario	B.C. Market Penetration	Alberta Market Penetration
BAU	300 MW	200 MW
SPS	1750 MW	1350 MW
SBC	500 MW	350 MW
EA	1340 MW	1100 MW

Table 5.1 - Market Penetration of Sustainables (MW) in 2025

The energy production from Sustainables (see Figures B.2 and C.2), in GWh production per year, has similar results, with the SPS and EA maximizing production.

5.1.2 Diversity of Sustainables Developed

The issue of technological diversity is illustrated in Table 5.2, showing how many different types of Sustainables are developed and operated under each policy mechanism. The SBC scenario achieves the highest diversity for Sustainables because it was designed in a manner which prioritizes diversity.

Table 5.2 - Capacity of New Sustainables Developed (MW) in 2025

Tech. Type	<u>SPS</u> B.C.	<u>SPS</u> Alberta	<u>SBC</u> B.C.	<u>SBC</u> Alberta	<u>EA</u> B.C.	<u>EA</u> Alberta
Biomass	1444	1000	67 ·	69	1037	868
Small Hydro	300	200	353.4	200	300	200
Wind	2.51	149	44.2	73	0	30
SolarPV	0	0	6.61	8.04	0	0
Tidal	0 .	0	25.82	0	0	0

5.1.3 Reduction of Unit Costs of Sustainables

Tables 5.3 and 5.4 outline how the *levelized-costs* of those technologies are reduced as a result of their increased participation in the market. Only the wind, solar PV and tidal technologies demonstrate substantial cost reductions because they are relatively new technologies. Large gains in unit cost reduction occur under those policies which result in some amount of market penetration of the technology whereby economies of manufacture are reached. Only the SBC policy fully stimulates a diversity of Sustainables, and hence it results in the highest unit cost reduction for emerging technologies. In contrast, the SPS and EA mechanisms tend to foster low-cost biomass technologies which do not realize economies of manufacture due to the technological stasis with boiler technologies (See Section 2.3.5.4).

An anomaly in the results from the Alberta model indicates that the unit cost reduction under the SPS scenario is less than that under the SBC, even though the SPS results in a higher market penetration of Sustainables by 2025 than the SBC. This is due to the fact that the market penetration under the SPS is in the last five years of the simulation, while those under the SBC occur over the entire duration of the simulation, realizing significant unit cost reductions on three occasions over the 30 year simulation period.

Tech. Type	<u>SPS</u>	1995	2025	<u>SBC</u>	1995	2025	<u>EA</u>	1995	2025
Biomass		0.0598	0.0587		0.0598	0.0587		0.0598	0.0587
		(cogen)			(cogen)			(cogen)	
Small		0.043 -	0.043 -		0.043 -	0.043 -		0.043 -	0.043 -
Hydro		0.068	0.068		0.068	0.068		0.068	0.068
Wind		0.0814	0.0666		0.0814	0.0492		0.0814	0.0666
SolarPV		0.349 8	0.2088		0.3498	0.1426		0.3498	0.2088
Tidal		0.1241	0.0812		0.1241	0.0483		0.1241	0.0812

Table 5.3 - Reduction in the Unit Cost of Sustainables (\$1995/kWh) - British Columbia

Тес ћ. Туре	<u>SPS</u>	1995	2025	<u>SBC</u>	1995	2025	<u>EA</u>	19 95	2025
Biomass		0.0599 (cogen)	0.0589		0.0599 (cogen)	0.0589		0.0599 (cogen)	0.0589
Small Hydro		0.043	0.043		0.043	0.043		0.043	0.043
Wind		0.0702	0.0448		0.0702	0.0409		0.0702	0.0518
SolarPV		0.2799	0.1671		0.2799	0.1111		0.2799	0.1671

Table 5.4 - Reduction in the Unit Cost of Sustainables (\$1995/kWh) - Alberta

5.1.4 Financial Impacts - Total Expenditure on Electricity Sales and Impacts on Spot Price

Figures 5.1 and 5.2 illustrate the incremental costs associated with the adoption of different policy scenarios relative to the BAU scenario, in essence the cost premium required to implement a policy. In Alberta, the EA mechanism is the most costly because of the large installed capacity of carbonintensive coal resources which are offset with more expensive natural gas and biomass technologies. In B.C., the SPS policy has the highest cost premium. In both cases, the expensive policy mechanism also results in a high market penetration of Sustainables and reduction in CO_2 emissions. An analysis of the cost-effectiveness of each policy in fostering Sustainables and reducting CO_2 emissions is presented in Section 5.2.

The SPS demonstrates a gradual increase in the cost premium, while the SBC applies the majority of costs in the first year, with a reduction in costs in 2020 after the original 25-year contract with the Sustainables facilities is expired and they are forced to compete at the market rate. The EA mechanism causes rapid cost increases due to the rapid development of less carbon intensive resources. In B.C., the rapid cost increases subside after the year 2000, while in Alberta, costs increase steadily over the entire simulation period.

Spot electricity prices are expected to climb slightly over the years, with sharper increases in *baseload* prices as existing low cost hydroelectric and coal generation sources are retired. The data are illustrated in Figures B.3, B.4 and C.3 and C.4 in the appendices. The difference in prices

between the BAU and SPS are minor, due to the fact that sustainable technologies have relatively low operating costs and zero fuels costs, allowing them to make low "offers" in to the power pool provided their fixed costs are covered through external contracts. The wholesale electricity price under the SBC scenario shows a small premium, due to the fact that a 3% wires charge is being added to customers' bills. The EA scenario indicates a larger premium, resulting from changes in the generation mix and from the environmental charge, even though the charge is redistributed to customers through reduced rates. This premium is higher under *peaking* conditions than *baseload* conditions because less efficient natural gas and coal plants are operated to meet the peak load with resulting increases in emissions and the value of the environmental charge.





Alberta - Policy Cost Premiums over Business-as-Usual

5.1.5 Annual Carbon Dioxide Emissions and Abatement Costs

 CO_2 emissions are expected to increase dramatically in B.C. under BAU (Figure 4.2) because of increased growth in natural gas generation, and a strong increase in demand for electricity in the province. Emissions are expected to more-or-less stabilize in Alberta (Figure 4.4) due to declining coal consumption in favor of natural gas, despite a steady increase in demand for electricity in the province. The EA scenario demonstrates sharp reductions in CO_2 emissions in Alberta through reduced operation of coal facilities and the increased market share of gas cogenerators, combined cycle gas turbines, biomass technologies, and small hydro technologies, which are developed at a faster rate under this scenario than under the BAU. The results of this scenario indicate a possible early retirement of some coal capacity facilities, as their production (GWh) is reduced by 11% in the year 2000, and 22.5% in 2010, relative to the BAU scenario.

The key indicator for assessing the economic efficiency of CO_2 emission reduction is the emission abatement cost curve that is the cost associated with the adoption of a policy mechanism relative to BAU, and the subsequent emissions reduction that follows from the adoption of that mechanism. A detailed analysis of the abatement cost will be provided in Section 5.2 where those policies are adjusted in order to result in approximately the same market penetration of Sustainables by the year 2025.

5.1.6 Administrative and Operational Simplicity

The SPS mechanism puts the onus of developing Sustainables on the electricity retailer without the need for extensive administration. The requirement for purchasing Sustainables is imbedded in the resource acquisition decisions of the electricity retailer. One challenge will be for the electricity marketer to predict how much power will need to be purchased from Sustainables over the year, given that load forecasts are not always accurate. Presumably, the SPS mechanism would have some flexibility for those marketers who underpredict demand, and hence underpredict the amount of Sustainables generation that is required. Enforcement of the requirements is straightforward because all electricity sales are transferred through the *Power Pool* and records are kept of the amount of generation from each technology type. Fines might be established to penalize those entities which consistently underproduce their requirement.

The SBC mechanism is also administratively simple to set-up, because the wires charge is added to customer bills in a similar fashion to a tax, and the fund is distributed according to the contracted price and the spot price of electricity. It does however require that an external organization administer the funds. Enforcement of the operation of Sustainables facilities is not necessary because they only get paid when they operate.

The EA mechanism is complex from an administrative or operational perspective, as every hourly transaction in the *Power Pool* for every generator includes an environmental charge. It is possible to automate the operation of the EA mechanism within the *Power Pool* computer system, but each new market entrant needs to be assessed in order to determine the "environmental adder" for that plant. Enforcement is automatic, within the operations of the *Power Pool*.

In summary, under the SPS policy, administrators only need to characterize technologies on a generic basis, as compliant or not compliant within the definition of Sustainables. In contrast, under the EA, each facility needs to be characterized and assessed a charge. The SBC requires an administrative body to manage the funds collected from the wires charge, and coordinate

5.2 Cost Effectiveness of Policies for Fostering Sustainables

This section provides an analysis of the cost-effectiveness of the policies to foster Sustainables. In order to undertake this analysis, each policy was adjusted to result in approximately the same installed capacity of Sustainables. The assessment of the cost-effectiveness of policy options is made using the capacity of Sustainables as the unit of comparison. The capacity of Sustainables developed must be made consistent among all policies as they may have different cost impacts associated with different levels of support for Sustainables. Also, in order to compare the policies on their financial merit, they should have the same policy design criteria. In previous chapters the design of the SBC mechanism prioritized technological diversity, while the SPS and EA mechanisms prioritized cost-effectiveness.

The parameters driving the EA and SBC scenarios were modified such that their outcome resulted in the same market penetration of Sustainables as the SPS scenario, approximately 10% of the capacity by the year 2025. Also, the technology mix for the SBC mechanism was changed to adopt only the most cost-effective technologies and not focus on fostering a diversity of technologies. A direct extension of the capacity of Sustainables is the annual energy production from Sustainables and the CO_2 emissions reduced as a result of the deployment of Sustainables. The cost-effectiveness of a policy to reduce CO_2 emissions is also included in this section.

The results from these simulation runs are not presented elsewhere in this study because they diverge from the magnitude of mechanisms that have been proposed or implemented in other jurisdictions. Thus, there is a tradeoff with the design of the policies between their cost-effectiveness and political feasibility.

5.2.1 Analysis of Scenario Parameters Adopted

As previously mentioned, the SBC and EA mechanisms were modified so that they result in the same installed capacity of Sustainables by 2025 as the SPS policy. Table 5.5 summarizes the policy parameters employed for this analysis.

Policy (and parameter name)	Parameter Value	B.C. Capacity of new Sustainables in 2025 (megawatts)	Alberta Capacity of new Sustainables in 2025 (megawatts)
SPS (portfolio requirement)	10% by 2025	1,746	1,349
SBC (wires charge)	10% of the PSP	1,755	1,114
EA (CO ₂ tax)	\$25 / tonne	1,596	1,312

	Table 5.5 - Policy	Parameters Emplo	yed for Cost-Effectivene	ss Analysis
--	--------------------	------------------	--------------------------	-------------

The size of the System Benefits Charge, included on all wholesale purchases, was increased from 3% to 10%, the former being the value employed in the previous chapters. Also, the payout from the SBC fund to Sustainables was changed so that only the most cost-effective Sustainables are supported. In previous chapters, the policy supported solar, wind, tidal, small hydro and biomass technologies, whereas for this analysis, biomass was the predominant technology supported. Table 5.6 lists the revised technology portfolio.

Technology Type	Share of Capacity in B.C.	Share of Capacity in Alberta
Biomass Gasifier Turbine	50%	50%
Wastefuel Cogeneration -biomass	25%	25%
Wind Farms	15%	25%
Small Hydro (high cost)	10%	0%

Table 5.6 - Technology Portfolio for the Cost-Effective SBC Scenario

The EA mechanism was adjusted by increasing the CO_2 tax to a level that would raise the installed capacity of Sustainables to about 1,750 MW in B.C., and 1,350 MW in Alberta, to match the 10% SPS mechanism. Several different tax levels were simulated (\$25,\$30,\$35,\$40) and the results were varied. The \$25/tonne level was selected because it resulted in the installed capacity of Sustainables reaching the target level in Alberta. In B.C., a \$40/tonne CO_2 tax is necessary to raise the installed capacity of Sustainables to the 1750 MW level. By raising the tax to a higher level in Alberta, for example to the \$30/tonne level, the installed capacity of Sustainables far exceeds the target value, resulting in 2,316 MW being developed.

5.2.2 Market Penetration of Sustainables

The market penetration of Sustainables in terms of installed capacity is approximately equal for all three policies, as their parameters were selected to achieve that objective. The values are shown in Figures 5.3 and 5.4.

Variation of the parameter for the EA mechanism (CO₂ tax in \$/tonne) has some interesting results related to the installed capacity of Sustainables. As the value of the tax was raised from \$20 / tonne (the value employed in previous chapters) to \$25 / tonne, the installed capacity of Sustainables went up from 1337 MW to 1596 MW in B.C.. However, when the tax was raised to \$30 / tonne, the installed capacity of Sustainables stayed approximately constant, reaching 1592 MW, showing a non-linearity between the level of CO₂ tax and the installed capacity of Sustainables. This is due to the fact that the additional \$5 / tonne tax (from \$25 to \$30) does not sufficiently cover the financial premium required to bring on an additional share of Sustainables

capacity, given that the most cost-effective Sustainables were already exhausted. When the tax was raised to \$40 / tonne, the capacity exceeded the 1750 MW target.

Figures 5.3 through 5.7 illustrate the installed capacity and market share of Sustainables in B.C. and Alberta under the revised scenarios. The installed capacities reach about 1750 MW by 2025 for B.C., and 1350 MW for Alberta. The key difference between the scenarios is the timing of development. Both the SPS and EA scenarios result in a gradual increase in the market share of Sustainables. In fact, the SPS demonstrates a perfectly linear growth curve because it is designed to do that. In contrast, the SBC demonstrates a sharp increase in the first year, with subsequent increases in 2005 and 2015. The percentage market share of Sustainables generation is consistent with the installed capacity figures, except that the SBC has a lower than expected value because of the low capacity factor of the wind technologies which are fostered under the policy.



Figure 5.3 - B.C. Installed Capacity of Sustainables (MW)

B.C. - Market Penetration of Sustainables



Figure 5.4 - Alberta Installed Capacity of Sustainables (MW)



Figure 5.5 - B.C. Market Share of Generation from Sustainables (percent)

B.C. - Market Share of Sustainables



Figure 5.6 - Alberta Market Share of Generation from Sustainables (percent)

5.2.3 Financial Impacts - Total Expenditure on Electricity Sales and Impacts on Spot Price

The cost premium over BAU for each policy in millions of dollars is illustrated in Figures 5.7 and 5.8, for B.C. and Alberta respectively. Other than the costs of the EA mechanism for Alberta, all of the costs are approximately equal in 2025, as the level of support for Sustainables is approximately equal. The EA is more expensive in Alberta because of the large capacity of coal power in the province which produces a high environmental charge. The charge results in a change in the *Power Pool* generating mix, bringing in more expensive resources such as biomass instead of coal. The cost premium does not include the value of the environmental charge as it is rebated off of customers' bills. The cost premium of the more expensive resources exceeds the value of the rebate by about \$200 million in 2025.

The SBC shows predictable cost increases as Sustainables capacity is brought on line, and a decrease in 2020 when the 25 year contract with Sustainables from 1996 expires¹¹. It is important to note that the SBC costs on consumers are distributed equally over the entire simulation period through a wires charge that is collected on all electricity sales. However, the costs borne by society, the allocation of investment capital into the electricity market, occur in three distinct periods when the contracts are established with Sustainables producers.

The distribution of costs over the 30 year simulation period varies significantly, and this becomes the major point of comparison between the policies from a risk management perspective. The SBC, by design, forces a large proportion of the investment costs to the beginning of the simulation period, with all that investment capital subject to risk of future increases in the cost of capital. In contrast, the required capital for the SPS mechanism is gradually increased as the required market share from *Sustainables* increases.

11

In the year 2020, a wind technology that was developed in 1996 has paid off all capital costs, and is outside of the original investment contract. As such, the facility would compete in an open market for renewables, with its value being equated to the value of a new wind facility in that year, including resultant capital cost reductions between those years.





Figure 5.8 - Alberta Policy Cost Premium over BAU for each Policy (million \$1995)

The costs of fostering Sustainables under each policy mechanism, relative to the BAU scenario, are outlined in Tables 5.7 and 5.8. These values were determined by dividing the policy cost premium by the total capacity of new Sustainables developed.

Table 5.7 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - B.C.

Scenario	2010	2025
SPS	\$45,612	\$7 5,101
SBC	\$ 92,365	\$7 5,969
EA	\$100,686	\$8 3,593

Table 5.8 - Cost Premium of Fostering Sustainables Development (\$1995/MW of Sustainables) - Alberta

Scenario	2010	2025
SPS	\$78,238	\$80,104
SBC	\$105,302	\$ 80,587
EA	\$192,093	\$153,644

The results indicate that the SPS and SBC mechanisms have similar costs in the year 2025, but the SBC has a dramatically higher cost in 2010. This is because the policy fostered a large capacity of Sustainables in the first year of the simulation which has to be paid off. Also, the costs in 2025 do not include a large portion of the original capital costs for the facilities built in 1996 as it was paid off in the year 2020. In contrast, the SPS demonstrates a gradual cost increase as the capacity of Sustainables is increased. The costs of the EA mechanism are significantly higher because it affects the operation of existing generation facilities, in addition to the investment patterns of the market, with resultant reductions in CO_2 emissions.

A cost-effectiveness indicator for the energy production from Sustainables is presented in Tables 5.9 and 5.10, indicating the price premium for Sustainables production. These costs are reflective of the cost premium of Sustainables over the average electricity cost.

 Scenario
 2010
 2025

 SPS
 \$ 0.007 / kWh
 \$ 0.010 / kWh

 SBC
 \$ 0.014 / kWh
 \$ 0.011 / kWh

 EA
 \$ 0.016 / kWh
 \$ 0.012 / kWh

Table 5.9 - Cost Premium for Sustainables Production (\$1995/kWh from Sustainables) - B.C.

 Table 5.10 - Cost Premium for Sustainables Production (\$1995/kWh of Sustainables)

 Alberta

Scenario	2010	2025
SPS	\$ 0.011/kWh	\$ 0.013 / kWh
SBC	\$ 0.015 / kWh	\$ 0.011 / kWh
EA	\$ 0.028 / kWh	\$ 0.021 / kWh

The cost premium over the BAU for every kilowatt-hour of Sustainable electricity ranges from 0.7 to 1.5 cents, except for the EA, which has costs as high as 2.8 cents. Both the SPS and SBC mechanisms demonstrate comparable costs in the year 2025, with the SBC incurring higher costs in 2010 because of the higher installed capacity of Sustainables in that year under the original 1996 development contract. The EA demonstrates higher costs for the same reasons previously mentioned.

The wholesale and peaking spot prices of electricity for B.C. and Alberta are illustrated in Figures 5.9 through 5.12.

Figure 5.9 - B.C. Wholesale Spot Price (\$1995/kWh)



B.C. Baseload Wholesale Spot Price


Figure 5.10 - B.C. Peaking Spot Price (\$1995/kWh)

Business as Usual Sustain. Port. Stand. Sys. Benefits Charge Environmental Adders









The spot prices differ from those in Appendices B and C (Figures B.3, B.4, C.3 and C.4) in a predictable fashion. The size of the SBC is higher for the analysis in this section, and as such, the gap between the BAU spot prices and the SBC spot prices is larger. Also, the size of the CO_2 tax is higher, and as such the spot price is higher. Again, the SPS and BAU demonstrate similar spot prices.

5.2.4 Annual Carbon Dioxide Emission Abatement Cost

 CO_2 emissions associated with the four scenarios are listed in Table 5.11. The emissions associated with the SPS and SBC are approximately the same, as the policies equally foster Sustainables to play a role in the market. The EA has less emissions, due to the inclusion of a CO_2 tax in all electricity sales of \$25/tonne, affecting not only investment decisions, which translates to increased investment in Sustainables, but also operating decisions, which affects existing technologies. In Alberta, the EA policy forces a reduction in emissions over the simulation period.

Scenario	B.C. Emissions (megatonnes)	Alberta Emissions (megatonnes)
BAU	23.065	43.196
SPS	18.687	40.303
SBC	18.760	40.425
EA	17.692	32.938

Table 5.11 - CO₂ Emissions in 2025 - B.C. and Alberta (megatonnes)

An important policy analysis indicator is the CO_2 abatement cost of a measure. This is the total cost associated with the reduction of one tonne of CO_2 emissions that results from the adoption of a policy measure. The costs are illustrated in Figures 5.13 and 5.14. In B.C. the abatement costs in 2025 are almost the same for all policies, ranging from \$25 - \$30/tonne. However, only the SPS mechanism kept the costs consistent over the simulation period. In Alberta, the SPS mechanism demonstrates an abatement cost of between \$30 and \$40/tonne, remaining more or less stable. The SBC costs range from \$30 - \$45/tonne, with a sharp decrease in 2020 when the technology costs are reduced. The EA mechanism stays flat at \$20/tonne approximately at the level of the tax.



Figure 5.13 - CO₂ Emissions Abatement Cost Curve (\$1995/tonne)- British Columbia

----- Sustain. Port. Stand. ----- Sys. Benefits Charge ----- Environmental Adders **B.C. CO2 Abatement Cost Curve**

105



-----Sustain. Port. Stand. ----Sys. Benefits Charge ---- Environmental Adders

Figure 5.14 - CO₂ Emissions Abatement Cost Curve (\$1995/tonne)- Alberta

106

Alberta CO2 Abatement Cost Curve

5.3 Summary of Analysis

The following table provides an overview of the analysis results, organized according to the six evaluative criteria outlined at the beginning of this chapter. A numeric indicator is provided by each policy mechanisms for each of the evaluative criteria. The indicators are not intended to be added to determine the optimal mechanism, but rather are intended to provide a basis for comparison between mechanisms within one evaluative criterion. In some cases, the indicator differs between Alberta and B.C. due to differences in the electrical systems in those jurisdictions.

Criterion	Policy Mechanisms		
	SPS	SBC	EA
Maximize Installed Sustainables Capacity	1	4	2
Maximize Sustainables Technology Diversity	4 - B.C. 2 - Alberta	1 - B.C. 2 - Alberta	4 - B.C. 3 - Alberta
Maximize Sustainables Cost Reduction	5 - B.C. 2 - Alberta	1	5 - B.C. 3 - Alberta
Minimize Financial Impacts	1	4	2 - B.C. 5 - Alberta
Minimize CO ₂ Abatement Cost	2	5	1
Administrative and Operational Simplicity	1	3	3

Table 5.12 -	Summary	Matrix	of Policy	/ Anal	lysis
--------------	---------	--------	-----------	--------	-------

(1 is for high achievement, 5 for low achievement)

For detailed comments on the basis for each of the indicators listed above, see the text in Sections 5.1 and 5.2. In summary, the SPS approach tends to maximize the installed capacity of Sustainables because it is designed to support the most cost-effective technologies or resources. The SBC mechanism, by design, enhances the diversity of Sustainables supported, and maximizes the cost reduction of emerging Sustainables due to that diversity of support. The SPS minimizes financial impacts due to a phasing-in of the requirement, and a continuous cost-minimization pressure on the market. The EA minimizes the CO_2 abatement cost because the policy mechanism is set-up as a CO_2 tax, thus driving the market to reduce emissions at the least cost. Finally, the

SPS is the most simple to operate and administer because it relies on the market to achieve the policy objectives.

5.4 Analysis of Uncertainty and Accuracy of the CEMPA Model

5.4.1 Uncertainty Analysis

Simulation modelling involves uncertainty. The modelling exercise can attempt to minimize the impacts of that uncertainty through careful selection of model structure or parameters, and/or by doing a sensitivity analysis of key parameters, assessing the robustness of model results under a wide range of conditions. In this study, the former approach was undertaken - to attempt to minimize the impacts of uncertainty, rather than doing an extensive sensitivity analysis. However, the modelling results have been compared with published information from similar studies to demonstrate accuracy.

The explicit approach undertaken to minimize the impacts of uncertainty included the following:

- The utilization of a high discount rate thus evaluating policy options from a conservative financial standpoint. By adopting a high discount rate, technology costs reach their upper limits, particularly among capital intensive technologies such as Sustainables.
- Application of generic technology parameters into the model, thus averaging out all the uncertainty associated with costs, capacity factors, heat rates, emissions, etc. Had an effort been made to characterize specific generation facilities, then uncertainty considerations would have been more prevalent.
- Using conservative estimates of fuel costs for competing technologies.
- Utilization of conservative resource potential parameters for emerging technologies.
- Defining driving policy parameters according to realistic levels being applied or proposed in other jurisdictions. The level of portfolio requirement under the SPS policy is on the lower-range of what is being proposed in the U.S., as is the carbon tax applied for the EA. The SBC is consistent with what is proposed in other jurisdictions. The parameters that were adopted for the EA and SBC mechanisms under the analysis in Section 5.2 are not necessarily realistic, but were selected in order to compare with the outcome of the SPS.

5.4.2 Accuracy of the Model

Three recent studies by the Canadian Electricity Association, the Tellus Institute, and BC Hydro, can be used to assess the accuracy of the CEMPA results, at least partially. Those studies include.

- 1. The Canadian Electricity Association study, Greenhouse Gas Management and the Canadian Electric Utility Industry (CEA, 1997).
- 2. The Tellus Institute study that looks at the costs and benefits of a national Renewable Portfolio Standard in the U.S., similar to the CEMPA SPS scenario (Bernow.et.al., 1997 & Figure 1.1).
- 3. An overview of CO₂ abatement costs, prepared by BC Hydro.

The first study demonstrates that B.C. CO_2 emissions will increase from 2.5 million tonnes in 1995 to 6MT in 2015 under the "base case" scenario. In comparison CEMPA predicts that emissions will rise from about 1MT to 10MT during that period (BAU). For Alberta, the CEA study predicts emissions will drop from 45MT to 39MT during that period, while CEMPA predicts they will remain flat at 39MT to 40.5 MT. There are 2 explanations for this divergence.

- The production estimates in GWh are significantly different than those of the CEA, with the CEMPA B.C. production ranging from 65,000 to 86,000 GWh per year over the simulation, and the CEA study range from 48,000 to 67,000 GWh. CEMPA utilizes an NRCan estimate for capacity demand, and a BC Hydro *load duration curve* to estimate annual production, and is in-line with the NRCan energy forecasts. For the CEMPA Alberta model, the difference was not as great.
- 2. The utilization of a higher discount rate in CEMPA (12% in CEMPA, 8% in CEA) results in the greater deployment of low capital cost fossil fuel technologies. For example, CEMPA predicts a considerable market penetration of simple-cycle gas turbines, which have lower capital costs than combined-cycle turbines. The combined-cycle turbines are more efficient, resulting in reduced emissions if they are used as peaking plants instead of simple-cycle technologies.

The second study (Bernow et.al., 1997) quantifies the costs and benefits of a policy similar to the SPS, but for the entire U.S. market. The results of this study were illustrated in Figure 1.1, showing that the rate impact of an SPS is close to zero, which is consistent with the results from CEMPA (Figures 5.9, 5.10, 5.11, 5.12). The Tellus study shows that the cost of a renewable

energy credit is about 1.03 to 1.2 US cents / kWh, and CEMPA results show a magnitude of 0.7 to 1 cent in B.C. (Table 5.9), and 1.1 to 1.3 cents in Alberta (Table 5.10), indicating a close resemblance between CEMPA and the Tellus Institute study (Ibid.).

Finally, BC Hydro produced a summary of CO_2 mitigation cost estimates which showed a range of about \$1/tonne and \$155/tonne, with the majority of values in the \$10-\$40 range. These figures are indicative of the cost of a policy to reduce CO_2 emissions and can be compared with the CO_2 emission abatement costs curves presented in Figures 5.13 and 5.14. CEMPA results indicate an average cost of \$20-\$40 per tonne for all scenarios, in-line with the estimates from BC Hydro.

5.4.3 Discussion on Limitations of Model

As briefly outlined in Chapter 2, CEMPA is only one of many different types of models, most of which are much more complex. Due to the scope of this study, and limitations on the modelling methodology, the results from CEMPA will inherently be less accurate than some other models. Those limitations include (including ones not related to those other models):

- CEMPA characterizes energy demand with the *load duration curve*, rather than a time-based function which provides real-time demand information to the model, representative of consumer demand. Also, there are price elasticity feedbacks between demand periods which are not represented such as the impact of a high electricity price in one period on the demand in the same period on a future date. This impacts on the accuracy of CEMPA's representation of market trading functions.
- CEMPA does not characterize the limitations of the transmission network, or attach a time or congestion-based charge for transmission wheeling, but rather applies an average transmission cost for all transactions. However those expenses may become prevalent under competitive markets as the number of transactions increase, and transmission constraints will start to dictate the location and types of technologies that will be developed.
- CEMPA does not characterize electricity trade between jurisdictions, which is expected to increase under competitive markets.
- CEMPA assumes that all generation facilities fall under a generic class of technology, yet in reality there is a wide divergence of costs and performance of facilities that fall under the same

generic class. The CEMPA "variation in investment behavior" function partially meets this concern.

- The characterization of hydroelectric facilities in CEMPA is unrealistic. In reality, they are energy-poor and capacity-rich, able to meet peaking loads at almost all times. CEMPA applies an annual average capacity factor to hydro facilities, thus limiting their ability to meet annual peak loads. This results in CEMPA over-building capacity in B.C. to meet peak loads, in order to compensate for the low annual average capacity factor of existing hydro facilities.
- Under the three sustainability scenarios (SPS, SBC, EA), CEMPA first runs the business-asusual scenario to determine cost-effective investment in technologies, and then characterizes the sustainability policy after the fact, adding Sustainables capacity to the mix. This is done to keep the model complexity to a minimum. As a result, the characterization of declining capital costs among emerging technologies in CEMPA resulting from the sustainability policies does not feed back to the regular investment model, thus possibly under-predicting the investment in Sustainables. For example, as a result of the SBC mechanism in Alberta, wind technologies become competitive with several natural gas technologies in later years of the simulation, and should offset some natural gas technology development, based on strict financial criteria. CEMPA does not capture that additional investment resulting from cost reductions. The model that captures the cost-effective investment was executed prior to the SBC mechanism model where the cost reductions occurred.
- Retirement of existing technologies is based on a straight-line function in percent retired per year, rather than reducing their capacity according to actual plant closures.

6. CONCLUSIONS

6.1 Overview

Sustainables will not be competitive under competitive electricity generation markets, and will essentially achieve zero market penetration if no specific instruments favoring them are applied in B.C. or Alberta. There are three reasons for the failure of Sustainables to achieve an increased market share: (1) the failure of governments to implement regulations or other mechanisms which reflect environmental externalities in decision making; (2) the low capital and operating costs of natural gas turbine technologies; and (3) the high capital cost / low operating cost characteristics of Sustainables which is especially problematic as the market becomes dominated by private investors with higher discount rates.

"Business as usual" market development is likely to result in negative environmental impacts through increased greenhouse gas and other emissions, which could have financial risk implications in the future if environmental standards are established for greenhouse gas emissions and other pollutants. The study demonstrates that greenhouse gas emission will climb from 1 to 23 Megatonnes over the next 20 years in B.C., and will increase slightly in Alberta from 39 to 43 Mt, currently the highest in Canada for the electricity sector.

Sustainables can be effectively fostered under competitive electricity markets. The three policy mechanisms that were analyzed in this study resulted in significant increases of Sustainables' capacity. Of the three, the Sustainables Portfolio Standard mechanism was most effective at fostering Sustainables within the policy objectives. The characteristics of this policy mechanism include.

- Large increases in installed capacity of Sustainables, reaching 1750 MW in B.C. and 1350 MW in Alberta by the year 2025.
- Zero impact on the spot price of electricity.
- Gradual electricity sector cost impacts as it is phased-in over the 30 year study period.
- Cost-effective support for Sustainables, averaging about one cent per kilowatt-hour of Sustainables electricity produced, and between \$45 and \$80 per kW of Sustainables capacity installed.

- Cost-effective CO₂ emission abatement costs, ranging from an average of \$25 / tonne in B.C. to about \$35/tonne in Alberta.
- Political acceptability and administrative simplicity.

The only failing of the SPS mechanism implemented in this study is that it does not result in a diversity of Sustainables technologies being developed, but rather only the most cost-effective ones - small hydro, biomass and in later years, wind. However, the policy could be set-up in such a way that requires electricity marketers to purchase a minimum share from various technology types, and hence a diversity of technologies would be fostered.

The SBC mechanism is cost-effective in the long run and can be easily designed to foster a diversity of technologies. Under the policy design implemented in this study, the majority of investment costs are borne in early years, and investments in Sustainables are made in block purchases. In order to reach the same market support for Sustainables as the SPS mechanism does, the wires charge needs to be at a politically unacceptable level of 10% of the PSP, or about 5% of the consumer rates for electricity. This far exceeds the magnitude that has been supported in other jurisdictions. For example, in the U.K., the wires charge is about 1.3%.

The EA mechanism results in cost-effective CO_2 emission abatement and good support for Sustainables, but has potentially significant impacts on the spot price of electricity, and has excessive financial implications in Alberta due to the large capacity of existing coal plants there. Also, the mechanism is administratively complex.

6.2 Future Work - What are the Next Steps?

Future modifications if similar analyses are to be undertaken include:

- The development of a more complex spot market sub-model, including trade between jurisdictions and a network analysis that assesses the costs of transmission line congestion.
- The inclusion of a greater number of competing technologies, to better represent the diversity of technologies available, and to specifically represent existing generation facilities.
- The development of a greater number of sustainability scenarios, to allow for a more detailed comparison of policy options, and to assess the sensitivity of the outcomes to a variety of factors.

7. REFERENCES

Alberta Energy. Moving to Competition - A Guide to Alberta's New Electric Industry Structure. Edmonton: Alberta Government, 1996.

Bernow, Steve; Dougherty, William; and Duckworth, Max. Quantifying the Impacts of a National, Tradable Renewables Portfolio Standard. In the Electricity Journal, May, 1997. pp. 42-52.

B.C. Hydro (BCH). Wholesale Transmission Services Application. B.C. Utilities Commission, 1996.

B.C. Hydro (BCH). Transmission Outages, BCH - Forced Outage Performance of Transmission and Station Equipment (1984-1992). Vancouver, 1993.

British Columbia Utilities Commission (BCUC). *Electricity Market Review - Report and Recommendations to Cabinet*. BCUC: Vancouver, 1995.

BCUC. Integrated Resource Planning (IRP) Guidelines. February, 1993.

Byrnes, Rahimzadeh, Baugh. Green Pricing - the Bigger Picture. In "Public Utilities Fortnightly', August, 1996. pp. 18-21.

Canadian Electricity Association. Greenhouse Gas Management and the Canadian Electric Utility Industry. Presented to the Government of Canada, Summer, 1997.

Canadian Electricity Association / NRCan. Electric Power in Canada 1995. published in 1996.

Canadian Electrical Association (CEA). Generating Unit Statistics for Canada. Table 1, Appendix B. Page 17. 1994.

CEERT - Centre for Energy Efficiency and Renewable Technologies. *Clean Power Journal*. Fall 1996 issue.

Drinnan, J.H.; and Spafford, K.R. A Stochastic Resource Planning Model. Presented at the CEA Spring Meeting on Generation and System Planning & Operation. 1987.

Dunn, William & Rossi, Mark. Practical Aspects of Restructuring. KEMA-ECC Inc. May 9, 1996. Submitted for publication in The Electricity Journal.

Energy Information Administration, U.S. Department of Energy. The National Energy Modelling System: An Overview. May, 1994.

Gorscurth, Helmut M. Design of and Experiences with the German Electricity Feed Law. ZEW, Mannheim. In the Seminar Proceedings, "European Seminar on Market Incentives to Renewable Energies", March 7-8, 1996. Lisboa, Portugal. pp. 69-74.

Hamrin, Jan -with William Marcus, Carl Weinberg and Fred Morse. Affected with Public Interest - Electricity Industry Restructuring in an Era of Competition. Washington: National Association of Regulatory Commissioners, Renewables Subcommittee of the Conservation Committee, 1994.

Henney, Alex and Crisp, Simon. Transmission Pricing, Constraints and Gaming in England and Wales - Lessons for California. Presented at the International Bar Association conference on Energy and Natural Resources Law. Oakland, California. June 17, 96.

Hunt, Sally and Shuttleworth, Graham. Competition and Choice in Electricity. Toronto: John Wiley and Sons, 1996.

IRP Report. Renewable Advocates Joust Over California Policy. February, 1996. pp. 15-18.

Jaccard, Mark; Bailie, Alison; and Nyboer, John. CO₂ Emission Reduction Costs in the Residential Sector: Behavioral Parameters in a Bottom-Up Simulation Model. In The Energy Journal, Vol.17, No.4, 1996.

Jaccard, Mark. Finding the Right Climate for Competitive Electricity and Natural Gas Markets. In Ecodecision, Winter, 1996. pp. 48-51.

Jaccard, Mark; and Makinen, Timo. Turning to Supply-Side Management: Estimating Industrial Cogeneration Potential. In Energy, Vol.19, No.2, pp. 237-249. 1993.

Johansson, Thomas; Kelly, Henry; Reddy, Amulya; and Williams, Robert. Renewable Energy: Sources for Fuels and Electricity. Washington: Island Press. 1993.

Kahn, Edward; Bailey, Shawn; and Pando, Luis. Simulating Electricity Restructuring in California: Interactions with the Regional Market. April 17, 1996.

Kelly, Henry and Weinberg, Carl. Utility Strategies for Using Renewables. In Renewable Energy: Sources for Fuels and Electricity. Washington: Island Press, 1993. Kirshner, Barkovich, Treleven, Walther. A Cost-Effective Renewables Policy Can Advance the Transition to Competition. In the "Electricity Journal", January/February, 1997. pp. 54-61.

Liu, Hao. Toward Reducing Green House Gas Emissions: Challenges to Canadian Electricity Supply Sector. Masters Degree Thesis, Simon Fraser University. 1995.

McGuigan and Moyer. Managerial Economics. West Publishing Company. 1989.

Dr. Catherine Mitchell. Renewable Energy - Policy and Costs. Presentation to the U.K. Parliamentary Renewable and Sustainable Energy Group, Feb. 19, 1996. Mitchell, Catherine. The Renewables NFFO - a review. In Energy Policy. Volume 23, No. 12, 1995, pp. 1077-1091.

National Association of Regulatory Utility Commissioners (NARUC). Promoting Environmental Quality in a Restructured Electric Industry. Prepared by the Tellus Institute. December 15, 1995.

Natural Resources Canada (NRCan). The Economic Characteristics of Large Scale Wind Energy Development in Alberta. 1996. Prepared by Nor'Wester Energy Systems Ltd.

Newbery, David M. Power Markets and Market Power. In Energy, Volume 16, no. 3, July, 1995, pp. 39-66.

Pape, Andrew. Hydroelectricity Pricing under Competitive Electricity Markets. Simon Fraser University, School of Resource and Environmental Management, 1996.

Pape, Andrew. Sustainable Electricity Technologies - Opportunities and Constraints within Competitive Electricity Markets. For 1996 Solar Energy Society of Canada Conference. 1996.

Power Pool of Alberta. Internet Web site (http://www.powerpool.ab.ca/news_releases/ releases.html#levy). Information about Power Pool "trading charge".

Rader, Nancy and Norgaard, Richard. Efficiency and Sustainability in Restructured Electricity Markets: The Renewables Portfolio Standard. In "The Electricity Journal", July, 1996, pp. 37-49.

Scott, Jeff, General Manager of The National Grid Group plc, U.K. Implementing Change: Post Reform Changes in the England and Whales System. Presented at the International Bar Association conference on Energy and Natural Resources Law. Oakland, California. June 17, 96. Tellus Institute. Recommended Values for Air Emissions in Massachusetts DPU Docket #89-239. Union of Concerned Scientists. America's Energy Choices. 1992.

United Kingdom (U.K.) Electricity Act. 1989.

.

United Kingdom Electricity Statistics - Summary Tables 1990-1996.

8. APPENDIX A - SPECIFIC METHODOLOGICAL DETAILS

8.1 Screening Curve Model

The specific equation that is applied to determine the intersection points of the least-cost technologies in the *screening curve* routine (Chapter 2, Section 2.3.5.1) is outlined in Equation A.1 below. Y is representative of the total cost of developing a technology, and is a function of X, the number of operating hours of that technology for a year. Intersection points are determined by equating the mathematical expressions for each of the two lines. Each intersection point translates to the number of operating hours for a technology.

Equation A.1 - Equation for the Optimization of Electricity Supply Resource Investment

Y = Intercept + (Slope* X)
 X = the number of operating hours in a year
 Y = cost of technology for a given number of operating hours (X)
 Intercept = Fixed Cost

Slope =
$$\left(\frac{(\text{Variable Cost} * X)}{X}\right)$$

$$Y = \left(\frac{(\text{Variable Cost} * X)}{X}\right) * X + \text{Fixed Cost}$$

$$Y = (Variable Cost * X) + Fixed Cost$$

Intersection at the Point where the line equations are equal $Y_1 = Y_2$

$$X_{\text{Intersection}} = \frac{(fixed_1 - fixed_2)}{(var_2 - var_1)}$$

Total Cumulative C	apacity (rom Ne	w and	Existin	g Techi	vologies	S - BC N	Nodel -	BAU						11
		KA	<u>Shere of mar</u>	test for non-t	enst cost t 2	¥	tice threshol	ki over leest	2 Auj 1908	2					
Technology Name	865 1	191	HA!	Ī	2000	201	202	202	M	205	Non	7467	ž	ž	
Mey seb elocia Mew	251.75	385.96	560.34	730.87	898.12	1008.53	1117.83	1226.04	1333 20	1439.31	1566.68	1736.78	1883.60	2028-11	217351
Simple cycle oli (b) Anologies	38		100	108	000	8	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
Combined cycle pil	280	200	10.00	5 Å	5 M C	130.20	1000			1/82.24	2021.48	7249.70	2476.89	2703.08	2928.29
Small hydro - Optimal	300.00	300.00	300.00	30.00	30.00	3000	3000	300 00	30.00						
Blomass steem	0.00	000	000	0.00	0.00	0.00	0.00	0.00	000	0.00	0.00	000	080	0.00	
Coel - Pressurized Fluidized Be	28	88	88	88	0.00	0.00	0.00	0.00	0.00	0.00	00	0.00	0.00	0.00	0.00
	25	88	8 S				0.00	800		0.00	0.00	0.00	0.08	0.00	0.00
Lama Hivim	D R C	9 8 9 8 9	- 2 3 3	- 0 - 0 - 0	- 100.04	0.00	194.35	213.79	233.09	252.25	279.30	306.16	332.69	359.45	305.84
Cost gasifier turbine	8	88	8	8	88	88	88		38	200	38	38		88	200
Small Hydro - Iower grade	0.00	0.00	00	0.00	0.00	00	8	8	8	0.00	88	88	88	88	500
Waste fuels cogen	0.00	00	8	0.00	0.00	0.00	0.00	0.00	0.0	00	0.00	0.00	0.00	0.00	0.0
Wind Generators	0.00	0.00	0.00	0.00	000	0.00	0.00	0.00	0.00	0.00	0.00	0.00	00	0.00	.0
Tidal Davies	200	88	200	88	88	88	88	1.00	00	00	0.00	0.0	0.00	0.00	0.00
Paraholio, Treinh Solar, Douar		30	3	88	8	28	200	0.00	0.00	0.00	0.00	0.00	0.00	000	08
Solar PV	0.00	0.00	0.00	0.00	88	88	000	0 0 8 8	0.00	000	88		0,00		
Natural Gas Existing	1020.69	1010.48	1000.38	990.37	980.47	970.67	960.96	961.35	941.84	832.42	923.09	913.86	804.72	895.66	886.72
		0.00		0.00	0.8	0.00	0.00	0.00	000	0.00	0.00	0.00	0.00	0.00	0.00
Hydro	11110.77	10999.66	10689.67	10780.77	10672.96	10566.23	10460.57		10252 40	10149 AS		0047 0n	110.53	110.70	105.17
Nuclear	000	000	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.0	0.00	0.00	000	0.00	0.00
Biomass	526.69	521.41	518.20	511.04	505.83	500.87	495.86	490.90	485.99	481.13	476.32	471.56	466.84	462.17	457.55
Total New Sustainables	300.00	300.00	300.00	300.00	300.00	300.00	300.00	00.000 //eer/er	00 00E	300.00	300 m			300.00	
Technology Name	2011	2012	2013	2014	2015	2016	2017							ž	Į
Simple cycle gas New	2323.66	2475.31	2628.50	2783.29	2939.73	3097.85	3257.71	3418.35	3582 83	3748.19	3913,99	4080.27	4247.08	411.49	4582.98
Combined cycle gas	3162.38	3396 66	3637.19	3878.04	0.00	4362.00	1815 IR	4866 AD	0.00	0.00	0.00	0.00	0.00	0.00	0.00
Combined cycle oil	0,00	000	0,00	0.00	0.00	00	0.00	0.90	8	0.00	0.00	80	80	8	0.00
Small hydro - Optimal	300.00	00.00	00 000	300.00	300.00	300 00	300.00	300.00	300.00	300.00	300.00	300.00	300.00	300.00	300.00
Cost - Pressurized Fluidized Be	88	88		8	88			88	0.00	28	200	88	28	88	200
Biomess gasifier turbine	0.00	8	00	0.00	8	8	00	88	88	000	0.0	000	88	88	88
Gas cogen retrofts	413.29	441.00	468 98	497.24	525.79	554.63	583 78	613.24	643.02	673.14	703.59	734,39	765.55	797.06	1000.00
Large Hydro	88	38	28	88	88	88	88	200	80	8	0	0.00	0.00	8	0.00
Small Hydro - lower grade	200	88	88	88	8	88	88	0.0	0.0	88	0.00		88	88	0.00
Wasta fuels cogen	0.00	8	20	8	0.00	000	0.00	0.00	00	8	000	0.0	0.00	0.0	0.00
Find Cell Commenter	88	88	0.00	88	88	88	88		88	8	0.00	0.00	8	8	0.00
r uei Ceil Cuyernerator Tidal Power	88	88	88	38	38	38	88		88	200	88	88	200	200	200
Parabolic-Trough Solar-Power	0,00	000	0.00	0.00	8	88	88	88	8	88	0.0	0.00		88	000
Solar PV	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.8	0.00	0.00	0.00
Coal Anologies	000	0.00	000	000	000	0.00	826.48	818.ZZ	810.03	801.83	783.91	785.98	778.12	770.33	762.63
2	99.91	94,91	90.17	85.86	81.38	77 31	7344	8 3 2	66.28	62.97	59.82	56,83	53.99	51,29	48.72
Hydro	9655.91	9460.35	9365.75	9272.09	9179.37	9087.58	8996.70	8906.73	8817.67	8729.49	8642.19	8555.77	8470.21	8385.51	8301.66
Biomass	452 69	1000		130.53	135 13	110 20	0.00	38	000	8	0.00		0.00	0.00	0.00
Total Suppty	17,186.0	17,487,9	6.M4/11	13,107.6	18,225,9	10,749.9	F620'61	194155	L 197.61	20.105.2	10459 A	20.020.07	401.01	1 620 E	70.050
Total New Sustainables	300.00	300.00	300.00	300.00	300.00	300 00	300.00	300.00	300.00	300.00	300.00	300.00	300.00	300.00	300.00

Table B.1 - Total generation capacity (MW) by technology type - BAU

9

Technology Name 1444 1447 1447 1448 1447 1448		Negalitati s	-											
Single gring by 131.75 Set By Model State 170 by	Cochronology Name	1996	1997	1998	1998	2000	2001	7967	2003	2004	2004	2000	2047	7
Severe grafe for the definition of the second set of the second set of the second set of the definition of the second set of the second se	Strain avair ann	251.75	365.00	560.34	730 97	899.12	1008.53	1117.03	1229.04	1333.20	1430.31	1500.00	1730.78	100
Constant graph is in the set of	inche codit all (b)	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0 00	0.00	0.00	0.00	0.00	
Constant System 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Combined systems	30.03	107.76	307.21	808 04	814.05	837.30	1058 87	1178.07	1287.81	1414.61	1500.00	1778 84	185
Same Series Developer - Cubined 200 00	Cambined pycie oli	0.00	8 00	0.00	D 00	0.00	0.00	0.00	0 00	0.00	0.00	0.00	0.00	
Name above TO 00 17:28 17:00 116:33 200 MB 344 (0 197 MB 137:24 377:85 378:34 378:34 378:34 378:34 378:34 378:34 378:34 378:34 378:34 378:34 378:34 378:34 378:34 378:34 378:34 378:34 377:35 300:10 000 <td>Small Judro - Californi</td> <td>300 00</td> <td>300.00</td> <td>300.00</td> <td>300 00</td> <td>300.00</td> <td>300.00</td> <td>300.00</td> <td>300 00</td> <td>200.00</td> <td>300.00</td> <td>300.00</td> <td>00.00t</td> <td>30</td>	Small Judro - Californi	300 00	300.00	300.00	300 00	300.00	300.00	300.00	300 00	200.00	300.00	300.00	00.00t	30
Card. Transmission flattant basel COO COO <t< td=""><td>ligned steen</td><td>0.00</td><td>37.24</td><td>78.06</td><td>116 42</td><td>158.33</td><td>200.60</td><td>244.10</td><td>287.80</td><td>312.47</td><td>377.63</td><td>379.34</td><td>378.54</td><td>37</td></t<>	ligned steen	0.00	37.24	78.06	116 42	158.33	200.60	244.10	287.80	312.47	377.63	379.34	378.54	37
Name of parts Color Color <thcolor< th=""> Color</thcolor<>	Cost, Press and Ridland Red	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	
Ges segment servers 12.30 ed. 27 ed. 27 ed. 27 ed. 27 ed. 28 116 did 174.77 114 33 213.78	inness meller brbins	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	44.08	60 52	13
Large News COO COO <thcoo< th=""> COO <thcoo< th=""> <thcoo< td=""><td>The course retrains</td><td>42 20</td><td>65.07</td><td>85.26</td><td>125 25</td><td>155.04</td><td>174.77</td><td>164.35</td><td>213.20</td><td>213 08</td><td>252.25</td><td>279 30</td><td>308.18</td><td>11</td></thcoo<></thcoo<></thcoo<>	The course retrains	42 20	65.07	85.26	125 25	155.04	174.77	164.35	213.20	213 08	252.25	279 30	308.18	11
Configueire verben 000	area làdra	0.00	0.00	0.00	6 00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	
Same Turber 0.00	Coal coalities testates	0.00	0.00	0.00	0.00	0.00	0.00	0.00	8 00	0.00	0.00	0.00	6 00	
Neurite Langen Dial Dial <thdial< th=""> Dial Dial</thdial<>	Count Martin - Journe results	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	
Weid Gumminford 000	Alasia Aula sociat	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	
Fuel Cate Communicator 0.00 0.0	Admit Generators	0.00	0.00	0.00	0.00	0.00	0.00	0.00	6.00	0.00	0.00	0.00	0.00	
Tail Triver 0 <th0< th=""> 0 <th0< td=""><td>Fuel Cell Conservator</td><td>0.00</td><td>0.00</td><td>0.00</td><td>0.00</td><td>0.00</td><td>0.00</td><td>0.00</td><td>0.00</td><td>0.00</td><td>0.00</td><td>0.00</td><td>0.00</td><td></td></th0<></th0<>	Fuel Cell Conservator	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	
Participant France/ Science Prevent 0.00	Tidal Prever	0.00	600	0.00	000	0.00	0.00	000	0.00	0.00	0.00	0.00	9.00	
Same D.00 D.00 <thd.00< th=""> D.00 D.00 <thd< td=""><td>Parabolo, Trough Solar, Power Plant</td><td>0.00</td><td>0.00</td><td>0.00</td><td>8 00</td><td>0.00</td><td>0.00</td><td>0.00</td><td>6 00</td><td>0.00</td><td>0.00</td><td>0.00</td><td>0.00</td><td></td></thd<></thd.00<>	Parabolo, Trough Solar, Power Plant	0.00	0.00	0.00	8 00	0.00	0.00	0.00	6 00	0.00	0.00	0.00	0.00	
Natural bar. 1070 bit	Solar PV	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	
Card 0 00 <th< td=""><td>Wetural Gee</td><td>1020 00</td><td>1010 44</td><td>1000.30</td><td>980 37</td><td>100 47</td><td>970 97</td><td>940 65</td><td>61.35</td><td>NIN .</td><td>61242</td><td>623.00</td><td>613.00</td><td></td></th<>	Wetural Gee	1020 00	1010 44	1000.30	980 37	100 47	970 97	940 65	61.35	NIN .	61242	623.00	613.00	
CP 213 mis 204 mir 1 Het P2 1 Het P7 1 Het P2 1 Het P7 1 H	Ceel I	0 00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	
Single phase 11110 77 1086 88 10980 87 10780 77 1087 28 1058 87 1022 20 1086 88 1000 00 0		215.05	204 07	164 82	104 00	175 85	106 87	158 52	150 80	143.07	135.01	129.12	122.00	10
Nuclear 0 00	indro I	11110.77	10000.00	10000 07	10700 77	10672 06	10546 23	10440.57	10355 84	10252.40	10140.00	10048 38	8647.00	984
Dimension \$220,00 \$211,41 \$110,40 505,87 500,07 445,84 440,10 445,84 440,112 417,823 417,843 11,844 11,814 <t< td=""><td>Autom I</td><td>0.00</td><td>0.00</td><td>0.00</td><td>0 00</td><td>0 00</td><td>0.00</td><td>0.00</td><td>0 00</td><td>0.00</td><td>0.00</td><td>0.00</td><td>0.00</td><td></td></t<>	Autom I	0.00	0.00	0.00	0 00	0 00	0.00	0.00	0 00	0.00	0.00	0.00	0.00	
Telai Bugaby TELART 7 TELE 77	Norme es .	524.00	521.41	\$16.20	511.04	505.83	500.87	495.06	480.90	405.00	401.13	476.32	471.50	40
Tetal New Swal abulates 300.00 337.26 376.06 418.42 458.33 500.80 544.10 567.86 622.47 677.85 723.43 768.86 Singlis rights gate 2223.86 2473.31 287.85 278.27 289.87 3007.85 3257.71 3418.36 3542.85 3740.66 3666.85 4602.81 Singlis rights rights rights 2500.06 2665.84 2873.02 3002.03 3224.81 3448.56 3644.00 3600.00 300.00 300.00 300.00 300.00 300.00 300.00 300.00 300.00 300.00 300.00 300.00 300.00 300.00 300.00 300.00 300.00 300.00 300.00 300.00	falii Biyyly	13,506.77	13,712.47	\$4,828.74	14,345.75	14,000.54	14,828.12	14,001.18	15,155.87	16,218.87	15,463.15	11,700.30	18,648.63	11,321
Technology Name 2011 2012 2013 2014 2015 2016 2017 1015 2019 2020 2021 2022 Stripts cycle gis 2.222 BI 3.475.31 2.272 BI 3.475.31 2.272 BI 3.475.31 2.275 BI 3.277.71 3.18 35 3.362 EI 3.740 BI 3.741 BI<	folal New Municipables	300.00	337.26	376.08	418.42	450.33	500.00	544.10	587.88	632.47	877.65	723.45	700.00	816
1042/metaling 2010 2012 2013 2014 2015 2017 2017 2019														
Sampe grade gas 222 all 2472 all 2473 all 2478 all 2477 all 2478 all 2478 all	ecreating theme		2012		2014	2713	2016	2017	2015	7919	2070	747	1721	
Continue (production (p) 0.00 0	Sempler syste of this	2323.00	24/3.31	2020.04	2/102 20	2030.73	3087.85	3437.11	3418.35	3062.63	3740.00		4007.01	420
Construction (processing) ZAUD M ZMP3 M2 ZMP3 M2 <thzmp3 m2<="" th=""> <thzmp3 m2<="" th=""> <thzmp3 <="" m2<="" td=""><td>Service cycle of (b)</td><td>0.00</td><td>0.00</td><td>10.00</td><td>0.00</td><td>0.00</td><td>0.00</td><td>0.00</td><td>0.00</td><td>4610.00</td><td>1980.10</td><td>4100.00</td><td>0.00</td><td></td></thzmp3></thzmp3></thzmp3>	Service cycle of (b)	0.00	0.00	10.00	0.00	0.00	0.00	0.00	0.00	4610.00	1980.10	4100.00	0.00	
Contrants Syche B 0.00 <td>Conversion by the gard</td> <td>2000.00</td> <td>2003.04</td> <td>20/3.02</td> <td>3002.03</td> <td>3234 81</td> <td>3448.38</td> <td>3044 00</td> <td></td> <td>SC. 90.09</td> <td>4233.10</td> <td>4410.31</td> <td>4111.43</td> <td>463</td>	Conversion by the gard	2000.00	2003.04	20/3.02	3002.03	3234 81	3448.38	3044 00		SC. 90.09	4233.10	4410.31	4111.43	463
Andre Preservind 200 00 <	Companied Cycle de	0.00		~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	200	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	~~~~~		0.00	0.00		0.00	V.00	
Barbaria Barbaria J/B J/B J/B J/B <thj b<="" tr=""> Call Chant Calles</thj>	Singer and Street Stree	300.00	300.00	300.00	300.00	300.00	300.00	300.00	300.00	200 00	400.00	300.00	440.00	
Class - retendance over 0.00 0.	Cool - Descripted Ebidiend Red	3/10.34	3/8.34	1/0.34	3/0.34	3/3.30	3/0.30	3/8.34	318.36	3/8.34	401.03	9.09.77	443.73	**
Construction Carbon Carbon <thcarbon< th=""> <thcarbon< th=""> <thcarb< td=""><td></td><td>101.00</td><td>999.90</td><td>10.00</td><td>414 77</td><td>434 77</td><td>434 77</td><td>414 77</td><td>414 12</td><td>414 77</td><td>414 77</td><td>434.32</td><td>445.00</td><td></td></thcarb<></thcarbon<></thcarbon<>		101.00	999.90	10.00	414 77	434 77	434 77	414 77	414 12	414 77	414 77	434.32	445.00	
Call Sugari National Clip J, Jac Call Sugari National Clip J, Jac Clip		411 14	441.00	444 44	487.54	4.74 70	64.1 M	644 76	811 14	429.77	424.17	444.77	443. <u>7</u> 5	10
Lange influence 0.00	anna Marina	414.40			0.00	023.70	0.00	363.76	0.00	0.00	0.00	0.00	0.00	16
State Link Cold Cold <t< td=""><td>Course of the section</td><td>0.00</td><td>0.00</td><td>0.00</td><td>0.00</td><td>0.00</td><td>0.00</td><td>0.00</td><td>0.00</td><td>0.00</td><td>0.00</td><td>0.00</td><td>0.00</td><td></td></t<>	Course of the section	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	
Weste field 0.00 0.00 0.00 11.60 62.56 14.27 197.07 220.86 29.05 310.81 344.72 370.26 Weste field 0.00	Sund Jacks - James ands	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	6.00	0.00	0.00	
Web Communication 0.00 <td>dante funit come</td> <td>0.00</td> <td>0.00</td> <td>0.00</td> <td>11.00</td> <td>a) 44</td> <td>114 77</td> <td>187 07</td> <td>220 80</td> <td>271 03</td> <td>310.01</td> <td>348 72</td> <td>120.36</td> <td></td>	dante funit come	0.00	0.00	0.00	11.00	a) 44	114 77	187 07	220 80	271 03	310.01	348 72	120.36	
Fuel Cell Cogenerator 0.00	Mod Generalors	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	10.00	-
Tidal Power 0.00	Fuel Call Conservator	0.00	0.00	0.00	0.00	0.00	0.00	0.00	000	0.00	0.00	0.00	8.00	
Parabolio-Trough Solar-Power Plant 0.00	Tidal Power	0.00	0.00	0.00	000	0.00	0.00	0.00	0.00	8 00	0.00	0.00	100	
Splar PV 0.00	Parabolio Trough Solar Power Plant	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	
Nukural Gase 877 85 688 07 680 38 881 78 843 28 834 83 824 84 81 22 810 03 801,83 703 81 785 98 Coal 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.	Sulu PU	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	8.00	
Coal 0.00 <th< td=""><td>Nehral Gas</td><td>127 65</td><td>669 07</td><td>660.36</td><td>851 78</td><td>MIN</td><td>640</td><td>878 48</td><td>B10 72</td><td>110 03</td><td>001 90</td><td>793 61</td><td>784.98</td><td>77</td></th<>	Nehral Gas	127 65	669 07	660.36	851 78	MIN	640	878 48	B10 72	110 03	001 90	793 61	784.98	77
C0 00 01 64 01 00 17 05 60 01 30 77 31 73 44 00 77 00 20 02.07 50 02 50	Coel	0.00	0.00	000	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	100	
Hydra 8555 81 8480 33 8385 75 9272 08 8178 37 9087 58 8098 70 8808 73 8817 87 8728 48 8642,18 8555.77 Nachear 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.		98 91	B4 81	80 17	85.66	91.30	77 31	73 44	60 77	M 28	87 87	58 62	54 63	
Nuclear 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.	Hadra	8555 81	8480 35	8385 75	8272 08	8178 37	8087 58	8096 70	8808 73	8017 87	8729 40	8442 18	8545 77	M7
Biornase 452.00 448.45 443.00 430.52 435.13 430.78 429.47 422.20 417.00 413.00 400.00 405.57 Total Supply 17,185.00 17,487.78 17,794.03 18,197.02 18,425.02 18,748.03 10,079.70 10,415.51 19,757.31 20,105.25 20,456.45 20,828.03	Nuclear	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	6.00	
Textal Supply 17,183.00 17,437.75 17,764.03 18,187.02 18,425.02 18,746.03 18,076.76 18,015.31 19,757.21 20,165.25 28,456.26 28,650.63	Diamene	452 44	448 45	41 M	434 52	415 11	430 78	426 47	477 20	417 1	413.00	400 04	405 57	46
and a second and a second and a second	Tetal Supely	17.185.00	17.487.75	17.794.93	18.107.02	18.425.07	18,740.03	10.079.78	16.413.51	11.757.31	21.144.21	N.454.44	71.671.63	21,487
Tetal New Sustainables 852.33 1012.72 1084.17 1118.00 1166.65 1218.36 1273.18 1325.10 1387.14 1437.85 1484.74 1846.86	Tatal New Real devices	862 33	1012 72	1064 17	1118.00	1165.64	1218 30	1271 18	1325.10	1380 14	1417 60	1408.24	1546 84	141

 3825

 4517.36

 0

 5170.48

 0.00

 500.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

 0.00

2000 1003.00 0.00 1050.00 0.00 279.34 0.00

.

						balle-Trough Salar-Power Plant 0.00	Time iou	Call Cogenerator 0.00	Generators 36 27					20 fK Marrie	- Pressuring Pudding Bod 0.00	D D D	Thydra - Optimal Job 00 1	bined sycle II 0.00	thread cycle parts in the second seco				1 New Busishmaken 447.47 4	1 New 11, 11, 11, 11, 11, 11, 11, 11, 11, 11			215.85	0.00	THE GAM 1020 00 11			Call Cogenerator 0.00	Ownerstore 31.65	te fuele cogen (111)			organ retroller 42.20		- Presurind Public Bed 0.00				the system of (D) 000	the syste gas 251.75	Andegy Name 1988		the second second sold it	tal Cumulative Canacity from N
	80				4.81	0.00 0.00							41.83 4.81 M	30.07 30.07	000 000	000 000	20.00	80 80			1142 2142		447.47 447.47		521,41 516,20		204.07 194.02	000	010.48 1000 34			000 000			0.00	0.00	80) 8X					77.55 325.00	80 080	11.015 BI STC	1997 1940			law and Eviation
	8		8	161.71	-	8		8		5 5	5	8	497.24	34.07		8	500	8	1712 CI 7C	200.20	NH.		417.47	K MAN	511 DA	10700,77	Ĩ.	8				8		25.25	0.00	0.00	122	2	Bi			675.00	0.00	71.007	Ī			m Tarkmalaal
1	000	0170.17	8	M3.76		8	33	- 1 8 2		2	8	8	525 79	58	8	08				7,000	A IS		47.47			10472.14	175.45	8		8	10.00	9.8		42.32	8	8	12.2	2	58		0.00	024.01	0.98	1	Ĩ			
	B	17.31	8	01.00	9.91	81	5	32		2.42	8	8	964.93	43. 8	8	8				3087.65	340		47.47		5 <u>9</u>	0500.23		0.8		20	ā 8	8			8	8			38			990.73	8	1008 \$3	ž		al - 300	282
			000	128 41 11	9.61	81	32		11.02	5.42	000	8	50,70	5 8	8	B S				3257.71 34	¥17		47.47 47.47			10440.57 103	199.52	81	4.91	000	10.00	8		12.22	0.90	0.00	2		3		0.00	1155 01 13	8	11171	ž			
		177 N.21	0.00 0.00	122 010.03					11.02	542 53,42	0.00	000 000							0.00	CB 205C 5C 81	2010 2010				0.00	15.M 10252.40				0.00 0.00				12.32 42.32	0.00 0.00	000				00.00 D0.00	0.00	10.54 1442 01						
	11214	12.07	ŝ	81.8	5		18	8.7	211	53.42	8	0.00	514					5176.00	0.00	3748.10	959				0.00	10148.00				0.98	i i		20.19	41 11	8	8	i a			100.00	9	1020.40				ä		
	11	59.62	8			22	19	442	22,11	13.43	0.00	0.00			38		9.9	SADA S	000	3013.00	2021				0.90	10048.30	120.12		4.01	0.00		35	20.10		8	8				100.00	0.00		81					
8	10000.77	54,65	8				8	4.3	11.02	53,42	.0 0	8			38		0.00	B704.30	0.00	4080.27	N			171.94	0.90	10047.00				0.00		12.DC	20.10	4.5	8				8	100.00	8				ļ			
000	1900 IS 07M	53,00 51.1	000 00				0.00	420 41	20.11 20.1	51.42 51.42	0.00						0.00	1073.01 0347.	0.00	1247.00 4414.	1000	40 H	THEN BUTT	411 H 412	0.00	1040.42 9740.1		804.72 MM	401 4	0.00			20.10	40.55				0.00	080	00K 00K	000							
00	10.1000 IS	19 44.72					00	X 14 X			8 00					0000	00	77 0055.21	8	4542.0		100		17 457.5	8					8		27 34.2						800	8 00	0.000								

Table B.3 - Total generation capacity (MW) by technology type - SBC

Total New Sund	Teta Survy	Bionates	No.	ł	2	Number Case	ALC: N	Parabolo-Troug	Tidal Power	Fuel Call Cogen	Wind Generator	With Line of	Smill Hydro - Is	Coal games in	Lange Hydro	Get togen retro	Nones pulle	Coal - Pressure	Manana staan	Small hydro - Ot	Combined cycle	Combined syste	Simple cycle of (Simple syste per	Technology Na	Total New Dust	Tela Supply	Blomer	Ĭ	Hay			Purpose Troug	Tidal Power	Fuel Cell Cogenu	Wind Generatory	Waste Nets copi		Coal an eller turb			Cod - meaning	Blome ut steem	do - ciphy gang	Cambined cycle (Combined syster	Simple syste of (Single cycle gas	Techniky of New	Total Cun
								h Solar-Pourt Plant		analor	-	3		Ŧ		2		a Tubed Ind		Ī	2	1	3										Solly-Power Plant		rator	_	3					d runne wa		I		1	e			iulative Cap
785.85	17,155,00	452.00	8	844		17.00		200	100	0.00	000	0.00	000	000	8	1000 00	M7.67	8	347.07	200 00	8	N 962	0.00	1940.05	3	 360.64	13,506,77	53N.00	8	11110.77	215 85		38		0.00	80	8	B	B	81	50		20.22	20.00	000	H.10			1990	acity from
OX KD	97,447,75	441 45	8	1000			999	28	8	8	8	8	8	8	8	1000.00	144.BD	8	244.85	100.00	ŝ	2743.75	8	1942.02	1412	370.00	12/12/07	621.41	0.00					8	0.00	0.00	8	8	83				N N	200.00	0.00	48,14	0.90	200.01	1947	n New and
\$12.77	17,714,03		0.00		5			88	8		8	000	0.00	88	ŝ	1000.00	201.34	8	111 JA	200.00	8	3054 QH	8	21117.20	14C	114.31	14,021,74	518.20	8					8	0.00	80	8	B	B	3		8	57.15	88	0.8	70.00	0.00	417.77	Ĩ	i Existing
100	11,107,12	439.52	8	9272.00				28	000	8	8	8	8	000	8	1000.00	у 82	.0.8	X BC	8 8	0.00	1010.17	0.00	Z243.71	344	450.29	14,345,70	511,04	0.00	10700.77			88	8	8	800	8	8	8	8		198	75.15	00.000	0.00		80	674.00	Ī	Technolo
M 003	11,425,12	\$.	8	1170.37			199	18	8	8	080	8	8	8	0.00	1000 00	315.47	8	315 47	88	000	Mer K	000	2071.50	3.EC	400 54	14,000,54	50,502	8	10072.M	i.		88	0.90	8	0 00	0.00		81				10.02	300.00	0.00	130.07	000	708.62	246	gies - BC N Inductri: 20
86 5 54	10,740,03	430.78	8	9097.54	12		W V	88	8	000	8	8	0.90	80	8	1000.00	332.78	8	332 78	88	0.00	3053.24	0.08	2500 64	2910	508.72	14,020,12	500.07	0.00	10544.23		N.V.	8	000	8	0.00	8	B	88			0.00	2	100.00	0.08	314.00	0.00	797.08	241	Aodel - EA
1009 29	11,071,70	429.47	8	10 mm				8	8	8	8	14.57	8	g	8	1000.00	347.35		347.35	1000	0.00	4110.14	0.00	20 00 25	3417	500.22	14,001,10	45.1	8	10440 57				0.00	08	80	8	B	Ba			0.00	116.01	200.00	8	494 83	0.00	NN 00	W2	. Stranbald over land
1053 47	11,011,51	422.20	8				1.10	88	8	8	8	20.01	8	8	8	1000.00	¥2.9	8	302.08	80.00	0.00	4307.07	0.08	1757.25	ž	554.55	16,186,67	490,80	8	10055 M			8	0.92	0.00	ĝ	8	B	88	308	199.1		12 121	200.00	0.00	470.74	0.00	973.57	2003	soul for breaking
1004.15	10,757.31	417.00	8				999		00	8	8	4.70	00	8	8	1000.00	376 00	0.00	170.00	100 QQ	90	4454.51	0.00	2007.00)HK	579,71	16,342,62	445.14	00	10252.40			0.00	0.8	0.00	0.00	8	B	Bi				X	300.00	000	X 928	0.00	1040,42	Ĩ	2
1134.07	201.106.23		0.00	8729.40			000	8	8	8	8	57,10	90	00	12.81	1009.00	ы.н	8		100.00	9.00		ŝ	3017.44	NN	102,70	11,100/11	4(1).(1)	00			19.9		8	0.00	8	8	ŝ	ŝ			9.9	151.35	100.00	0.00	1034.54	8	1744.54	Ĩ	*
1170.02	21.451.45		8	MA2.10	5.2	10.51	0.90		0.00	8	0.00	70.11	8	0.00		1000.00	492,82	0.00	402.02	100.00	0.00	5203.00	0.00	3140.19	NH	835.14	E THU'LL	479.32	200					2.00	0.00	99	8	8	88			8	107.54	300.00	0.00	1200.30	0.00	1200.02	Ĩ	
1215.62	N LAN IN	60 57	8	1005.77	58		0.00	18	8	8	0.80	8.W	08	8	14.18	1000.00	410.13		410.13	X00.00	0.00	5481.14	0.98	3270.00	29622	007.41	10,000,00	471.58	0.98			N.N.		8	0.00	8	8	8	81			0.90	100,71	00.00	0.00	1530.64	0.90	1100	346)	
1255.00	27.187.10	401.51		M70.21		770.12	00.0	8	0.00	0.00	ŝ	B4.71	0.00	0.08	52.51	1000.00	429.49	0.00	429 40	300 00	0.00	5792.20	0.08	3412.00	(2M	000.47	10,329,00	18.1	0.00					0.00	0.00	0.00	8	B	B			8	100.74	300.00	0.00	1785.54	9	1500.32	ĩ	
1299.22	21,500,73	307.50	8	1065 51		10.33	0.90	8	8	0.00	80	110.22	8	08		1000.00	443,88	000	413.00	200.00	0.00	0047.25	0.8	3540.07	XX	M 7164	10,010,10	442.17	0.8	9748.03			88	0.8	0.00	0.00	8	8	Bi	0.00	215.07	0.90	215.07	300.00	0.00	2033.15	0.00	1027.21	Ĩ	
1337.23	21.MH.12	340.52	0.00			702.03	0.90		0.00	0.00	0.00	121.84	0.00	0.00	78.70	1000.00	454.07	0,00	454.47	¥0.00	0.00	B336.14	0.9	10.1000	ASK	700.01	10,000,52	457.55	0.00					0.00	0.00	0.00	0.00		83			0.00	221.51	300.00	0.00	10,102	0.00	1745.27	2 2 2	

Table B.4 - Total generation capacity (MW) by technology type - ADD

																	2
Generating St	atistics -	8.C. M	ođel - B	AU		Innual GWI	production a	d Revenue I	Received for I	Electricity (m	illilion \$)						
	1995 CM/b	Million B	1996	SAUlon C	1997 CM/h	Million B	1996	Adding the	1999	Million R	2000	Million C	2001		2002	Adultan P	2003
Simple cycle ops	00	0.0	A23 8	18.6	D44 1	20.2	1355.8	43.6	1743.8	57 R	2118.0	72.8	2974.0	R3 A	2630.0		2000
Simple cycle oil (b)	0.0	0.0	0.0	0.0	00	0.0	0.0	0.0	0.0	00	00	00	00	0.0	0.0	0.0	2000.0
Combined cycle gas	0.0	0.0	109.0	3.2	683 7	20.2	1632.2	48.2	2729.5	81.9	3943.1	110.0	4825.6	147.4	5759 1	178.6	8726 5
Combined cycle oil	0.0	0.0	00	0.0	00	0.0	0.0	0.0	00	0.0	0.0	0.0	0.0	0.0	00	0.0	00
Small hydro - Optimal	0.0	0.0	1708.2	30 t	1708 2	32.3	1708.2	34.6	1708.2	36.4	1708.2	39.0	1708.2	40.3	1708.2	40.8	1708.3
Biomass steam	00	D.0	0.0	00	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
Coal - Pressurized Flui	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
Biomass gasifier turbin	0.0	0.0	0.0	00	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
Gas copen retrofits	0.0	0.0	262.8	5.0	436.1	8.2	638.4	12.9	839 3	17.9	1039.0	23.7	1171.2	27.7	1302.4	31.1	1432.7
Large Hydro	0.0	0.0	0.0	00)	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
Coel gasifier turbine	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0[0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
Small Hydro - lower gra	0.0	0.0	0.0	00	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
Waste fuels cogen	0.0	0.0	0.0	00	0.0	0.0	0.0	0.0	00	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
Wind Generators	0.0	0.0	0.0	0.0	00	0.0	0.0	0.0	0.0	00	0.0	0.0	0.0	0.0	0.0	0.0	0.0
Fuel Cell Cogenerator	0.0	0.0	00	0,0	00	0.0	0.0	00	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
Tidal Power	00	0.0]	0.0	0.0	0.0	0.0	0.0	00	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
Parabolic-Trough Solar	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
Solar PV	0.0	0.0	0.0	0.0	0.0	0.0	0.0	00	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
NA	0.0	0.0Į	00	0.0	0.0	0.0	0.0	0.0	0.0	اه ه	0.0	0.0	0.0	0.0	0.0	0.0	0.0
NVA	0.0	0.0	0.0	00	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	Q.Q	O .(
Natural Gas	1884.5	57.9	1768.1	55.9	1684.2	55.0	1574.3	55.2	1489.3	53.5	1373.7	53.1	1312.P	52.8	1253.5	52.5	1203.7
Coel	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	00	0.0	0.0	0.0	0.0	0.0	0.0
O#	148.3	5.1	133.0	5.0	127.1	5.2	120.6	5.3	114.7	5.5	109.0	5.5	103.5	5.5	98.4	5.5	\$3 .4
Hydro	62120.4	1140.0	02054.5	1138.2	62284.6	1219.4	62587.6	1307.3	82741.4	1373.2	02777.1	1485.9	02556.3	1500.6	62264.9	1513.0	61989.1
Nuclear	0.0	սո	ŰŲ	0.0	00	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
	FCBE	40.01	894.0	اه هد	400.4	40.0	440.0	44.4	440.0	ام مه		ام هم		40.7		40.0	
Total (GW/h Milliona)	558.5	19.0	534.9	15,5	499,1	18.6	449.8	18.1	419.9	18.2	398.3	16.4	382.1	18.7	368.2	19.0	354.7
Total (GWh, Millions)	558.5 \$4,712 \$	<u>19.0</u> <u>1,222</u>	534,9 67,218	16,8 1,273	499.1 68,367	18.6 8 1,389	449.8 70,067	<u>18.1</u> 8 1,525	419.9 71,766	18.2 \$ 1,644	398.3 73,464	18.4 8 1,799	<u>382.1</u> 74,435	18.7 1,982	<u>368.2</u> _75,405	19.0 8 1,833	<u>354.1</u> 76,374
Total (GWh, Millions)	558.5 64,712 8 2011	19.0 1,222	534.9 67,215 2012	18,8 1,273	499.1 68,367 2013	18.6 \$ 1,389	449.8 70,067 2014	18.1 \$ 1,525	419.9 71,766 2015	18.2 5 1,644	396.3 73,464 2016	16.4 8 1,799	382.1 74,435 2017	18.7 3 1,982	368.2 75,405 2018	19.0 8 1,933	354.1 76,374 2011
Total (GWh, Millions)	558.5 64,712 8 2011 GWh 5336 7	19.0 1,222 Million \$ 186.3	534.9 67,218 2012 <u>GWh</u> 5661.1	18,8 1,273 Million S 100 2	499.1 68,367 2013 GWh 6028.8	18.6 \$ 1,389 <u>Million \$</u> 212.4	449.8 70,067 2014 GWh 6362.1	18.1 \$ 1,525 Million \$ 225 9	419.9 71,766 2018 GWh 6725.3	18.2 § 1,644 Million \$ 239 3	398.3 73,464 2016 GWh 7070.0	16.4 8 1,799 Million \$ 252.9	382.1 74,435 2017 GWh 7430.3	18.7 8 1,862 Million \$ 267 1	368.2 75,405 2018 GWh 7799.3	19.0 \$ 1,833 <u>Million \$</u> 281 7	354.3 76,374 2011 <u>GWI</u>
Total (GWh, Millions) Simple cycle gas Simple cycle gas	558.5 64,712 8 2011 <u>GWh</u> 5338.7 0.0	<u>19.0</u> <u>1,222</u> <u>Million \$</u> 186.3 0.0	534.9 67,215 2012 <u>GWh</u> 5661.1	16.6 8 1,273 Million S 199.2 0.0	499.1 68,367 2013 GWh 6028.8 0.0	18.6 1,389 Million S 212.4 0.0	449.8 70,067 2014 GWh 6382.1 0.0	18.1 \$ 1,525 Million \$ 225.9 0.0	419.9 71,766 2015 GWh 6725.3 0.0	18.2 § 1,644 Million \$ 239.3 0.0	396.3 73,464 2016 GWh 7070.0 0.0	18.4 8 1,799 <u>Million \$</u> 252.9 0.0	382.1 74,435 2017 	18.7 1,062 Million \$ 267.1 0.0	368.2 75,405 2018 <u>GWh</u> 7799.3 0.0	19.0 \$ 1,933 <u>Million \$</u> 281.7 0.0	354.3 76,374 2011 GWI 8174.3
Simple cycle gas Simple cycle oli (b) Combined cycle gas	558.5 64,712 8 2011 <u>GWh</u> 5338.7 0.0 18216.3	19.0 1,222 Million \$ 186.3 0.0 544.0	534.9 67,215 2012 <u>GWh</u> 5681.1 0.0 19990 5	18.8 1,273 Million S 199.2 0.0 595.4	499.1 68,367 2013 GWh 6028.8 0.0 21798.0	18.6 \$ 1,389 <u>Million \$</u> 212.4 0.0 647.9	449.8 70,067 2014 GWh 6382.1 0.0 23623.6	18.1 \$ 1,525 <u>Million \$</u> 225.9 0.0 701.0	419.9 71,766 2015 GWh 6725.3 0.0 25468.2	18.2 \$ 1,644 Million \$ 239.3 0.0 754.9	396.3 73,464 2016 GWh 7070.0 0.0 27332.3	18.4 8 1,799 Million \$ 252.9 0.0 609.5	382.1 74,435 2017 GWh 7430.3 0.0 29208.6	18.7 1,002 Million \$ 267.1 0.0 873.1	368.2 75,405 2018 GWh 7799.3 0.0 31099.1	19.0 \$ 1,933 <u>Million \$</u> 281.7 0.0 929 4	354.7 76,374 2011 <u>GWI</u> 8174.7 0.0 33012.3
Simple cycle gas Simple cycle gas Simple cycle oli (b) Combined cycle oli	558.5 64,712 8 2011 GWh 5338.7 0.0 18216.3 0.0	19.0 1,222 Million \$ 186.3 0.0 544.0 0.0	534.9 67,218 2012 <u>GWh</u> 5681.1 0.0 19990 5 0.0	16.8 1,273 Million S 199.2 0.0 595.4 0.0	499.1 68,367 2013 GWh 6028.8 0.0 21798.0 0.0	18.6 1,389 Million S 212.4 0.0 647.9 0.0	449.8 70,067 2014 GWh 6382.1 0.0 23623.6 0.0	18.1 \$ 1,525 Million \$ 225.9 0.0 701.0 0.0	419.9 71,766 2015 GWh 6725.3 0.0 25468.2 0.0	18.2 5 1,644 Million 5 239.3 0.0 754.9 0.0	396.3 73,484 2016 GWh 7070.0 0.0 27332.3 0.0	16.4 8 1,799 <u>Million \$</u> 252.9 0.0 609.5 0.0	382.1 74,435 2017 <u>GWh</u> 7430.3 0.0 29206.8 0.0	18.7 1,002 Million \$ 267.1 0.0 873.1 0.0	368.2 75,405 2018 GWh 7799.3 0.0 31099.1 0.0	19.0 \$ 1,933 Million \$ 281.7 0.0 \$29.4 0.0	354.1 76,374 2011 6W1 8174.1 01 33012.3 0.0
Simple cycle gas Simple cycle gas Simple cycle oli (b) Combined cycle oli Combined cycle oli Small hydro - Optimal	558.5 64,712 8 2011 GWh 5338.7 00 18216.3 00 1708.2	19.0 1,222 Million \$ 186.3 0.0 544.0 0 0 47.7	534.9 67,218 2012 GWh 5681.1 0.0 19990.5 0.0 1708.2	16,6 8 1,273 Million 8 199,2 0,0 595,4 0,0 48,6	499.1 68,367 2013 GWh 6028.8 0.0 21798.0 0.0 1708.2	18.6 8 1,389 <u>Million \$</u> 212.4 0.0 647.9 0.0 48.7	449.8 70,067 2014 GWh 6382.1 0.0 23623.6 0.0 1706.2	18.1 8 1,525 Million \$ 225.9 0.0 701.0 0.0 48.8	419.9 71,766 2015 GWh 6725.3 0.0 25468.2 0.0 1708.2	18.2 1,644 Million \$ 239.3 00 754.9 00 48.9	396.3 73,464 2016 GWh 7070.0 0.0 27332.3 0.0 1706.2	18.4 8 1,799 Million \$ 252.9 0.0 609.5 0.0 48.9	382.1 74,435 2017 GWh 7430.3 0.0 29208.6 0.0 1706.2	18.7 1,002 Million \$ 267.1 0.0 873.1 0.0 49.4	368.2 75,405 2018 GWh 7799.3 0.0 31099.1 0.0 1708.2	19.0 \$ 1,933 <u>Million \$</u> 281.7 0.0 929.4 0.0 49.5	354.7 76,374 2011 6W/ 8174.7 0.0 33012.3 0.0 1708.3
Simple cycle gas Simple cycle gas Simple cycle oli (b) Combined cycle oli Small hydro - Optimal Biomass steam	558.5 64,712 8 2011 <u>GWh</u> 5338.7 00 18216.3 00 1706.2 0.0	19.0 1,222 Million \$ 186.3 0.0 544.0 0.0 47.7 0.0	534.9 67,218 2012 GWh 5681.1 0.0 19900 8 0.0 1708.2 0.0	16,6 9 1,273 Million S 1990 2 0.0 595.4 0.0 595.4 0.0 48.6 0.0	499.1 68,367 2013 GWh 6028.8 0.0 21798.0 0.0 1708.2 0.0	18.6 8 1,389 Million 5 212.4 0.0 647.9 0.0 48.7 0.0 0.0	449.8 70,067 2014 GWh 6382.1 0.0 23623.6 0.0 1706.2 0.0	18.1 8 1,525 Million \$ 225.9 00 701.0 00 48.8 00	419.9 71.766 2018 GWh 6725.3 0.0 25468.2 0.0 1708.2 0.0	18.2 § 1,644 Million \$ 239.3 0.0 754.9 0.0 48.9 0.0	396.3 73,484 2018 GWh 7070.0 0.0 27332.3 0.0 1708.2 0.0	18.4 8 1,799 Million 8 252.9 0.0 609.5 0.0 48.9 0.0	382.1 74,435 2017 GWh 7430.3 0.0 29208.6 0.0 1706.2 0.0	18.7 9 1,902 Million \$ 267.1 0.0 873.1 0.0 49.4 0.0	368.2 75,405 2018 GWh 7799.3 0.0 31099.1 0.0 1708.2 0.0	19.0 § 1,833 <u>Million \$</u> 281.7 0.0 929.4 0.0 49.5 0.0	354. 76,374 2011 6W/ 8174. 0.0 33012. 0.1 1708. 0.0
Simple cycle gas Simple cycle gas Simple cycle oil (b) Combined cycle gas Combined cycle oil Small hydro - Optimal Biomass steam Coel - Pressurized Flui	558.5 64,712 8 2011 GWh 5338.7 0.0 18216.3 0.0 1708.2 0.0 0.0	19.0 1,222 Million \$ 186.3 0.0 544.0 0.0 47.7 0.0 0.0	534,9 67,218 2012 <u>GWh</u> 5681,1 0.0 19990.6 0.0 1708.2 0.0 0.0	16,6 9 1,273 Million S 1990 2 0.0 595.4 0.0 48.6 0.0 0.0	499.1 68.367 2013 GWh 6028.8 0.0 21798.0 0.0 1708.2 0.0 0.0 0.0	18.6 8 1,389 <u>Million \$</u> 212.4 0.0 647.9 0.0 48.7 0.0 0.0 0.0 0.0 0.0	449.8 70,067 2014 GWh 6382.1 0.0 23623.6 0.0 1706.2 0.0 0.0 0.0	18.1 8 1,525 Million \$ 225.9 00 701.0 00 48.8 00 00 00	419.9 71.766 2015 GWh 6725.3 0.0 25468.2 0.0 1708.2 0.0 0 00	18.2 § 1,644 Million \$ 239.3 0.0 754.9 0.0 48.9 0.0 0.0 0.0	396.3 73,464 2016 GWh 7070.0 0.0 27332.3 0.0 1708.2 0.0 0 0 0 0	18.4 8 1,799 Million \$ 252.9 0.0 609.5 0.0 48.9 0.0 0.0 0.0	382.1 74,435 2017 GWh 7430.3 0.0 29206.6 0.0 1706.2 0.0 0.0	18.7 3 1,992 Million \$ 267.1 0.0 873.1 0.0 49.4 0.0 0.0 0.0	368.2 75,405 2018 GWh 7799.3 0.0 31099.1 0.0 1708.2 0.0 0.0	19.0 § 1,833 <u>Million \$</u> 281.7 0.0 929.4 0.0 49.5 0.0 0.0	354.1 76,374 2011 6W/ 8174.1 0.0 33012.3 0.0 1708.2 0.0 0.0
Simple cycle gas Simple cycle gas Simple cycle oil (b) Combined cycle gas Combined cycle oil Small hydro - Optimal Biomass steam Coel - Pressurized Flui Biomass gasifier turbin	558.5 44,712 8 2011 GWh 5338.7 0.0 18216.3 0.0 1708.2 0.0 0.0 0.0 0.0	19.0 1,222 Million \$ 186.3 0.0 544.0 0.0 47.7 0.0 0.0 0.0 0.0	534,9 67,218 2012 GWh 5681,1 0.0 19990.8 0.0 1708.2 0.0 0.0 0.0 0.0	16,6 8 1,273 Million 5 199.2 0.0 595.4 0.0 48.6 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	499.1 68.367 2013 GWh 6028.8 0.0 21798.0 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0	18.6 \$ 1,389 Million \$ 212.4 0.0 647.9 0.0 48.7 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	449.8 70,067 2014 GWh 6362.1 0.0 23623.6 0.0 1706.2 0.0 0.0 0.0 0.0	18.1 \$ 1,525 Million \$ 225.9 00 701.0 00 48.8 00 00 00 00	419.9 71.766 2015 GWh 6725.3 0.0 25468.2 0.0 1708.2 0.0 1708.2 0.0 0.0	18.2 § 1,644 Milition \$ 239.3 00 754.9 00 48.9 00 00 00 00	306.3 73,484 2018 GWh 7070.0 0.0 27332.3 0.0 1708.2 0.0 1708.2 0.0 0.0	18.4 8 1,799 Million \$ 252.9 0.0 609.5 0.0 48.9 0.0 0.0 0.0 0.0	382.1 74,435 2017 GWh 7430.3 0.0 29206.6 0.0 1706.2 0.0 0.0 0.0	18.7 3 1,992 Million \$ 267.1 0.0 873.1 0.0 49.4 0.0 0.0 0.0 0.0	368.2 75,405 2018 GWh 7799.3 0.0 31099.1 0.0 1708.2 0.0 0.0 0.0	19.0 8 1,833 <u>Million \$</u> 281.7 0.0 929.4 0.0 49.5 0.0 0.0 0.0	354.1 76,374 2011 8174.1 0.0 33012.3 0.0 1708.2 0.0 0.0 0.0 0.0
Simple cycle gas Simple cycle gas Simple cycle oil (b) Combined cycle oil Small hydro - Optimal Biomass steam Coel - Pressurized Flui Biomass gasifier turbin Gas cogen retrofits	558.5 44,712 8 2011 GWh 5338.7 0.0 18216.3 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	19.0 1,222 Million \$ 186.3 0.0 544.0 0.0 47.7 0.0 0.0 0.0 0.0 77.3	534.9 67,218 2012 GWh 5681.1 0.0 19900 8 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	16,6 1,273 Million \$ 190.2 0.0 595.4 0.0 595.4 0.0 48.6 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	499.1 88,367 2013 <u>GWh</u> 6028.8 00 21798.0 00 1708.2 00 00 3142.8	18.6 8 1,389 Million S 212.4 0.0 647.9 0.0 48.7 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	449.8 70,067 2014 GWh 6382.1 0.0 23623.6 0.0 1706.2 0.0 0.0 0.0 0.0 33332.2	18.1 8 1,525 Million \$ 225.9 00 701.0 00 48.8 00 00 00 00 95 2	419.9 71.766 2018 GWh 6725.3 00 25468.2 00 1708.2 00 1708.2 00 00 00 3523.5	18.2 8 1,844 Million \$ 239.3 00 754.9 00 48.9 00 00 00 00 00 100.8	3963 73,484 2018 GWh 7070.0 00 27332.3 0.0 1708.2 00 1708.2 00 00 00 00 00 00	18.4 8 1,799 Million \$ 252.9 0.0 609.5 0.0 48.9 0.0 0.0 0.0 0.0 106.5	382.1 74,435 2017 GWh 7430.3 00 29206.6 0.0 1706.2 0.0 0.0 0.0 0.0 0.0 0.0	18.7 3 1,992 Million \$ 267.1 0.0 873.1 0.0 49.4 0.0 0.0 0.0 0.0 113.2	368.2 75,405 2018 GWh 7799.3 0.0 31099.1 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	19.0 8 1,833 <u>Million \$</u> 281.7 0.0 929.4 0.0 49.5 0.0 0.0 0.0 0.0 119.1	354.3 76,374 2011 6174.3 0.0 33012.3 0.0 1708.3 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0
Simple cycle gas Simple cycle di (b) Combined cycle di (b) Combined cycle di Small hydro - Optimal Biomass steam Coel - Pressurized Flui Biomass gasifier turbin Gas cogen retrofits Large Hydro	558.5 44,712 8 2011 GWh 5336.7 0 0 18216.3 0 0 1706.2 0 0 0 0 0 0 0 0 2769 6 0 0	19.0 1,222 186.3 0.0 544.0 00 47.7 0.0 0.0 0.0 77.3 0.0	534.9 67,218 2012 GWh 5681.1 0.0 19990 8 0.0 19990 8 0.0 19990 8 0.0 19990 8 0.0 0 1708.2 0.0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	16,6 8 1,273 Million 5 100.2 0.0 505.4 0.0 48.6 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	499.1 88.367 2013 <u>GWh</u> 6028.8 0.0 21798.0 0.0 1708.2 0.0 0.0 0.0 3142.8 0.0	18.6 8 1,389 Million \$ 212.4 0.0 647.9 0.0 48.7 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	449.8 70,067 2014 GWh 6382.1 00 23823.6 00 1708.2 00 00 00 00 33322 00	18.1 8 1,525 Million \$ 225.9 00 701.0 00 48.8 00 00 00 00 00 00 00 00 00 0	419.9 71,766 2018 GWh 6725.3 00 25468.2 00 1708.2 00 1708.2 00 00 00 00 00 00	18.2 8 1,844 Million \$ 239.3 00 754.9 00 48.9 00 00 00 00 00 00 00 00 00 0	3963 73,484 2018 GWh 7070.0 00 27332.3 0.0 1708.2 00 00 00 00 0.0 00 0.0 0.0 0.0 0.0 0.	18.4 8 1,799 252.9 0.0 609.5 0.0 48.9 0.0 0.0 0.0 0.0 106.5 0.0	382.1 74,435 2017 GWh 7430.3 00 29206.6 0.0 1706.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.7 3 1,992 Million \$ 267.1 0.0 873.1 0.0 49.4 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	368.2 75,405 2018 GWh 7799.3 0.0 31099.1 0.0 1708.2 0.0 0.0 0.0 4109.8 0.0	19.0 8 1,833 <u>Million S</u> 281.7 0.0 929.4 0.0 49.5 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	354.3 76,374 2011 3012.3 0.0 1708.3 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0
Simple cycle gas Simple cycle gas Simple cycle oli (b) Combined cycle oli Small hydro - Optimal Biomass steam Coel - Pressurized Flui Biomass gasifier turbin Gas cogen retrofits Large Hydro Coel gasifier turbine	558.5 44,712 8 2011 <u>GWh</u> 5338.7 0.0 18216.3 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	19.0 1,222 Million \$ 186.3 0.0 544.0 0.0 47.7 0.0 0.0 0.0 77.3 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	534.9 67,218 2012 GWh 0.0 19990.8 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	16,8 8 1,273 Million S 19922 0.0 595.4 0.0 595.4 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	499.1 88,367 2013 <i>GWh</i> 6028.8 0.0 21798.0 0.0 1708.2 1708.0 0.0 0.0 0.0 0.0 3142.8 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.6 \$ 1,389 Million \$ 212.4 0.0 647.9 0.0 48.7 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	449.8 70,067 2014 GWh 6382.1 0.0 23623.6 0.0 1708.2 0.0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	18.1 1,525 Million \$ 225.9 00 701.0 0.0 48.8 0.0 0.0 0.0 0.0 0.0 0.0 0.0	419.9 71,765 2015 GWh 6725.3 0.0 25468.2 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 3523.5 0.0 0.0 0.0 0.0	18.2 8 1,644 Million 5 239.3 0.0 754.9 0.0 48.9 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	3963 73,484 2016 GWh 7070.0 0.0 27332.3 0.0 27332.3 0.0 27332.3 0.0 27332.3 0.0 0.0 0.0 3716.8 0.0 0.0	16.4 8 1,799 252.9 0.0 609.5 0.0 48.9 0.0 0.0 106.5 0.0 106.5 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	382.1 74,435 2017 GWh 7430.3 0.0 29208.6 0.0 29208.6 0.0 0.0 0.0 0.0 0.0 0.0 3912.1 0.0 0.0	18.7 3 1,992 Million \$ 267.1 0.0 873.1 0.0 49.4 0.0 0.0 0.0 0.0 113.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	368.2 78,405 2018 GWh 7799.3 0.0 31099.1 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 4109.8 0.0 0.0	19.0 8 1,833 Million S 281.7 0.0 929.4 0.0 49.5 0.0 0.0 0.0 0.0 119.1 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	354.3 76,374 2011 6W1 8174.3 0.0 33012.3 0.0 1708.3 0.0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
Simple cycle gas Simple cycle gas Simple cycle oli (b) Combined cycle gas Combined cycle oli Small hydro - Optimal Biomass steam Coal - Pressurized Flui Biomass gasifier turbin Gas cogen retrofits Large Hydro Coal gasifier turbine Small Hydro - lower gra	558.5 44,712 8 2011 GWh 5338.7 0.0 18216.3 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	19.0 1,222 Million \$ 186.3 0.0 544.0 0.0 544.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	534.9 67,218 2012 GWh 5681.1 0.0 19990.8 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	16.8 1,273 Million S 199.2 00 595.4 00 595.4 00 00 00 00 00 00 00 00 00 0	499.1 88,367 2013 <u>GWh</u> 6028.8 0.0 21798.0 0.0 1708.2 0.0 0.0 0.0 3142.8 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.6 8 1,369 Million \$ 212.4 0.0 647.9 0.0 647.9 0.0 647.9 0.0 647.9 0.0 647.9 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	449.8 70,067 2014 GWh 6382.1 0.0 23623.6 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.1 9 1,525 Million \$ 225.9 0.0 701.0 0.0 48.8 0.0 0.0 0.0 0.0 0.0 0.0 0.0	419.9 71,766 2015 GW/h 6725.3 0.0 25468.2 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.2 8 1,644 Million \$ 239.3 0.00 754.9 0.0 48.9 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	3963 73,484 2018 GWh 70700 0 0 0 273323 0 0 1708.2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	16.4 8 1,799 252.9 0.0 609.5 0.0 48.9 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	382.1 74,435 2017 GWh 7430.3 0.0 29208.6 0.0 1708.2 0.0 0.0 0.0 0.0 3012.1 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	18.7 3 1,992 3 1,992 3 1,992 3 0,0 3 7.1 0 0 8 73.1 0 0 8 73.1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	368.2 78,405 2018 GWh 7799.3 0.0 31099.1 0.0 1708.2 0.0 0.0 0.0 4109.8 0.0 0.0 0.0 0.0	19.0 8 1,833 Million S 281.7 0.0 929.4 0.0 49.5 0.0 0.0 0.0 119.1 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	354.3 76,374 2011 GWI 8174.1 0.0 33012.3 0.0 1706.3 1706.3 0.0 0.0 4309.3 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0
Simple cycle gas Simple cycle gas Simple cycle oil (b) Combined cycle gas Combined cycle gas Combined cycle oil Small hydro - Optimal Biomass staam Coel - Pressurized Flui Biomass gasifier turbin Gas cogen retrofits Large Hydro Coel gasifier turbine Small Hydro - Iower gra Waste fuels cogen	558.5 44,712 8 2011 GWh 5336.7 0.0 18216.3 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	19.0 1,222 Million \$ 186.3 0.0 544.0 0.0 544.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	534.9 67,218 2012 GWh 5681.1 00 19990.8 00 1708.2 00 00 00 00 00 00 00 00 00 00 00 00 00	16,8 1,273 Million S 100,2 0,0 0,0 0,0 0,0 0,0 0,0 0,0	499.1 88,367 2013 <u>GWh</u> 6028.8 00 21798.0 00 1708.2 00 00 00 00 3142.8 00 00 00 00 00 00 00 00 00 0	18.6 8 1,389 Million \$ 212.4 0.0 647.9 0.0 647.9 0.0 647.9 0.0 647.9 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	449.8 70,067 2014 GWh 6382.1 0.0 23623.6 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.1 9 1,525 Million \$ 225.9 0 00 701.0 00 48.8 00 00 00 95.2 00 00 00 00 00 00 00 00 00 0	419.9 71,766 2018 GWh 6725.3 0.0 25468.2 0.0 1708.2 0.0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	18.2 8 1,644 Million \$ 239.3 000 754.9 00 48.9 00 48.9 00 100.4 00 00 00 00 00 00 00 00 00 0	306.3 73,484 2018 GWh 7070.0 0.0 27332.3 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	16.4 8 1,799 0.0 609.5 0.0 48.9 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	382.1 74,435 2017 GWh 7430.3 0.0 29206.8 0.0 1706.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.7 3 1,992 3 1,992 3 1,992 3 267.1 0 00 873.1 0 0 873.1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	368.2 78,405 2018 GWh 7799.3 0.0 31099.1 0.0 1708.2 0.0 0.0 0.0 4109.6 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	19.0 8 1,833 <u>Million S</u> 281.7 0.0 929.4 0.0 0.0 0.0 0.0 119.1 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	354.3 76,374 2011 GWI 8174. 0.0 33012.2 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0
Simple cycle gas Simple cycle gas Simple cycle oil (b) Combined cycle oil Small hydro - Optimal Biomass steam Coel - Pressurized Flui Biomass gasifier turbin Gas cogen retrofits Large Hydro Coal gasifier turbine Small Hydro - lower gra Waste fuels cogen Wind Generators	558.5 44,712 0 2011 GWh 5338.7 0.0 18216.3 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	19.0 1,222 Million \$ 186.3 0.0 544.0 0.0 47.7 0.0 0.0 0.0 0.0 0.0 0.0 0.0	534.9 67,218 2012 GWh 5681.1 0.0 19900 8 00 1708.2 00 00 00 00 00 00 00 00 00 00 00 00	16.6 1,273 Million S 199.2 0.0 555 4, 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.	499.1 88,367 2013 <u>GWh</u> 6028.8 00 21798.0 00 1708.2 00 00 3142.8 00 00 3142.8 00 00 00 00 00 00 00 00 00 0	18.6 8 1,389 Million S 212.4 0.0 647.9 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	449.8 70,067 2014 GWh 6382.1 00 23623.6 0.0 1708.2 00 00 00 00 00 00 00 00 00 00 00 00	18.1 9 1,525 Million \$ 225.9 000 701.0 00 701.0 00 48.8 00 00 00 00 00 00 00 00 00 0	419.9 71.766 2018 GWh 6725.3 00 25468.2 00 1708.2 00 00 00 00 00 00 00 00 00 00 00 00	18.2 8 1,844 Million \$ 239.3 00 754.9 00 48.9 00 00 00 00 00 00 00 00 00 0	306.3 73,484 2018 GWh 7070.0 00 27332.3 00 1708.2 00 00 00 00 00 00 00 00 00 00 00 00 00	16.4 8 1,799 0.0 609.5 0.0 48.9 0.0 0.0 108.5 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	382.1 74,435 2017 GWh 7430.3 00 29206.8 0.0 1706.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.7 3 1,992 Million \$ 267.1 0.0 873.1 0.0 49.4 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	368.2 78,405 2018 GWh 7799.3 0.0 31099.1 0.0 1708.2 0.0 0.0 0.0 4109.8 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	19.0 8 1,833 Million S 281.7 0.0 929.4 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	354.3 76,374 2011 3074.3 0.0 33012.3 0.0 1708.3 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0
Simple cycle gas Simple cycle gas Simple cycle oli (b) Combined cycle oli Somali hydro - Optimal Biomass steam Coel - Pressurized Flui Biomass gasifier turbin Gas cogen retrofits Large Hydro Coal gasifier turbine Small Hydro - lower gra Waste fuels cogen Wind Generators Fuel Cell Cogenerator Tude Lower	558.5 44,712 8 2011 GWh 5338.7 00 18216.3 00 1706.2 00 00 00 00 00 00 00 00 00 0	19.0 1,222 Million \$ 186.3 0.0 544.0 0.0 47.7 0.0 0.0 0.0 0.0 0.0 0.0 0.0	534.9 67,218 2012 GWh 5681.1 0.0 19960 8 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	16.6 1,273 199.2 0.0 595.4 0.0 595.4 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	499.1 88.367 2013 <u>GWh</u> 6028.8 00 21798.0 00 1708.2 00 00 3142.8 00 00 00 00 00 00 00 00 00 0	18.6 8 1,389 Million \$ 212.4 0.0 647.9 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	449.8 70,067 2014 GWh 6382.1 0.0 23623.6 0.0 1706.2 0.0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	18.1 9 1,525 Million \$ 225.9 000 701.0 00 701.0 00 48.8 00 00 00 00 00 00 00 00 00 0	419.9 71.766 2018 GWh 6725.3 00 25468.2 00 1708.2 00 00 00 00 00 00 00 00 00 00 00 00 00	18.2 8 1,844 Million \$ 239.3 00 754.9 00 48.9 00 00 100.8 00 00 00 00 00 00 00 00 00 0	3863 73,484 2018 GWh 7070.0 00 27332.3 00 1708.2 00 00 00 00 00 00 00 00 00 00 00 00 00	16.4 8 1,799 252.9 0.0 609.5 0.0 48.9 0.0 106.5 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	382.1 74,435 2017 GWh 7430.3 00 29206.8 0.0 1706.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.7 3 1,992 Million 5 267.1 0.0 873.1 0.0 49.4 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	368.2 78,405 2018 GWh 7799.3 0.0 31099.1 0.0 1708.2 0.0 0.0 0.0 0.0 4109.8 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	19.0 8 1,933 281.7 0.0 929.4 0.0 49.5 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	354. 76,374 2011 6174. 0.0 33012.3 0.0 33012.3 0.0 0 0 0 4309.3 0.0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
Simple cycle gas Simple cycle gas Simple cycle oli (b) Combined cycle oli Small hydro - Optimal Biomass steam Coel - Pressurized Flui Biomass gasifier turbine Gas cogen retrofits Large Hydro Coal gasifier turbine Small Hydro - lower gra Waste fuels cogen Wind Generators Fuel Cell Cogenerator Fuel Cell Cogenerator	558.5 44,712 8 2011 GWh 5338.7 0.0 18216.3 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	19.0 1,222 186.3 0.0 544.0 0.0 47.7 0.0 0.0 0.0 0.0 0.0 0.0 0.0	534.9 67,218 2012 GWh 5681.1 0.0 19990.8 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	16.6 1.273 Million S 0.0 595.4 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	499.1 88.367 2013 GWh 6028.8 0.0 21798.0 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.6 8 1,369 212.4 0.0 647.9 0.0 647.9 0.0 647.9 0.0 647.9 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	449.8 70,067 2014 GWh 6382.1 0.0 23623.6 0.0 1708.2 0.0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	18.1 1,525 Million \$ 225.9 000 701.0 0.0 48.8 0.0 0.0 0.0 0.0 0.0 0.0 0.0	419.9 71,765 2015 GWh 6725.3 0.0 25468.2 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.2 8 1,644 Million 5 239.3 000 754.9 00 48.9 00 00 00 00 00 00 00 00 00 0	366.3 73,484 2016 GWh 7070.0 0.0 27332.3 0.0 27332.3 0.0 27332.3 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	16.4 8 1,799 00 509.5 00 48.9 00 00 00 00 00 00 00 00 00 0	382.1 74,435 2017 GWh 7430.3 0.0 29208.6 0.0 29208.6 0.0 29208.6 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.7 3 1,992 3 1,992 3 1,992 3 0.0 3 73.1 0.0 8 73.1 0.0 8 73.1 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	368.2 78,405 2018 GWh 7799.3 0.0 31099.1 0.0 31099.1 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	19.0 8 1,933 281.7 0.0 929.4 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	354. 76,374 2011 6Wi 600 33012.3 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0
Simple cycle gas Simple cycle gas Simple cycle oli (b) Combined cycle gas Combined cycle oli Small hydro - Optimal Biomass staam Coal - Pressurized Flui Biomass gasifier turbine Gas cogen retrofits Large Hydro Coal gasifier turbine Small Hydro - lower gra Waste fuels cogen Wind Generators Fuel Cell Cogenerator Fuel Cell Cogenerator Tidal Power Parabolic-Trough Solar Scier PV	558.5 44,712 2011 - GWh 5338.7 0.0 18216.3 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	19.0 1,222 Million \$ 186.3 0.0 544.0 0.0 47.7 0.0 0.0 0.0 0.0 0.0 0.0 0.0	534.9 67,218 2012 GWh 5681.1 0.0 19990.8 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	16.8 1,273 Million S 199.2 00 595.4 00 595.4 00 00 00 00 00 00 00 00 00 0	499.1 88,367 2013 <u>GWh</u> 6028.8 00 21798.0 00 1708.2 00 00 00 00 00 00 00 00 00 0	18.6 8 1,369 Million \$ 212.4 0.0 647.9 0.0 647.9 0.0 647.9 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	449.8 70,067 2014 GWh 6382.1 0.0 23623.6 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.1 9 1,525 Million \$ 225.9 0.0 701.0 0.0 48.8 0.0 0.0 0.0 0.0 0.0 0.0 0.0	419.9 71,766 2015 GW/h 6725.3 0.0 25468.2 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.2 8 1,644 Million \$ 239.3 0.0 754.9 0.0 48.9 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	3063 73,484 2018 GWh 70700 0 0 273323 0 0 273323 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	16.4 8 1,799 000 000 000 000 000 000 000	382.1 74,435 2017 GWh 7430.3 0.0 29208.6 0.0 1706.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.7 3 1,992 3 1,992 3 1,992 3 0,0 3 73.1 0 0 8 73.1 0 0 8 73.1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	368.2 78,405 2018 GWh 7799.3 0.0 31099.1 0.0 1708.2 0.0 0.0 0.0 4109.8 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	19.0 8 1,933 281.7 0.0 929.4 0.0 49.5 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	354. 76,374 2011 GWI 6174. 0.0 33012. 0.0 33012. 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0
Simple cycle gas Simple cycle gas Simple cycle gas Combined cycle gas Combined cycle gas Combined cycle gas Combined cycle gas Combined cycle gas Combined cycle gas Small hydro - Optimat Biomass gasifier turbin Gas cogen retrofits Large Hydro Coal gasifier turbine Small Hydro - Iower gra Waste fuels cogen Wind Generators Fuel Cell Cogenerator Tidal Power Parabolic-Trough Solar Solar PV N/A	558.5 44,712 8 2011 GWh 5336.7 0.0 18216.3 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	19.0 1,222 Niillon \$ 186.3 0.0 544.0 0.0 544.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	534.9 67,218 2012 GWh 5681.1 0.0 19990.8 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	16.8 1,273 Million S 100.2 000 000 000 000 000 000 000	499.1 88,367 2013 <u>GWh</u> 6028.8 00 21798.0 00 1708.2 00 00 00 3142.8 00 00 00 00 00 00 00 00 00 0	18.6 8 1,369 Million \$ 212.4 0.0 647.9 0.0 647.9 0.0 647.9 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	449.8 70,067 2014 GWh 6382.1 0.0 23623.6 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.1 9 1,525 Million \$ 225.9 000 701.0 00 48.8 000 00 00 00 00 00 00 00 00	419.9 71,766 2018 GW/h 6725.3 0.0 25468.2 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.2 8 1,644 Million \$ 239.3 000 754.9 00 48.9 00 48.9 00 100.8 00 00 00 00 00 00 00 00 00 0	306.3 73,484 2018 GWh 7070.0 0.0 27332.3 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	16.4 8 1,799 0.0 609.5 0.0 48.9 0.0 0.0 106.5 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	382.1 74,435 2017 GWh 7430.3 0.0 29206.6 0.0 1706.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.7 3 1,992 3 1,992 3 1,992 3 267.1 0 00 873.1 0 0 873.1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	368.2 78,405 2018 GWh 7799.3 0.0 31099.1 0.0 1708.2 0.0 0.0 0.0 4109.6 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	19.0 8 1,833 281.7 0.0 929.4 0.0 49.5 0.0 0.0 119.1 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	354. 76,374 2011 GWU 8174. 00 33012. 00 1708. 00 01 4309. 00 4309. 00 00 00 00 00 00 00 00 00 00 00 00 00
Simple cycle gas Simple cycle gas Simple cycle gas Combined cycle gas Combined cycle gas Combined cycle gas Combined cycle oli Small hydro - Optimal Biomass steam Coel - Pressurized Flui Biomass gasifier turbine Gas cogen retrofits Large Hydro Coal gasifier turbine Small Hydro - tower gra Waste fuels cogen Wind Generators Fuel Cell Cogenerator Fuel Cell Cogenerator Fuel Cell Cogenerator Fuel Cell Cogenerator Solar PV NVA	558.5 44,712 8 2011 GWh 5338.7 00 18216.3 00 1708.2 00 00 00 00 00 00 00 00 00 0	19.0 1,222 Million \$ 186.3 0.0 544.0 0.0 47.7 0.0 0.0 0.0 0.0 0.0 0.0 0.0	534.9 67,218 2012 GWh 5661.1 0.0 19900 8 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	16.6 1.273 Million S 1.90.2 0.04 0.05	499.1 88,367 2013 <u>GWh</u> 6028.8 000 21798.0 00 1708.2 00 00 00 00 00 00 00 00 00 0	18.6 8 1,389 Million S 212.4 0.0 647.9 0.0 647.9 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	449.8 70,067 2014 GWh 6382.1 0.0 23623.6 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.1 9 1,525 000 701.0 00 48.8 00 00 00 95.2 00 00 00 00 00 00 00 00 00 0	419.9 71,766 2018 GWh 6725.3 0.0 25488.2 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.2 8 1,644 Million \$ 239.3 00 754.9 00 48.9 00 00 00 00 00 00 00 00 00 0	306.3 73,484 2018 GWh 7070.0 0.0 27332.3 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	16.4 8 1,799 0.0 609.5 0.0 108.5 0.0 108.5 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	382.1 74,435 2017 GWh 7430.3 00 29206.8 00 1706.2 00 00 00 00 00 00 00 00 00 00 00 00 00	18.7 3 1,992 Million \$ 267.1 0.0 873.1 0.0 49.4 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	368.2 78,405 2018 GWh 7799.3 0.0 31099.1 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	19.0 8 1,833 Million S 281.7 0.0 929.4 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	354. 76,374 2019 GWU 8174. 0.0 33012.2 0.1 1708. 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0
Simple cycle gas Simple cycle gas Simple cycle oli (b) Combined cycle oli Somil hydro - Optimal Biomass steam Coel - Pressurized Flui Biomass gasifier turbin Gas cogen retrofits Large Hydro Coal gasifier turbine Small Hydro - lower gra Waste fuels cogen Wind Generators Fuel Cell Cogenerator Tidal Power Parabolic-Trough Solar Solar PV IV/A N/A N/A	558.5 44,712 8 2011 5338.7 00 18216.3 00 1708.2 00 00 00 00 00 00 00 00 00 0	19.0 1,222 186.3 0.0 544.0 0.0 47.7 0.0 0.0 0.0 0.0 0.0 0.0 0.0	534.9 67,218 2012 GWh 5661.1 0.0 19990.8 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	16.8 1,273 199.2 0.0 595.4 0.0 595.4 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	499.1 88.367 2013 GWh 6028.8 00 21798.0 00 1708.2 00 00 00 00 00 00 00 00 00 0	18.6 8 1,369 212.4 0.0 647.9 0.0 647.9 0.0 647.9 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	449.8 70,067 2014 GWh 6382.1 0.0 23623.6 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.1 1.525 Million 5 225.9 000 701.0 0.0 48.8 0.0 0.0 0.0 0.0 0.0 0.0 0.0	419.9 71,765 2018 GWh 6725.3 0.0 25468.2 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.2 8 1,644 239.3 000 754.9 00 48.9 00 00 00 00 00 00 00 00 00 0	3063 73,484 2016 GWh 70700 00 273323 0.0 273323 0.0 273323 0.0 273323 0.0 273323 0.0 273323 0.0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	16.4 8 1,799 00 509.5 0.0 609.5 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	382.1 74,435 2017 GWh 7430.3 0.0 29208.8 0.0 29208.8 0.0 29208.8 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	18.7 3 1,992 3 1,992 367.1 0.0 873.1 0.0 49.4 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	368.2 78,405 2018 GWh 7799.3 0.0 31099.1 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	19.0 8 1,933 281.7 0.0 929.4 0.0 49.5 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	354. 76,374 2011 6W 00 33012: 0.0 33012: 0.0 33012: 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.
Simple cycle gas Simple cycle gas Simple cycle oli (b) Combined cycle oli Combined cycle oli Small hydro - Optimal Biomass steam Coel - Pressurized Flui Biomass gasifier turbine Gas cogen retrofits Large Hydro Coal gasifier turbine Small Hydro - lower gra Wind Generators Fuel Cell Cogenerator Fuel Cell Cogenerator Fuel Cell Cogenerator Fuel Cell Cogenerator Solar PV N/A N/A NA Natural Gas Coal	558.5 44,712 2011 <u>GWh</u> 5338.7 0.0 18216.3 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	19.0 1,222 Million \$ 186.3 0.0 544.0 0.0 47.7 0.0 0.0 0.0 0.0 0.0 0.0 0.0	534.9 67,218 2012 GWh 5681.1 0.0 19990.8 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	16.6 1.273 1.99.2 0.0 5.95.4 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	499.1 88,367 2013 GWh 6028.8 0.0 21798.0 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.6 8 1,369 212.4 0.0 647.9 0.0 647.9 0.0 647.9 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	449.8 70,067 2014 GWh 6382.1 0.0 23623.6 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.1 1.525 Million \$ 225.9 000 701.0 0.0 48.8 000 000 000 000 000 000 000	419.9 71,766 2015 GWh 6725.3 0.0 25468.2 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.2 8 1,644 Million \$ 239.3 0.0 754.9 0.0 48.9 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	3963 73,484 2016 GWh 7070.0 0.0 27332.3 0.0 1708.2 0.0 0.0 0.0 0.0 3716.8 0.0 0.0 3716.8 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	16.4 8 1,799 000 609.5 0.0 48.9 0.0 0.0 106.5 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	382.1 74,435 2017 GWh 7430.3 0.0 29208.6 0.0 1706.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.7 3 1,992 3 1,992 3 1,992 3 267.1 0.0 873.1 0.0 873.1 0.0 0.0 0.0 113.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	368.2 78,405 2018 GWh 7799.3 0.0 31099.1 0.0 1708.2 0.0 0.0 0.0 0.0 4109.8 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	19.0 8 1,933 281.7 0.0 929.4 0.0 49.5 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	354.3 76,374 2011 GWI 600 33012.3 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0
Simple cycle gas Simple cycle gas Simple cycle oli (b) Combined cycle oli (b) Combined cycle oli Small hydro - Optimal Biomass steam Coal - Pressurized Flui Biomass gasifier turbine Gas cogen retrofits Large Hydro Coal gasifier turbine Small Hydro - lower gra Waste fuels cogen Wind Generators Fuel Cell Cogenerator Fuel Cell Cogenerator Tidal Power Parabolic-Trough Solar Solar PV IV/A N/A Natural Gas Coal Oti	558.5 44,712 2011 -GWh 5338.7 0.0 18216.3 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	19.0 1,222 Million \$ 186.3 0.0 544.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	534.9 67,218 2012 GWh 5681.1 0.0 19990.8 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	16.8 1,273 1990.2 000 596.4 00 596.4 00 00 00 00 00 00 00 00 00 0	499.1 88.367 2013 <u>GWh</u> 6028.8 00 21798.0 00 1708.2 00 00 00 00 00 00 00 00 00 0	18.6 8 1,369 Million \$ 212.4 0.0 647.9 0.0 647.9 0.0 647.9 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	449.8 70,067 2014 GWh 6382.1 0.0 23623.6 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.1 1.525 Million \$ 225.9 0.0 701.0 0.0 48.8 0.0 0.0 0.0 0.0 0.0 0.0 0.0	419.9 71,768 2015 GW/h 6725.3 0.0 25468.2 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.2 8 1,644 Million \$ 239.3 000 754.9 00 48.9 00 00 00 00 00 00 00 00 00 0	3963 73,484 2018 GWh 70700 00 273323 00 1708.2 00 00 00 00 00 00 00 00 00 00 00 00 00	16.4 8 1,799 000 609.5 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	382.1 74,435 2017 GWh 7430.3 0.0 29208.6 0.0 1706.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.7 3 1,992 3 1,992 3 1,992 3 1,992 3 0,0 3 7,1 3 0,0 3	368.2 78,405 2018 GWh 7799.3 0.0 31099.1 0.0 1708.2 0.0 0.0 0.0 4109.8 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	19.0 8 1,833 281.7 0.0 929.4 0.0 49.5 0.0 0.0 0.0 119.1 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	354. 76,374 2011 GWI 8174. 0.0 33012.3 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0
Simple cycle gas Simple cycle gas Simple cycle gas Combined cycle gas Combined cycle gas Combined cycle oli Small hydro - Optimal Biomass steam Coel - Pressurized Flui Biomass gasifier turbin Gas cogen retrofits Large Hydro - tower gra Waste fuels cogen Wind Generators Fuel Cell Cogenerator Tidal Power Parabolic-Trough Solar Solar PV NVA Natural Gas Coel Oil Hydro	558.5 44,712 8 2011 GWh 5336.7 0 0 18216.3 0 0 1708.2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	19.0 1,222 Million \$ 186.3 0.0 544.0 0.0 544.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	534.9 67,218 2012 GWh 5681.1 0 0 0 19990 8 0 0 0 1708.2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	16.6 1.273 Million S 1.002 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.0000 0.00000 0.00000 0.00000 0.00000000	499.1 88,367 2013 <u>GWh</u> 6028.8 000 21798.0 00 1708.2 00 00 00 00 00 00 00 00 00 0	18.6 8 1,369 Million \$ 212.4 0.0 647.9 0.0 647.9 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	449.8 70,067 2014 GWh 6382.1 0.0 23623.6 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.1 9 1,525 Million \$ 225.9 000 701.0 00 48.8 00 00 00 00 00 00 00 00 00 0	419.9 71,766 2018 GW/h 6725.3 0.0 25488.2 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.2 8 1,644 Million \$ 239.3 000 754.9 00 48.9 00 48.9 00 00 00 00 00 00 00 00 00 0	3063 73,484 2018 GWh 70700 00 273323 00 1708.2 00 00 00 00 00 00 00 00 00 00 00 00 00	16.4 8 1,799 0.0 609.5 0.0 106.5 0.0 106.5 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	382.1 74,435 2017 GWh 7430.3 0.0 29206.6 0.0 1706.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.7 3 1,992 Million \$ 267.1 0.0 873.1 0.0 873.1 0.0 0.0 0.0 113.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	368.2 78,405 2018 GWh 7799.3 0.0 31099.1 0.0 1708.2 0.0 0.0 0.0 4109.6 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	19.0 8 1,833 281.7 0.0 929.4 0.0 49.5 0.0 0.0 119.1 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	354.3 76,374 2011 GWI 8174.1 0.0 0.0 1706.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0
Simple cycle gas Simple cycle gas Simple cycle oli (b) Combined cycle oli (b) Combined cycle oli Small hydro - Optimal Biomass steam Coal - Pressurized Flui Biomass gasifier turbine Gas cogen retrofits Large Hydro Coal gasifier turbine Small Hydro - tower gra Wind Generators Fuel Cell Cogenerators Fuel Cell Cogenerators Fuel Cell Cogenerators Fuel Cell Cogenerators Fuel Cell Cogenerators Fuel Cell Cogenerators Solar PV N/A N/A N/A N/A N/A N/A N/A N/A N/A N/A	558.5 44,712 8 2011 GWh 5338.7 00 18216.3 00 1708.2 00 00 00 00 00 00 00 00 00 0	19.0 1,222 186.3 0.0 544.0 0.0 47.7 0.0 0.0 0.0 77.3 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	534.9 67,218 2012 GWh 5681.1 0.0 19990.8 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	16.6 1.273 Million S 1.902	499.1 88,367 2013 GWh 6028.8 000 21798.0 00 1708.2 00 00 00 00 00 00 00 00 00 0	18.6 8 1,389 Million S 212.4 0.0 647.9 0.0 48.7 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	449.8 70,067 2014 GWh 6382.1 0.0 23623.6 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.1 9 1,525 000 701.0 00 48.8 00 00 00 00 00 00 00 00 00 0	419.9 71,766 2018 GWh 6725.3 0.0 25488.2 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.2 8 1,644 Million \$ 239.3 000 754.9 00 48.9 00 00 00 00 00 00 00 00 00 0	306.3 73,484 2018 GWh 7070.0 0.0 27332.3 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	16.4 8 1,799 0.0 609.5 0.0 0.0 106.5 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	382.1 74,435 2017 GWh 7430.3 0.0 29206.8 0.0 1706.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.7 3 1,992 Million 5 267.1 0.0 873.1 0.0 49.4 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	368.2 78,405 2018 GWh 7799.3 0.0 31099.1 0.0 1708.2 0.0 0.0 0.0 4109.8 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	19.0 9 1,933 281.7 0.0 929.4 0.0 929.4 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	354. 76,374 2011 6174. 00 33012.3 0.0 33012.3 0.0 00 4309.3 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0
Simple cycle gas Simple cycle gas Simple cycle oli (b) Combined cycle oli (b) Combined cycle oli Small hydro - Optimal Biomass steam Coel - Pressurized Flui Biomass gasifier turbin Gas cogen retrofits Large Hydro Coal gasifier turbine Small Hydro - lower gra Wind Generators Fuel Cell Cogenerator Tidal Power Parabolic-Trough Solar Solar PV N/A N/A N/A N/A Natural Gas Coal Oti Hydro Nuclear Biomass	558.5 44,712 8 2011 5338.7 00 18216.3 00 1708.2 00 00 00 00 00 00 00 00 00 0	19.0 1,222 186.3 0.0 544.0 0.0 47.7 0.0 0.0 0.0 0.0 0.0 0.0 0.0	534.9 67,218 2012 GWh 5661.1 0.0 19990.6 0.0 19990.6 0.0 0.0 0.0 0.0 0.0 0.0 0.0	16.6 1.273 1.99.2 0.0 5.95.4 0.0 5.95.4 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	499.1 88.367 2013 GWh 6028.8 00 21798.0 00 1708.2 00 00 00 01708.2 00 00 00 00 00 00 00 00 00 0	18.6 8 1,369 212.4 0.0 647.9 0.0 647.9 0.0 647.9 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	449.8 70,067 2014 GWh 63821 0.0 23623.6 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.1 1.525 Million 5 225.9 000 701.0 0.0 48.8 00 00 00 00 00 00 00 00 00 0	419.9 71,765 2018 GWh 6725.3 0.0 25468.2 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.2 8 1,644 Million 5 239.3 000 754.9 00 48.9 00 00 00 00 00 00 00 00 00 0	3063 73,484 2016 GWh 7070.0 0 0 27332.3 0.0 27332.3 0.0 27332.3 0.0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	16.4 8 1,799 00 509.5 00 609.5 00 48.9 00 00 00 00 00 00 00 00 00 0	382.1 74,435 2017 GWh 7430.3 0.0 29208.8 0.0 1706.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	18.7 3 1,992 Million \$ 267.1 0.0 873.1 0.0 49.4 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	368.2 78,405 2018 GWh 7799.3 0.0 31099.1 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	19.0 9 1,933 281.7 0.0 929.4 0.0 49.5 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	354.3 76,374 2011 GW1 8174.1 0.0 33012.3 0.0 33012.3 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0

t

	2004	1	2006	1	2005	t	2007	1	2008)	2009	1	2010	}
Million C	CWh	Million C	GWh	Million S	GWh	Million \$	GWh	Million S	GWh	Million S	GWh	Million \$	GWh	Million S
105.6	3131.4	117.0	3375.5	128 6	3704.7	142 6	4043 1	156.3	4379.9	170.9	4893.4	185.0	5006.2	199.2
0.0	0.0	0.0	0.0	0.0	00	00	00	00	0.0	0.0	0.0	0.0	0.0	0.0
209 8	7735 0	242.4	8773.5	276.4	10258.1	325.9	11771.4	372.6	13304.5	421.1	14898.2	471.8	16505.7	523.0
0.0	0.0	00	0.0	0.0	0.0	00	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
42.6	1708.2	43.0	1708.2	45.3	1708.2	46 0	1708.2	47.1	1708.2	47.4	1/08.2	48.0	1/08.2	49.0
00	0.0	00	0.0	0.0	0.0	0.0	0.0	0.01	0.0	0.0	0.0	0.01	0.0	0.0
00	00	0.0	0.0	0.01	0.0	0.01	0.0	0.01	0.0	0.0	0.0	0.0	0.0	0.0
	15420	40.0	1400.5	44.8	1871 7	50.4	2051.8	50.0	2230.8	61.9	2408.8	68.5	2585.7	75.4
35.7	1302.0		00	0.01	0.0	0.0	00	0.0	0.0	0.0	0.0	0.0	0.0	0.0
00	0.0	0.0	0.0	0.0	0.0	0.0	0.0	00	0.0	0.0	0.0	0.0	0.0	0.0
00	0.0	00	00	0.0	0.0	0 0	00	00	0.0	0.0	0.0	0.0	0.0	0.0
00	0.0	0.0	0.0	0.0	0.0	00	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.0
00	0.0	00	00	00	0.0	00	00	00	0.0	0.0	0.0	0.0	0.0	0.0
00	0.0	0.0	0.0	0.0	00	0.0	00	0.01	0.0	0.01	0.0	0.0	0.0	0.0
0.0	00	0.0	0.0		0.0	0.0	00	0.0	0.0	0.0	0.0	0.0	0.0	0.0
	0.0	0.0	00	0.01	0.0	00	00	00	0.0	0.0	0.0	0.0	0.0	00
0.0	0.0	a a	00	00	00	0.01	00	00	0.0	0.0	0.0	0.0	0.0	0.0
00	0.0	0.0	0.0	0.0	0.0	0.0	0.0	00	0.0	0.0	0.0	0.0	0.0	0.0
52.4	1164.2	52.6	1125 5	52 8	1071.3	51.6	1016.9	50 0	950.2	48.4	906.9	47.5	864.7	48.5
00	0.0	0.0	0.0	0.0	0.0	0.0	00	00	0.0	0.0	0.0	0.0	0.0	0.0
5.4	88.8	5.4	64.3	5.3	80 1	51	723	47	60.1	4.0	48.5	3.2	37.5 69104.4	1775.0
1506 2	81812.0	1003.5	01225.5	1640 1	60872.6	1851.3	60480 4	10/01	60068.1	10/1.8/	0.0	1099.0	500100.4	0.0
0.0		0.0	2200	10.0	300 A	10.0	292.6	19.0	289.7	19.1	256.8	19.2	283.9	19.3
10.3	77 343	8 2 128	70 317	8 2.213	79.877	8 2 292	B1 437	1 2 3 9 2	82,992	\$ 2.445	84,546	8 2.543	86,100	8 2,642
			1 10.014									-		
		, ·	10,012			<u></u>								
	2020	<u> </u>	2021		2022		2023	Million \$	2024	Million \$	2025 GWb	Million S		
Million \$	2020 GWh	Million \$	2021 GWh	Million \$	2022 GWh	Million \$	2023 GWh 8259.7	Million \$	2024 GWh	Million \$	2025 GWh 9056.8	Million \$		
Million \$	2020 GWh 8556.6	Million \$	2021 GWh 6938.8		2022 GWh 9321.1 0 0	Million \$ 350 3 0 0	2023 GWh 8259.7 0.0	Million \$ 326.3 0 0	2024 GWh 8628.4 0 0	Million \$ 348.3 0.0	2025 GWh 9056.8 0.0	Million \$ 368.7 0.0		
Million \$ 296 8 D 0 986 5	2020 GWh 8556.6 0.0 34946.9	Million \$ 312.2 0.0 1044.4	2021 GWh 8938.8 00 36910.0	Million \$ 331.0 0.0 1118.2	2022 GWh 9321.1 0 0 38902 1	<u>Million \$</u> 350 3 0 0 1194.9	2023 GWh 8259 7 0 0 40923 9	Million \$ 326.3 0 0 1274 4	2024 GWh 8628.4 0 0 42972.5	Million \$ 346.3 0.0 1366.0	2025 <u>GWh</u> 9056.8 0.0 43818.6	<u>Million \$</u> 368.7 0.0 1414.7		
Million \$ 296 8 0 0 988 5 0 0	2020 GWh 8556.6 0.0 34946.9 0.0	Million \$ 312 2 0.0 1044.4 0 0	2021 GWh 9938.8 00 36910.0 00	<u>Million \$</u> 331.0 0.0 1118 2 0.0	2022 GWh 9321.1 0 0 38902 1 0 0	Million \$ 350 3 0 0 1194.9 0.0	2023 GWh 8259 7 0 0 40923 9 0 0	Million \$ 326.3 00 1274.4 00	2024 <u>GWh</u> 8628.4 0 0 42972 5 0.0	<u>Million \$</u> 346.3 0.0 1368.0 0.0	2025 GWh 9056.8 0.0 43818.6 0.0	Million \$ 368.7 0.0 1414.7 0.0		
Million \$ 296 8 0 0 986 5 0 0 49 8	2020 GWh 8556.6 0.0 34948.9 0.0 1708.2	Million \$ 312.2 0.0 1044.4 0 0 49.7	2021 GWh 6938.8 0 0 36910.0 0 0 1708 2	Million \$ 331.0 0.0 1118 2 0.0 50.4	2022 GWh 9321.1 0 0 38902 1 0 0 1708.2	Million \$ 350 3 0 0 1194 9 0 0 51.2	2023 GWh 8259.7 0.0 40923.9 0.0 1708.2	Million \$ 328.3 00 1274.4 00 52.0	2024 GWh 8628.4 0 0 42972.5 0.0 1708.2	<u>Million \$</u> 346.3 0.0 1368.0 0.0 53 2	2025 <u>GWh</u> 9056.8 0.0 43618.6 0.0 1708.2	Million \$ 368.7 0.0 1414.7 0.0 53.9		
Million \$ 296 8 0 0 988 5 0 0 49 8 0 0	2020 GWh 8556.6 0.0 34946.9 0.0 1708.2 0.0	<u>Million \$</u> 312 2 0.0 1044.4 0 0 49.7 0.0	2021 GWh 8938.8 00 36910.0 00 1708.2 00	Million \$ 331.0 0.0 1118 2 0.0 50.4 0.0	2022 GWh 9321.1 0 0 38902 1 0 0 1708.2 0 0	Million \$ 3503 00 1194.9 00 51.2 00	2023 <u>GWh</u> 8259 7 0 0 40923 9 0.0 1708 2 0.0	Million \$ 326.3 00 1274.4 00 520 00	2024 GWh 8628.4 0 0 42972.5 0.0 1708.2 0.0	Million \$ 346.3 0.0 1368.0 0.0 53 2 0.0	2025 <u>GWh</u> 9056.8 0.0 43618.6 0.0 1708.2 0.0	Million \$ 368.7 0.0 1414.7 0.0 53.9 0.0		
Million \$ 296 8 0 0 986 5 0 0 49 8 0 0 0 0	2020 GWh 8556.6 0.0 34946.9 00 1708.2 00 00	<u>Million</u> \$ 312 2 0.0 1044.4 0 0 49 7 0.0 0 0	2021 2021 30916 36910.0 00 1708.2 00 00	Million \$ 331.0 00 1118 2 0.0 50.4 00 00	2022 GWh 9321.1 00 38902.1 00 1708.2 00 00	Million \$ 350 3 0 0 1194 9 0 0 51 2 0 0 0 0	2023 GWh 8259 7 0 0 40P23 9 0.0 1708 2 0.0 0.0 0.0	Million \$ 326.3 00 1274 4 00 52 0 00 00	2024 GWh 8628.4 00 42972 5 0.0 1708 2 0.0 0 0 0	Million \$ 348.3 0.0 1368.0 0.0 53.2 0.0 0.0 0.0	2025 GWh 9056.8 0.0 43618.6 0.0 1706.2 0.0 0.0 0.0	<u>Million \$</u> 368.7 0.6 1414.7 0.0 53.9 0.0 0.0 0.0		
Million \$ 296 8 00 968 5 00 49 8 00 00 00 00 00 00 00 00 00 00 00 00 00	2020 GWh 8556.6 0.0 34946.9 00 1708.2 0.0 0.0 0.0	Million \$ 312 2 0.0 1044.4 0 0 49 7 0.0 0 0 0 0	2021 GWh 6938 8 00 36910 0 00 1708 2 00 00 00	Million 3 331.0 00 1118 2 0.0 50.4 00 00 00	2022 GWh 9321.1 00 38902.1 00 1708.2 00 00 00 00	Million \$ 350 3 0 0 1194 9 0 0 51.2 0 0 0 0 0 0 0 0 0 0 0 0	2023 GWh 82597 00 409239 00 17082 00 00 00 00	Million \$ 326 3 00 1274 4 00 52 0 00 00 156 1	2024 GWh 8628.4 00 42072.5 0.0 1708.2 0.0 00 00 00 00	<u>Million \$</u> 3463 0.0 1368.0 0.0 53.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	2025 GWh 9056.8 0.0 43618.6 0.0 1706.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	<u>Million 5</u> 368.7 0.0 1414.7 0.0 53.9 0.0 0.0 0.0 211.6		
Million 5 296 8 0 0 988 5 0 0 49 8 0 0 0 0 0 0 0 0 125 1	2020 GWh 85566 0.0 34946.9 0.0 1706.2 0.0 0.0 0.0 0.0 4511.0	<u>Million \$</u> 312 2 0.0 1044.4 0 0 49 7 0.0 0.0 0.0 131.2 0.0	2021 2021 3938 8 00 36910 0 00 1708 2 00 00 00 24715 1	<u>Million 3</u> 331.0 0.0 1118 2 0.0 50.4 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	2022 GWh 9321.1 00 38902 1 00 1708.2 00 00 00 4921.5	Million \$ 350 3 00 1194 9 00 51.2 00 00 00 147.5 00	2023 <u>GWh</u> 82597 00 40923 9 00 1708 22 00 00 00 5130 3 00	Million \$ 326 3 0 0 1274 4 0 0 52 0 0 0 0 0 0 0 156 1 0 0	2024 2024 GWh 6628.4 0 0 42972.5 0.0 1708 2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.	<u>Million \$</u> 346.3 0.0 1366.0 0.0 53.2 0.0 0.0 0.0 0.0 166.2 0.0	2025 GWh 9056 8 0.0 43818.6 0.0 1708 2 0.0 0.0 0.0 0.0 0.0 0.0 6701.4 0.0	Million \$ 368.7 0.0 1414.7 0.0 53.9 0.0 0.0 0.0 211.6 0.0		
Million \$ 296 8 00 968 5 00 49 8 00 00 125 1 00	2020 GWh 85566 0.0 34946.9 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	Million \$ 312 2 000 1044.4 00 497 0.0 00 131.2 000 131.2 000 00 00 00 00 00 00 00 00 00 00 00 0	2021 30010 0 36910 0 0 0 1708 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	<u>Million \$</u> 331.0 0.0 1118 2 0.0 50.4 0.0 0.0 0.0 139 2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.	2022 GWh 9321.1 00 38902 1 00 1708.2 00 00 00 4921.5 00 00	Million \$ 350 3 0 0 1194 9 0 0 51.2 0 0 0 0 0 0 147.5 0 0 0 0	2023 <u>GWh</u> 82597 00 40923 9 00 1708 2 00 00 00 5130 3 00 00	Million \$ 326 3 0 0 1274 4 0 0 52 0 0 0 0 0 0 0 0 0 156 1 0 0 0 0	2024 2024 GWh 8628.4 0 0 42972.5 0.0 1708 2 0.0 0 0 0 0 0 0 0 0 5341.5 0 0 0 0	Million \$ 348.3 0.0 1368.0 0.0 532. 0.0 0.0 0.0 0.0 0.0 0.0 0.0 168.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	2025 GWh 9056 8 0.0 43818.6 0.0 1708 2 0.0 0.0 0.0 0.0 0.0 6701.4 0.0 0.0 0.0	Million \$ 368.7 0.0 1414.7 0.0 53.9 0.0 0.0 0.0 211.6 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0		
Million \$ 200 8 00 000 498 00 00 00 00 00 00 00 00 00 00 00 00 00	2020 GWh 85566 0.0 34946.9 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	<u>Million</u> \$ 312 2 0.0 1044.4 0 0 49 7 0.0 0 0 0 0 131.2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2021 2021 GW/h 80388 00 369100 00 1708 2 00 00 00 00 00 00 00 00 00 0	<u>Million \$</u> 331.0 0.0 1118 2 0.0 50.4 0 0 0 0 0 0 139 2 0 0 0 0 0 0 0 0 0 0 0 0 0	2022 GWh 9321.1 00 38902 1 00 1708.2 00 00 00 4921.5 00 00 4921.5 00 00	Million \$ 00 1194.9 00 51.2 00 00 147.5 00 00 00 00 00 00 00 00 00 00 00 00 00	2023 <u>GWh</u> 82597 00 40923 9 00 1708 2 00 00 00 5130 3 00 00 00 00 00 00 00 00 00 0	Million \$ 326 3 0 0 1274 4 0 0 52 0 0 0 0 0 0 0 156 1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2024 GWh 8628.4 00 42972.5 0.0 1708.2 0.0 00 00 5341.5 00 00 00 00 00	Million \$ 348.3 0.0 1368.0 0.0 53 2 0.0 0 0 0 0 0 0 0 0 0 168.2 0.0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2025 GWh 9056 8 0.0 43818.6 0.0 1708 2 0.0 0.0 0.0 0.0 6701.4 0.0 0.0 0.0 0.0 0.0	Million \$ 369.7 0.0 1414.7 0.0 53.9 0.0 0.0 0.0 211.6 0.0 0.0 0.0 0.0 0.0 0.0 0.0		
Million \$ 296 8 296 8 00 668 5 00 49 8 00 00 125 1 00 00 00 00 00 00 00 00 00 00 00 00 00	2020 GWh 8556 6 00 34946 9 00 1706 2 00 00 00 4511.0 00 4511.0 00 00 00 00 00 00	<u>Million \$</u> 312 2 0.0 1044.4 00 497 0.0 00 00 00 00 00 00 00 00 00 00 00 00	2021 2021 GW/h 80388 00 369100 00 17082 00 00 00 00 00 00 00 00 00 0	Million \$ 3310 1118 2 00 1118 2 00 50.4 00 00 139 2 00 00 00 00 00 00 00 00 00	2022 GWh 9321.1 00 38902 1 00 1708.2 00 1708.2 00 00 00 4921.5 00 00 00 00	Million \$ 00 1194.9 00 51.2 00 00 147.5 00 00 00 00 00 00 00 00 00 00 00 00 00	2023 <u>GWh</u> 82597 00 409239 00 17082 00 00 00 51303 00 00 00 00 00 00 00 00 00	Million \$ 3263 00 1274 4 00 520 00 520 00 00 00 156 1 00 00 00 00 00 00 00 00 00 00 00 00 00	2024 GWh 8628.4 00 42972.5 0.0 1708.2 0.0 00 5341.5 00 00 5341.5 00 0.0	Million \$ Millio	2025 GWh 9056 8 0.0 43818.6 0.0 1708 2 0.0 0.0 0.0 6701.4 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	Million \$ 369.7 0.0 1414.7 0.0 53.9 0.0 0.0 0.0 0.0 0.0 211.6 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0		
Million \$ 296 8 296 8 00 968 5 00 49 6 00 125 1 00 00 00 00 00 00 00 00 00 00 00 00 00	2020 GWh 8556 6 00 34946 9 00 1708 2 00 00 00 4511.0 00 4511.0 00 00 00 00 00 00 00 00 00 00 00 00 0	Million \$	1000000000000000000000000000000000000	Million 3 331.0 00 1118 2 00 50.4 00 00 139 2 00 00 00 00 00 00 00 00 00 00 00 00 00	2022 GWh 9321 1 00 38902 1 00 1708 2 00 1708 2 00 00 4921 5 00 00 4921 5 00 00 00 00 00 00	Million 5 350 3 0 0 1194 9 0 0 51 2 0 0 0 0 0 0 147 5 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2023 GWh 8259 7 00 40923 9 00 1708 2 00 00 00 00 5130 3 00 00 00 00 00 00 00 00 00 0	Million \$ 326.3 00 1274.4 00 52.0 00 156.1 00 156.1 00 00 00 00 00 00 00 00 00 00 00 00 00	2024 2024 2072 5 00 42072 5 00 1708 2 00 1708 2 00 00 00 00 00 00 00 00 00 0	Million \$ 348.3 0.0 1368.0 0.0 53.2 0.0 0.0 0 166.2 0.0 166.2 0.0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2025 GWh 9058 8 0.0 43818.6 0.0 1708 2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.	Millon \$ 388.7 0.6 1414.7 0.0 53.9 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0		
Million \$ 296 8 296 8 00 968 5 00 49 8 00 125 1 00 00 00 00 00 00 00 00 00 00 00 00 00	2020 GWh 8550.6 0.0 34946.9 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	Million \$	1000000000000000000000000000000000000	Million 3 331.0 00 1118 2 0.0 50.4 00 00 1139 2 00 139 2 00 00 00 00 00 00 00 00 00 00 00 00 00	2022 GWh 9321 1 00 38902 1 00 1708 2 00 00 4921 5 00 00 4921 5 00 00 00 00 00 00	Million 5 350 3 0 0 1194 9 0 0 51 2 0 0 0 0 0 0 147.5 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2023 GWh 8259 7 00 40923 9 00 1708 2 00 00 00 00 00 00 00 00 00 0	Million \$ 2833 00 1274 4 00 520 00 156 0 00 156 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2024 2024 2072 5 0.0 1708 2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.	Million \$ Million \$ 346.3 0.0 1368.0 0.0 53.2 0.0 00 166.2 0.0 166.2 0.0 00 00 00 00 00 00 00 00 00 00 00 00	2025 GWh 9058 8 0.0 43818.6 0.0 1708 2 0.0 0.0 0.0 6701 4 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.	Million \$ 388.7 0.0 1414.7 0.0 53.9 0.0 0.0 0.0 211.6 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0		
Million \$ 296 8 00 966 5 00 49 8 00 125 1 00 00 00 00 00 00 00 00 00 00 00 00 00	2020 <u>GWh</u> 8550.6 0.0 34946.9 0.0 1708.2 0.0 0.0 0.0 4511.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	Million \$ 312 2 0.0 1044.4 0 497 0.0 0.0 1044.4 0 0.0 0.0 131.2 0.0	1000000000000000000000000000000000000	Million 3 331.0 00 1118 2 0.0 50.4 00 00 1139 2 00 00 00 00 00 00 00 00 00 00 00 00 00	2022 GWh 9321.1 00 38902 1 00 1708.2 00 00 4921.5 00 4921.5 00 00 00 00 00 00 00	Million \$ 350 3 0 0 1194 9 0 0 51 2 0 0 0 0 0 0 147.5 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2023 GWh 8259 7 00 40923 9 00 1708 2 00 00 00 00 00 00 00 00 00 0	Million \$ 263 3263 00 1274 4 00 520 00 1561 00 1561 00 00 00 00 00 00 00 00 00 00 00 00 00	2024 2024 20Wh 8628.4 0 0 42072.5 0.0 1708 2 0.0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Million \$ 346.3 0.0 1368.0 0.0 53.2 0.0 0.0 0 0 168.2 0.0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2025 GWh 9056.8 0.0 43618.8 0.0 1708.2 0.0 0.0 0.0 6701.4 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	Million \$ 386.7 0.0 1414.7 0.0 53.9 0.0 0.0 211.6 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0		
Million 1 2968 5 00 498 00 125 1 00 00 125 1 00 00 00 00 00 00 00 00 00 00 00 00 00	2020 <u>GWh</u> 8550.6 0.0 34946.9 0.0 1708.2 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	Million \$ 312 2 0.0 1044.4 0 0 1044.4 0 0 1044.4 0	1000000000000000000000000000000000000	Million 3 331.0 331.0 0.0 1118 2 0.0 50.4 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	2022 GWh 9321.1 00 38902 1 00 1708.2 00 00 4921.5 00 4921.5 00 00 00 00 00 00 00 00 00 00	Million \$ 350 3 00 1194 9 00 51 2 00 147.5 00 147.5 00 00 00 00 00 00 00 00 00 00 00 00 00	2023 GWh 8259 7 00 40923 9 00 1708 22 00 00 00 00 00 00 00 00 00	Million \$ 3263 00 12744 00 520 00 00 1561 00 00 00 00 00 00 00 00 00 00 00 00 00	2024 2024 GWh 6628.4 0 0 42972.5 0.0 1708 2 0.0 0 0 0 0 5341.5 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Million \$ 346.3 0.0 1366.0 0.0 53.2 0.0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2025 GWh 9056 8 0.0 43818.8 0.0 1708 2 0.0 0.0 0.0 6701.4 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	Million \$ 386.7 0.0 1414.7 0.0 53.9 0.0 0.0 211.6 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0		
Million \$ 296 8 00 668 5 00 49 8 00 00 125 1 00 00 00 00 00 00 00 00 00 00 00 00 00	2020 <u>GWh</u> 85506 00 349469 00 17082 00 17082 00 4511.0 00 4511.0 00 00 00 00 00 00 00 00 00	Million \$ 312 2 00 1044.4 00 1044.4 00 00 1044.4 00 00 00 131.2 00	1000 1000 2021 GWh 309388 00 36910.0 00 1708.2 00 000 00 1708.2 00 000 00 000 00 000 00 000 00 000 00 000 00 000 00 000 00 000 00 000 00 000 00 000 00 000 00 000 00 000 00 000 00 000 00 000 00	Million 3 331.0 1118 2 0.0 50.4 00 00 139 2 0.0 00 00 00 00 00 00 00 00 00 00	2022 GWh 9321.1 00 38902 1 00 1708.2 00 00 4921.5 00 00 4921.5 00 00 00 00 00 00 00 00 00	Million \$ 350 3 00 1194 9 00 51 2 00 00 147.5 00 00 00 00 00 00 00 00 00 00 00 00 00	2023 GWh 82597 00 40923 9 00 1708 22 00 00 00 00 00 00 00 00 00	Million \$ 263 3263 00 12744 00 520 00 1561 00 00 00 00 00 00 00 00 00 00 00 00 00	2024 2024 GWh 6628.4 0 0 42972.5 0.0 1708 2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.	Million \$ 346.3 0.0 1366.0 0.0 53.2 0.0 00 166.2 0.0 00 166.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	2025 GWh 9056 8 0.0 43818.6 0.0 1708 2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.	Miilion \$ 388.7 0.0 1414.7 0.0 53.9 0.0 0.0 211.6 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0		
Million \$ 200 8 00 00 00 49 0 00 00 00 00 00 00 00 00 00 00 00 00	2020 GWh 8556 6 00 34946 9 00 1706 2 00 00 4511.0 00 4511.0 00 00 00 00 00 00 00 00 00 00 00 00 0	Million \$ 312 2 00 1044.4 00 1044.4 00 1034.4 00	2021 GWh 9938.8 00 36910.0 00 1708.2 00 00 00 00 00 000 00	Million \$ 331.0 100 1118 2 0.0 50.4 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	2022 GWh 9321.1 00 38902 1 00 1708.2 00 00 00 4921.5 00 00 4921.5 00 00 00 00 00 00 00 00 00 00 00 00	Million \$ 350 3 350 3 00 1194 9 00 51 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2023 GWh 82597 00 40923 9 00 1708 2 00 00 00 00 5130 3 00 00 00 00 00 00 00 00 00 0	Million \$	2024 GWh 8628.4 0 0 42972.5 0.0 1708 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Million \$ 0.0 1368.0 0.0 1368.0 0.0 532.0 0.0 00 168.2 0.0 00 168.2 0.0 00 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.	2025 GWh 9056 8 0.0 43818.6 0.0 1708 2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.	Million \$ 388.7 0.0 1414.7 0.0 53.9 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0		
Million \$ 296 8 296 8 00 668 5 00 49 8 00 49 8 00 00 125 1 00 00 00 00 00 00 00 00 00 00 00 00 00	2020 GWh a556.6 0.0 34946.9 0.0 1706.2 0.0 0.0 0.0 4511.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	Million \$ 312 2 00 1044.4 00 1044.4 00 1041.0 00	1000000000000000000000000000000000000	Million 3 331.0 00 1118 2 00 50.4 00 00 139 2 00 139 2 00 00 00 00 00 00 00 00 00 0	2022 GWh 9321 1 00 38902 1 00 1708 2 00 00 4921 5 00 00 4921 5 00 00 00 00 00 00 00 00 00 00 00 00 00	Million \$ 300 1194 0 0 1194 0 0 512 0 0 0 0 0 0 147.5 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2023 GWh 8259 7 00 40923 9 00 1708 2 00 00 00 00 00 00 00 00 00 0	Million \$ 326 3 00 1274 4 00 52 0 00 156 1 00 156 1 00 00 00 00 00 00 00 00 00 00 00 00 00	2024 2024 300 8628.4 00 42072.5 0.0 1708.2 0.0 00 00 00 00 00 00 00 00 0	Million \$ 348.3 0.0 1368.0 0.0 53.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	2025 GWh 9058 8 0.0 43818.6 0.0 1708 2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.	Million \$ 388.7 0.6 1414.7 0.0 53.9 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		
Million \$ 296 8 296 8 00 688 5 00 49 8 00 49 8 00 00 125 1 00 00 00 00 00 00 00 00 00 00 00 00 00	2020 GWh 8556 6 00 34946 9 00 1708 2 00 00 4511.0 00 4511.0 00 00 00 00 00 00 00 00 00	Million \$ 312 2 00 1044.4 00 497 00 01 02 00 01 01 02 00 00 01 02 031 2 00 00 01 02 03 04 05 06 07 08 09 00 <td>1000000000000000000000000000000000000</td> <td>Million 3 331.0 00 11182 00 50.4 00 00 1392 00 00 00 00 00 00 00 00 00 0</td> <td>2022 GWh 9321 1 00 38902 1 00 1708 2 00 00 4921 5 00 00 4921 5 00 00 00 00 00 00 00 00 00 00 00 00 00</td> <td>Million 5 350 3 0 0 1194 9 0 0 51 2 0 0 51 2 0 0 0 0 0 0 147 5 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0</td> <td>2023 GWh 8259 7 00 40923 9 00 1708 2 00 00 00 00 00 00 00 00 00 0</td> <td>Million \$ 3263 3263 00 12744 00 520 00 00 1561 00 00 00 00 00 00 00 00 00 00 00 00 00</td> <td>2024 2024 2072 5 0.0 1708 2 0.0 1708 2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.</td> <td>Million \$ Million \$ Million \$ 346.3 0.0 1368.0 0.0 53.2 0.0 00 166.2 0.0 166.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0</td> <td>2025 GWh 9058 8 0.0 43818.8 0.0 1708 2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.</td> <td>Million 5 388.7 0.6 1414.7 0.0 53.9 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0</td> <td></td> <td></td>	1000000000000000000000000000000000000	Million 3 331.0 00 11182 00 50.4 00 00 1392 00 00 00 00 00 00 00 00 00 0	2022 GWh 9321 1 00 38902 1 00 1708 2 00 00 4921 5 00 00 4921 5 00 00 00 00 00 00 00 00 00 00 00 00 00	Million 5 350 3 0 0 1194 9 0 0 51 2 0 0 51 2 0 0 0 0 0 0 147 5 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2023 GWh 8259 7 00 40923 9 00 1708 2 00 00 00 00 00 00 00 00 00 0	Million \$ 3263 3263 00 12744 00 520 00 00 1561 00 00 00 00 00 00 00 00 00 00 00 00 00	2024 2024 2072 5 0.0 1708 2 0.0 1708 2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.	Million \$ Million \$ Million \$ 346.3 0.0 1368.0 0.0 53.2 0.0 00 166.2 0.0 166.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	2025 GWh 9058 8 0.0 43818.8 0.0 1708 2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.	Million 5 388.7 0.6 1414.7 0.0 53.9 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0		
Million \$ 296 8 296 8 00 968 5 00 49 6 00 49 6 00 125 1 00 00 00 00 00 00 00 00 00 00 00 00 00	2020 GWh 8556.6 0.0 34946.9 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	Million £ 312 2 00 1044.4 00 1044.4 00 1014.4 00	1000000000000000000000000000000000000	Million 3 331.0 00 1118 2 0.0 50.4 00 00 1118 2 0.0 00 00 00 00 00 00 00 00 0	2022 GWh 9321 1 00 38902 1 00 1708 2 00 00 4921 5 00 00 4921 5 00 00 00 00 00 00 00 00 00 00 00 00 00	Million 5 350 3 0 0 1194 9 0 0 51 2 0 0 0 0 0 0 147.5 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2023 GWh 8259 7 00 40923 9 00 1708 2 00 00 00 00 00 00 00 00 00 0	Million \$ 263 3263 00 12744 00 520 00 1561 00 1561 00 00 00 00 00 00 00 00 00 00 00 00 00	2024 2024 2072 5 00 42072 5 00 1708 2 00 00 00 00 00 00 00 00 00 0	Million \$ Million \$ 346.3 0.0 1368.0 0.0 53 2 0.0 00 166.2 0.0 00 166.2 0.0 00 00 00 00 00 00 00 00 00 00 00 00	2025 GWh 9058 8 0.0 43818.6 0.0 1708 2 0.0 0.0 0.0 6701 4 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.	Miilion \$ 388.7 0.0 1414.7 0.0 53.9 0.0 0.0 211.6 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0		
Million \$ 296 8 296 8 00 968 5 00 49 6 00 49 6 00 125 4 00 00 00 00 00 00 00 00 00 00 00 00 00	2020 GWh 8550.6 0.0 34946.9 0.0 1708.2 0.0 0.0 0.0 4511.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0	Million £ 312 2 0.0 1044.4 00 1044.4 00 1041.4 00 00 01 131.2 00	1000000000000000000000000000000000000	Million 3 331.0 00 1118 2 0.0 50.4 00 00 139 2 00 00 00 00 00 00 00 00 00 0	2022 GWh 9321.1 00 38902.1 00 1708.2 00 00 4921.5 00 00 00 00 00 00 00 00 00 0	Million \$ 00 1194.9 00 51.2 00 147.5 00 00 147.5 00 00 00 00 00 00 00 00 00 00 00 00 00	2023 2023 6259 7 6259 7 00 40973 9 00 1708 2 00 00 00 00 00 00 00 00 00 0	Million \$ 5263 00 12744 00 520 00 1561 00 00 00 00 00 00 00 00 00 00 00 00 00	2024 2024 GWh 8628.4 0 0 42072.5 0 0 42072.5 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Million \$ Million \$ 346.3 0.0 1368.0 0.0 53 2 0.0 00 168.2 0.0 00 168.2 0.0 00 00 00 00 00 00 00 00 00 00 00 00	2025 GWh 9058 8 0.0 43618.8 0.0 1708 2 0.0 0.0 6701.4 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	Million 5 388.7 0.0 1414.7 0.0 53.9 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0		
Million \$ 296 8 296 8 00 968 5 00 49 8 00 125 1 00 00 125 1 00 00 00 00 00 00 00 00 00 00 00 00 00	2020 GWh 8550.6 0.0 34946.9 0.0 1708.2 0.0 1708.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	Million £ 312 2 0.0 1044.4 00 1044.4 00 1041.4 00 00 131.2 00 00 00 00 00 00 00 00 00 00 00 00 00 00 00 00 00 01 02 03 04 05 05 05 05 05 05 05	1000000000000000000000000000000000000	Million 3 331.0 331.0 1118 2 0.0 50.4 00 00 139 2 00 00 00 00 00 00 00 00 00 0	2022 GWh 9321.1 00 38902 1 00 1708.2 00 4921.5 00 4921.5 00 00 00 00 00 00 00 00 00 0	Million \$ 00 1194.9 00 51.2 00 147.5 00 147.5 00 00 00 00 00 00 00 00 00 00 00 00 00	2023 GWh 8259 7 00 40923 9 00 1708 2 00 00 00 00 00 00 00 00 00 0	Million \$ 3263 3263 00 12744 00 520 00 1561 00 00 00 00 00 00 00 00 00 00 00 00 00	2024 2024 GWh 8628.4 0 0 42972.5 0.0 1708 2 0.0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Million \$ 346.3 0.0 1366.0 0.0 53.2 0.0 0.0 0.0 166.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	2025 GWh 9058 8 0.0 43818.8 0.0 1708 2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.	Miilion \$ 386.7 0.0 1414.7 0.0 53.9 0.0 0.0 211.6 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0		
Million \$ 200 8 200 8 00 00 00 49 8 00 00 00 00 00 00 00 00 00 00 00 00 00	2020 GWh 8556 6 00 34946 9 00 1708 2 00 1708 2 00 1708 2 00 1708 2 00 00 00 00 00 00 00 00 00 0	Million \$ 312 2 0.0 1044.4 00 1044.4 00 1041.4 00 00 00 00 00 00 00 00 00 00 00 00 00 00 00 00 00 00 00 01 02 03 04 05	1000000000000000000000000000000000000	Million 3 331.0 331.0 0.0 1118 2 0.0 50.4 0.0 0.0 0.0 0.0 0.0 0.0 0.0	2022 GWh 9321.1 00 38902 1 00 1708.2 00 4921.5 00 4921.5 00 00 00 00 00 00 00 00 00 0	Million \$ 350 3 00 1194 9 00 51 2 00 147.5 00 00 147.5 00 00 00 00 00 00 00 00 00 00 00 00 00	2023 GWh 8259 7 00 40923 9 00 1708 22 00 00 00 00 00 00 00 00 00	Million \$ 5263 00 12744 00 520 00 1561 00 00 00 00 00 00 00 00 00 00 00 00 00	2024 2024 GWh 6628.4 0 0 42972.5 0.0 1708 2 0.0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Million \$ 346.3 0.0 1366.0 0.0 53.2 0.0 0.0 0.0 0.0 166.2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0	2025 GWh 9056 8 0.0 43818.8 0.0 1708 2 0.0 0.0 0.0 0.0 0.0 0.0 0.0 0.	Miilion \$ 386.7 0.0 1414.7 0.0 53.9 0 0 0 0.0 0.0 0.0 0 0 0 0 0 0 0 0 0 0		

	Solar PV Solar PV NIA NAA Nakaral Gas Cost Cost Cost Cost Cost Cost Cost Cos	Gas coper retrolls Large Hydro Coal gasifier turbine Small Hydro - lower Small Hydro - lower Waste fuelt congen Wate Cell Comercions Fuel Cell Comercions Fuel Cell Comercions	Simple syste get Simple syste of (b) Sombred syste of (b) Combined syste get Combined syste get Small hydre - Cythin Simall hydre - Cythin Scall - Pressurfeed for Bonnass assider for	Simple cycle (a) Simple cycle (a) Combined cycle (a) Sonabled cycle (a) Sonabled cycle (a) Sonable the Sonable of the Case segment of Case segment Sonable the Sonable of Coperation Packable-Though Sof Solar PV WA Solar PV WA Solar PV WA Solar PV WA Solar PV WA Solar PV WA Solar PV Solar PV	Generating St
	50000 0000 00000 00000 000000 000000000		2011 GMA Megn 1003 3 11118-1 11118-1 11118-1 1004 2 1415-8 1415-8 1415-8		atistics (HIS GMA MEan
			177 0 476 7 61 0 0 0 0 0		Dollary A
	812 3 57804 2 2941 4		2012 2012 2017 2017 2017 1012 11012 11012 11012		nual GMA prod. 1996 GMA Mi
-	- 70 Naŭ 10 60 000 Naŭ 10 00000				ation and Reven
	57401		1200 1 1200 1 1000 1000 1000 1 1000 1000 1000 1000 1000 1000 100000000	544	ue Received for 1997 GWA
	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0		100 100 100 100 100 100 100 100 100 100	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Electricity (million Million Dollars
	500 500 500 500 500 500 500 500 500 500	6701 4 0 0 0 0 0 0 0 0 0 0	2814 GWA 90594 17062 2005 7	7784 1165 7 1166 1 1166 1 1166 1 1166 1 1170 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	1) 1913 GWA
	1000 A A A A A A A A A A A A A A A A A A	00 00 00 00 00	Million Dodiers 1811 4 829 4 80 1 80 1 78 0 78 0 78 0		Million Dollars
	542 20 542 20 54 542 20 54 54 54 54 54 54 54 54 54 54 54 54 54		2015 2015 2015 2015 1700 2275.4 2275.4	1025 5 341 9 341 9 344 2 344 2 344 2 344 2 344 2 00 00 00 00 00 00 00 00 00 00 00 00 00	9111 9111 W
	- 00 - 00 - 00 - 00 - 00 - 00 - 00 - 00		67 (00 100 00 00 00 00 00 00 00 00 00 00 00		lan Dallers
	34730 34730 2400 2400 2400 2400 2400 2400 2400 24		2010 4051 0 10734 0 1706 2 2470 5 2470 5	1269 274 274 200 20 2	ŝÌ.
	1.00 x 0 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	00000000000000000000000000000000000000		152 152 152 152 152 152 152 152 152 152	
	55 107 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	470 00 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2017 2017 21547 21547 21547 21547 21548 25484 25484 25484	14259 73297 73297 73297 73297 73297 73297 8002 8002 8002 8002 8002 8002 8002 800	a 28 3 € 1
	2022 2022 1145	2485 000 000 000 000 000 000 000 000 000 0	Minor Dollary 213 8 0 0 0 0 0 0 0 0 0 0 0 0 0 0	10 10 10 10 10 10 10 10 10 10	illion Dollars
	54010 1 10 1 10 1 10 1 10 1 10 1 10 1 10	9701.4 9701.4 0.0 231.0 0.0 0.0 0.0	2418 0.0 23452 4 1706 2 3666 1 200 0 0 0	15710 1704 2 1704 2 170	<u>}</u>
	2011.1 1155		Willion Doller 2048 915.7 915.7 98.3 98.3		

r—			_										_		-	_	_	_	-				-	1		Т		_	-	-	_			-	-	-	_	-	-	-	-	-	_	-	-	_	-	-	-	
3112.3	54088	20	0	4365				33			, M		00	00	6701 4	21007.0	00	2007.0	1708 2	00	25415 2	00	4620 7	GWA MA	2019	14113		Lancon		00	884	0				00	00	00	00	00				2 100	1700 2	00	1029 5	00	17100	2003 GWA 144
114 8	1905 9	20	00				0			00	17 8	00	00	00	247 4	103 6	00	103 8	8	00	1 066	00	235 7	on Dollers				2 6921	11	00	56 7	0				00	00	00	0	0.0	0.0			22.2	47	00	8.58	00	8	Ion Dollars
2 1000	53579 2	185				00	00	0	0	00	450 7	00	00	707	8701 4	1 CD62	00	1 0062	1708 2	0	27304 9	•	40171	GWA	2020	NO.1		0218 1		0		000				0	00	00	0			C 7 PG		050	1708 2	00	2502 3	0	1849 8	G 100
100	1978 0				00	00	00	00	00	00	10.0	00	00	28	247.0	107 3	00	t 101	8	00		0	247 2	Million Dollars		10.00	00	1790 9	7.0	00	8				00	00	00	0	0.0			2		25.0	40 7	00	1135	00		Million Dollary
3050 4	52903 8	-			0	00	00	00	0	00	553 1	00	00	142 1	6701 4	3000 2	0	2000	1708.2	00	29378 0	00	5015 0		2421	2141.0	00	0 9000	E	00				00	00	00	0	0			6701 4	775 5	00	751 3	1708 2	00	3242 0	00		2905
00	1981			00	00	00	00	00	00	00	7 05	00	00	5 3	250 0	112.2	0	112 2		0.0	1152 2	00	10 090	Million Dollars		16.60	0.0	1055 2	•	0			000	00	00	00	0				202	20 2	00	28	51 0	00	1450			
2010	57464 0	17 0	1 180	00	00	00	00	00	0	00	857 2	00	00	214.4	1701 4	2001 5	00	30948 5			11107 0	00	53160 T			2294.4	00	50924 0	8			0		00	00	0	0				6701 4	9204	00	6 CN0	1708 2	0	4354 8		196	200
	1944.9		11	00	0	0	0	0	00	0	24 0	0	0		253.5	117 2										97.2	00	1802 5		0		000	00	00	00	0					208 8	35 4	00	2	5 2	0	102 5			
00	510304		C 10C	00	00	0	0	0	0		762 5	00		207.5											ŗ	2425.2	0	50710 2	71				00	00	0	8	5			00	6701 4	1074 7	0	1025 9	1708 2	00	5554 6	• CBCZ	GWM	2997
0		. 00	8 00	00	0	0		0	0		20 2									C DCCI			Artio Under			97 8	0	1072 0				00	00	00	00				0	00	218 1	414	0	40 6	556	00	244.4	1 061	Million Dollars	
00			371 8	00		000														1 21565		9 02 00	GWM	2024		2540.4	00	50437 8	700			00	00	0	00				0	00	8701 4	1226 2	00	11774	1700 7			2550 6	GWA	2001
0.0		00	30 7	00										7767	1.8.1	0.0	1201	1	0	1450.3	0.0	304.6	Million Dollars	i		102 0	0.0	2016.0		52.5	00	0.0	0	0				00	00	00	224 9	47.5	00	48.5	50			142.7	Million Dollars	
0.0	151	00	350 4	0										• 10/ •	3400 4	0.0	14004	1708 2	00	37001 0	00	C CM95	GWA	2025		2459.2	00			0 878	00	0	0	00			0.0	00	00	00	8701.4		00	11127			00	2734 6	QVM	Ĩ
00	10	00	205										6.41	205 4	124.7	00	111.7	67.7	00	1551.3	0.0	320.0	Million Dollars			195				51.9	00	0	00				00	00	00	00	228 8				00	9 666	00	154.1	Million Dotters	_
																										2781.1	CONTROL		0.0	657.0	00	00					00	0.0	0	00	8701 4		144/4	1708 2	00	12020	00	20105	014	2010
								-																						51.3	00		2			0	0.0	0	0								00	1050	Million Dollary	



Figure B.1 - Investment in Sustainables Capacity (MW) - all scenarios



Figure B.2 - Generation by Sustainables (GWh) - all scenarios

128



Sys. Benefits Charge
 Environmental Adders

*Business as Usual •Sustain. Port. Stand.

Figure B.3 - Peak wholesale electricity rates (\$1995/kWh) - all scenarios



Figure B.4 - Baseload wholesale electricity rates (\$1995/kWh) - all scenarios

Business as Usual Sustain. Port. Stand. Sys. Benefits Charge Environmental Addets **B.C. Baseload Wholesale Spot Price**



Figure B.5 - Total cost of electricity (million \$1995) - all scenarios

Total Cumulative Capacity from New and Existing Technologies - Alberta Model - BAU Share of market for non-least cost technol 20% Price threshold over least cost for invest 20% MegaWatts **Technology Name** 1996 1997 1990 1999 2000 2001 2002 2004 2003 2005 2006 2007 2008 2009 2010 1525.04 11.39 157 00 301.46 442 10 579.05 718 84 852.70 986 69 1118.85 1249.24 1387.99 Simple cycle gas 1000.44 1794.23 1928.45 Simple cycle oil (b) 0.00 -0.00 0 00 0.00 0 00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 Combined cycle gas 1.09 12 58 204.62 401.41 597.98 798.71 993.67 1188.87 1382.30 1574.18 1779.25 1982.75 2184.67 2385.05 2583.92 Combined cycle oil 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 Small hydro 13.58 194.83 200.00 200.00 200.00 200.00 200.00 200.00 200.00 200.00 200.00 200.00 200.00 200.00 200.00 Biomass steam 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 Coal - Pressurized Fluidi 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 Biomass gasifier turbine 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0 00 0.00 0.00 0.00 0.00 0.00 26.05 50.44 74.54 146.17 192.94 335.46 Gas cogen retrofits 1.88 98.35 122.40 169.68 215.98 240.52 264.84 268.94 312.81 0.00 0.00 0.00 Large Hydro 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 Coal gasifier turbine 0.00 0.00 0.00 0.00 0.00 0 00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 Waste fuels cogen 0.00 0.00 0.00 0.00 0.00 Wind Generators 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0 00 0.00 0.00 0.00 Fuel Cell Cogenerator 0.00 0 00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 Tidal Power 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 000 0.00 0.00 0.00 0.00 0.00 0.00 Parabolic-Trough Solar-0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0 00 0.00 0.00 0.00 0.00 0.00 0.00 Soler PV 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 Natural Gas 1712.78 1672.53 1633 23 1520.77 1485.03 1450.13 1418.05 1382.78 1318.55 1227.76 ng 1594.85 1557.37 1350 28 1287.56 1257.31 Coel 5850.21 5712.73 5578.48 5447.30 5319.37 5194 37 5072.30 4953.10 4836.70 4723.04 4294.48 4193.58 -4612.05 4503.67 4397.83 18.19 Oil 21.48 20 98 20.49 20.00 19.53 19.07 17.34 15.40 18.63 17.76 16.94 16.54 10.15 15.77 Hydro 814.77 806.62 798 58 790.57 774.64 759.42 751.82 744.31 782 66 767.09 738.88 729.49 722.20 714.98 707.83 Nuclear 0.00 0.00 0.00 0.00 0.00 0 00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 188.46 184.04 179.71 175.49 171.38 167.34 163.40 159.58 152.15 148.58 145.09 138.35 135.10 Biomass 155.81 141.68 Total Supply 8.615.64 0.787.65 8.967.00 9.146.34 9.325.68 9.512.34 9.698.99 9.005.65 10,072.31 10.258.97 10,472.47 10,685.97 10,899.47 11,112.97 11,326.47 Total Sustainables 13.58 194.83 200.00 200.00 200.00 200.00 200.00 200.00 200.00 200.00 200.00 200.00 200.00 200.00 200.00 Technology Name 2011 2012 2013 2014 2015 2016 2017 2018 2019 2020 2021 2022 2023 2024 2025 Simple cycle gas 2062.38 2198 89 2338.03 2473.87 2612.48 2751.88 2892.18 3033.42 3175.68 3319.00 3462.15 3605.20 3748.20 3891.20 4034.27 Simple cycle oil (b) 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 65 0.00 0.00 0.00 0.00 4755.17 Combined cycle gas 2788.23 2993.25 3199.09 3405 82 3613.55 3822.35 4277.65 4515 60 4041.23 4997.79 5243.54 5492.54 5744.90 8000.74 Combined cycle oil 0.00 0.00 0.00 0.00 0.00 0 00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 Small hydro 200.00 200.00 200.00 200.00 200.00 200 00 200.00 200.00 200.00 200.00 200.00 200.00 200.00 200.00 200.00 Biomass steam 0.00 0.00 0.00 000 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 Cost - Pressurized Fluidi 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 Biomass gasifier turbine 0.00 0.00 0.00 0.00 0.00 0 00 0 00 0.00 0.00 D.00 0.00 0.00 0.00 0.00 0.00 Gas cogen retrofits 360.76 385 15 409 65 434 26 459 00 483 88 500.00 500.00 500.00 500.00 500.00 500.00 500.00 500.00 500.00 Large Hydro 0.00 0.00 0.00 0 00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0 00 0.00 0.00 Coal gasifier turbine 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 Waste fuels cogen 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0 00 0.00 0.00 Wind Generators 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0 00 0.00 0.00 0.00 0.00 Fuel Cell Cogenerator 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0 00 0.00 0.00 0.00 Tidal Power 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 Parabolic-Trough Solar 0.00 0.00 0.00 0.00 0.00 0.00 0 00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 Solar PV 0.00 0.00 000 0.00 0 0.00 0.00 Natural Gas 1198.91 1170 73 1143 22 1116 35 1090.12 1064 50 1039 49 1015.08 987.91 945 17 ng 091 21 922.95 901.27 880.09 859.40 4095.01 3996 78 3904.61 Coal 3813 05 3723.44 3635.94 3550.49 3487.08 3385.58 3306.02 3152.46 2935.40 43 3228.33 3078.38 3008.04 Oil 15.04 14 68 14.34 14.00 13 87 13 35 13.04 12.73 12 43 12 14 11.85 11.58 11.30 11.04 10.78 693 74 Hydro 700.75 666.60 679.94 673 14 659 74 653 14 633 75 666.41 646,61 640 15 627.41 621.13 614.92 608.77 Nuclear 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0 00 0.00 Biomass 131.92 128.82 119.95 117.13 125.79 122.84 114.38 111.69 109.07 108.50 104.00 101.58 99.17 96.84 94.56 **Total Supply** 11.553.00 ,784.06 12,019.74 12,260.13 12,505,33 12,755.44 13,010.55 13,270,76 13,536,17 13,806,90 14,083.03 14.364.69 14.451.99 4.945.03 15.243.93 **Total Sustainables** 200.00 200.00 200.00 200.00 200.00 200.00 200.00 200.00 200.00 200.00 200.00 200.00 200.00 200.00 200 00

BAU t type technology ą (MM) pacity BO generation Total . <u>:</u> Table

			na siontating	n heid oot beb	ndogini. 2014	(1914) C - Lanouti et	r 3 hreiheid enschauts	ant for investors	11	•					
Technology Nume	1990	111)	198	198	Ĩ	7991	2002	C162	Xet	2005	2005	2007	2000	200	1 B
Sinche syde gas	11.30	197.00	301.40	442.10	579.05	710.84	62.70		11111	120.2	1307.90	1525.04	1000.44	1744.23	
Simple cycle of (b)	8	18			2.9			8			1471 M	1443.07	1007.01		
Contened system (11	88					8	8	8		8	8	8	8	8	
	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Í.	88	200 00	200.00	200	200.00	200.00	200.00	200.90	200.00	200.00	200.00	200.00	
	8	8	8	8	0.00	0.00	8	8	8	0.00	8		8	8	
Coal - Pressurised Fluidbood Bed	8	8	000	8	8	8	080	0.00		8	18				
Blamste gauller turbine	18 73		4.5	70.10							240 52			512.91	
Gas oogen retrolls) - 3 1	3 9	, 8 3	32	- 4 - 5	10.00	- B	88	81	81	81		0.00	98	
	38	2	88	88	88	B	8	8		8	8	8	8	8	
	8	8	8	0	8	0	0.00	900	8	0.98	8	0.00	0.00	0.00	
Wind Generators	8	0.00		8	8	0.03	0.00	000	8	8	0.98	0.90		88	
Find Call Cognition	0.00	0.00	000	8	8	8		28	88		8	28	88		
Tidal Power	88	88	88	2	88	88	8	88	8	ŝ	8	8	8	8	
	88	8	8	8	8	8	8	8	0.00	08	0.00	0.00	8	0.00	
	0.00	88	0.00	990	990	9,00	0.00	0.00	0.00	0.00	9,00	0,00		0.00	
Natural Geo	1712.70	1072 53	100.23		1957.37	1520.77		1430.13		1,002,11	1330.20			42044	
					10 51				17.70	17.1	1	1	10.15	15.77	
	14.77		ž	790 57	712.00	774.W	707.00	754.42	751.02	744 31	77M.W	729.49	722.70	714 90	
Nuclear	8	9.98	0.00	0.00	0.00		0.00	8		8	8				
					T THE P		1717.04	11 You 1	10,000,05	10,277,01	10,401.11	14,794,01	HI, MIL 11	11,131.01	- 1
	20.02	214.56	347.15	279.10	311.70	344 98	370 M	413 42	44.02	4144	521.05	5440.32	10.01	61 W	1
Tanàn atanàn Manan	¥	ž	ž	H I	2445	2416	1417	ä	đ	Kik		7172	CIN	2424	
Single cycle get	2002.30	2149.44	CC 9CCC	2472 87	2012.48	2751.00	2002.10	2010 20	3175.00	3318.00	3442.15	202.500	3749.20	OC 1995	
Simples cycles off (b)	0.98		8	100	0.98										
Combined cycle get	2202.30					81		8	8	0.00	0.0	000	8	0.00	
	200 00	200.00	200.00	200	20 8	200 00	200.00	200.00	300.00	200.00	200.00	200.00	200.00	200.00	
Skyrne to stream	11.40	65 .32	1	130.10	102.20	220 11	270.74	370.10	312,44		457.47	500.00			
Cost - Pressuriesd Fuldered Bed	8	8	18	888	88	58						88	5		
Conversion and a second	340.76			44.8	454 98		500.00	500.00	500.00	500.00	500.00	500.00	500 00	500 00	
Large Hydro	8	28	8	8		8	8	000	0.00	8	8	8	8	200	
Coal gauller turbine	8	8	88	8	88	38	88	38	88	88	8	88	88	88	
Whate fuels cogen	38	88	88		88	88	88	81	81	8	8	5 82	52 84	100.00	
Fuel Cell Cogenerator	0.0	8	8	8	8	8	0.00	8	8	0.00	0.00	0.00	8	8	
Tidad Power	0.00	90	8	0.00	000	8	0.00	8	8			88	88	28	
Parabolic Trough Solar-Power Plant	000	8	8		88	28	38	38	38	88			8	88	
SOME PV	88	88	B	88	88	88	8	8	98	0.00	0,00	0.00	9.90	000	1
Nutural Gen	1100 01	1122	22 Chil	1110 25	1000.12	1044.50	1039.41	1015.06	W1.21	947.01	MS.17	822.95	801.27		
Com	4095 01	1000 11	3404 81	30.0116	3723.44	JACK BA	3550 48	3407.00	3306 54	1304.02	3220.33	3152 48	1070.30	3004.04	
8	15 04	14.00	I N	18	109	13.35	10 61	6721	12.43	12.14	3 ::	11.30			
Hydra	700 75	980 74							8			00	08	08	
Nones	131.92		12.79	122 H	119.95	61.71	114.30	111.00	109.07	104.50	104.00	101.56	99,17	9.94	1
Total Supply	11,571.64	11,002.70	12,000,00	12,211,21	12,523.07	12,774.00	13,028.10	13,299.40	13,944 11						
Tela Sudanaho	110 40	100.32		1.10	A 1 4	0.00.11	1.916		1444						ł

Table C.2 - Total generation capacity (MW) by technology type - SPS

Total Cumulative Cap	pacity from	n New ai	nd Existin	g Technolo	gles - Alb	erta Model -	ŠBC								
	Non-Mintle	S	hace of marinel fo	<u>c non-local cost le</u>	ahnalogies: 20	ni Edia	en threebold over least	cost for investm	ect: 2	0%					
Technology Name	1996	1957	1996	1000	2000	2001	2002	2003	2994	2005	2000	2007	2005	2000	2010
Simple cycle gas	11.30	157.00	301.48	442.10	579.05	718.84	652.70	995.00	1110.05	1248.24	1307.98	1525.04	1000.44	1784.23	1028.45
Simple cycle of (b)	0.00	0.00	0.00	0.00	0 00	0.00	0.00	0 00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
Combined cycle gas	0.00	0.00	102.52	299.31	445.97	994.82	001.50	1005.78	1280.28	1449.54	1054.03	1050.13	2000.05	2260.43	2450.30
Combined cycle of	0.00	0 00	0.00	0.00	0 00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
Smell hydro	13.50	194.93	200.00	200.00	200.00	200.00	200.00	200.00	200.00	200.00	200.00	200.00	200.00	200.00	200.00
Cost - Press aland Ebidined Red	0.00	0.00	0.00	6.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
Biomese gesiler turbine	33,21	33,21	33.21	33 21	33,21	33 21	33,21	33 21	33,21	34 92	30.02	36.92	30.92	38.82	30.92
Ges cogen retroits	1.00	28 05	50.44	74.54	98 36	122.40	140.17	100 00	102.84	215.06	240.52	264.64	200.04	312.01	338.48
Larga Hydro	0.00	0.00	0.00	0.00	9.00	0.00	0.00	0 00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
Coal gasiller turbine	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0 00	0.00	0.00	0.00	0.00	0.00	0.00
Weste Tuele cogen	17.50	17.50	17.58	17.50	17.50	17.50	17.50	17.50	17.50	20 00	20.00	20.00	20.00	20.00	20.00
Fund Call Composition	47.02	97.02	97.02	97.02	47.02	47.02	47.02	47.02	47.02	0.00	38.14	30.14	39.14	0.00	30.14
Tidal Preset	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	8.00	0.00	0.00	0.00
Parabolic-Trough Solar-Power Plant	0.00	0 00	0.00	0.00	0 00	0 00	0.00	0.00	0.00	0 00	0.00	0.00	0.00	0.00	0.00
Solar PV	4 20	4.28	4 28	4 20	4 20	4 20	4.20	4 28	4.28	5.96	5.90	5.96	5.90	5.96	5.86
	0.00	0.00	0.00	0.00	0.00	0.00		0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
Natural Ges	1712.78	1072 53	1633.23	1584.85	1557.37	1520.77	1405.03	1450.13	1418 05	1302.78	1350 28	1318.55	1207.58	1257.31	1227.76
Coal	5850.21	5712.73	5576 48	5447.30	5319 37	5184 37	5072.30	4953.10	4836.70	4723.04	4612.05	4503 67	4367.83	4294.40	4193.56
	21.40	20.00	20 48	20.00	19 53	18 07 774 84	10 03	760.43	17.78	17.34	754.04	10.54	18.15	15.77	10.40
hydro	0.00	0.00	199.39	/00.5/	100	7/4 (84	/1/ 00	738 42	131.02	0.00	1.34 04	1,00	144.20	6.00	/0/.03
Ricman	100.44	184 04	179 71	175 48	171 36	187.34	103.40	159 58	155 81	152 15	148 58	145.00	141 00	138.35	135 10
Total Bupphy	8,718.64	8,878.00	8,907.00	8,148.34	8,325.68	8,512.34	8,008.00	8,685.65	10,072.31	10,258.87	10,472.47	16,685.97	18,008.47	11,112.97	11,326.47
Total Bustainables	115.00	296 93	302.09	302.00	302.09	302.09	302.09	302.00	302.09	324.02	324.02	324.62	324.02	324.62	324.02
Technology Nume	2011	2012	2013	2014	2015	2016	2017	2016	2018	2626	2021	2622	2023	2624	2025
Simple cycle ges	2082.30	2100 00	2336 03	2473 07	2012 48	2751.00	2002.18	3033 42	3175 68	3318.00	3482.15	3005,20	3748.20	3001.20	4034.27
Simple cycle of (b)	0.00	0 00	0 00	0.00	0 00	0.00	0 00	0 00	0.00	0.00	0 00	0.00	0.00	0.00	0.00
Combined cycle ges	2063.61	2000 63	3074.47	3261 20	3464 03	3672 83	3001.71	4120.13	4306.08	4005 05	4848 27	5004.02	5343 02	5505.38	5051.22
Combined cycle of	0.00	0.00	0 00	0.00	0.00	0 00	0.00	0 00	0 00	0 00	0.00	0.00	0.00	0.00	0.00
Small hydro	200.00	200.00	200.00	200.00	200 00	200 00	200.00	200 00	200 00	200.00	200.00	200.00	200.00	200.00	200.00
Signal - Descripted Daffred Red	000	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
Loss + Proverses Plasses and	30.00	10.00	10.00	10.00	44.01	44.81	44.81	44 91	44.91	44.61	44.01	44.81	44.81	44 81	44 81
Ges coor retrolts	300.70	385.15	409 65	434.20	458.00	483.00	500 00	500 00	500 00	500.00	500.00	500.00	500.00	500.00	500.00
Large Hydro	0.00	0 00	0 00	0 00	0 00	0.00	0 00	0 00	0.00	0 00	0.00	0.00	0.00	0.00	0.00
Coal gasiller turbive	0.00	0 00	0 00	0 00	0.00	0.00	0.00	0 00	6.00	0 00	0.00	0.00	0.00	0.00	0.00
Weste fuels oogen	20.60	20 00	20 60	20 80	23.70	23 78	23.70	23 78	23 70	23.78	23.70	23.70	23.70	23.78	23.70
Wind Generators	59.14	58.14	58.14	50 14	72.70	72 7	72.70	72.70	777	72.70	72.7	72.70	72.70	72.78	72.7
Fuel Cell Cogenerator	0.00	0.00	000	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
Paraballo, Trauch Solar, Pourse Diant		0.00	0.00	0.00	800	000	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
Solar PV	5 80	5 94	5 96	5 94	0.04	0 04	8 04	8 04	8 04	8 04	04	8.04	9.04	6.04	8.04
	0.00	0 00	0 00	0 00	0 00	0 00	0 00	0 00	0 00	0.00	0.00	0.00	0.00	0.00	0.00
Natural Ges	1100.01	1170 73	1143 22	1118 35	1090 12	1064 50	1038 49	1015 08	991.21	907.91	945.17	922.95	901.27	880.08	850.40
Coef	4005 01	3000 79	3804 81	3813.05	3723 44	3635 84	3550 49	3487.08	3385.58	3308.02	3220.33	3152 48	3078.38	3008.04	2935.40
	15 04	14 88	14 34	14 00	13 87	13 35	13 04	12.73	12 43	12.14	11.05	11.50	11.30	11.04	10.70
No. of Contract of	1 """	0.00		0.00	0.00	0.00	0.00	14	0.00	040.15	100 10	#21 41 0 M	#21.13 0.00	014.92	WUB.71
Rivmett	131.02	128.92	125.70	122 84	119.05	117 13	114 14	111.60	109.07	108.50	104.00	101 64	0.00	0.00	0.00
Total Busely	11.553.00	11,764.06	12.010.74	12,200.13	12,545.33	12,755.44	13,010.55	13,279,76	13,536,17	13,006.90	14,003,03	14,364,66	14,051.00	14.045.02	15,243.03
Total Bustainables	324 82	324 82	324 62	324 82	348 52	349 52	349 52	349 52	349 52	349.52	349.52	349.52	349 52	349.52	349.52

otal Cumulative Capa	city from	New an	d Existing	Technologi	es - Alberu	ugrafie vrz - reconu	raincial over land, con	the burntment:	2016						
F							7462	246	2004	2005	7996	2007		100172	É,
Anatom Hund	Í	1007	1946	1		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	111.37	704.17	105 42	100.75	1002.3			8	
	7.01	110.12	212.04	313.45	412.42	81	8	0.00	0.00	0.98				2318.00	¥
	8	8	8	18		47648	705 77	ED2 14	1157.20	CP BLEL		B	0 80	8	
mbined syste gas	2.19	20.14		3	8	8	8	000		8		200 000	200.00	200,00	×
mbined cycle of	0.00	8			200.00	200 00	200 00	200 00		138.67		150.01	173.34	(07.00	
				4.72	12	134	17,70			8	8	0.00	8	8	
	1.12		3	8	8	8	8			124.57	E u	199.91	171.30	1.1	
al - Pressurined Fluidand Bed		5	838	4.72	50 92	77.44	07.70		8	00 00	500.00	500.00	500.00	5000	
succe berger ground	1 12		225 M	429 10	500.00			8	8	0.00	0.08		88	88	
a codimi tantoura	8	8	0.98	8	8	38	81	8	8	8	0.98		88	8	
	8	8	0.00	8		3	8	8	8	00		38	81	8	
	000	8	0 98	38	38	81	8	0.98	0.00	8	88	88	8	8	
and Conversions	0.00	8	8	38	88	8	0.00	8	0.00	88	38	B	8	8	
al Call Cogeneration	000	8	8	88	8	8	8	8	88	88	88	8	8	8	
the Power		88		81	8	0.00	8	8	38	88	00	0.00	0.08	8	
mbole Trough Salar-Power Plant			88	ĝ	8	08	8	38		8	0.00	0.00	0.00		
12	B	88	0.90	0.00	6 90	000		1450 13	1410.05	1002.78	82.05C1	1310.55	1207.00		• :
	1712.70	1072.53	1600.20	SE VES	1307.37	1040.71		4053.10	4096.70	4723.04	4012.00			15.77	
	5450-21	5712.73	5578.44	5447.30		1.07	118	10.10	17.74	17.34			12.3	714.00	
	21.44	20 80	2			774. M	767.08	750.42	751.12		81	8	8	8	
-	014.77			8	8	8	000			157.15		145.00	141.99	K KI	
Licker			120.21	175.40	K 111	H. 191		04.041	10.0/2.1	10,254,07	10,472.47	10,005,07		11,112 V	į
Heure H	LUILN	0,707.00	1,947.00	E. 144.34	N 125'	1.2 IC	175.40	103 12	431.53	450.15	440.02	517.01	21.000	10.010	
	10.72	231.20	240.52	14. REC	20 810							X		N 24	
			ł		Ĩ	i H	2417	1010	2MK		K KK	2121.70	1770.23	2000.15	
Technicky Hame	146	141			1007 M	2005 40	2100 35	2212.30	Z210 02		0.90	1.90	8	8	
Subst cher Bas				8	ĝ	1.90				4850.01	5104.14	5440 14	5002.01	5047.12	
Emple syste of (b)				3491.00	16 1646	101210		8	080	8	08	0.00	18		
	8		8		88		200 00	200 00	200.00	200 00	200.00	200.00			
Concerne open on	200 00	200.00	200 00	200.00		2001	201	32044	5.8	10 000			900	0.98	
	210 40	60 ICZ	245.77	83	8	000	000	8				170.70	100.54	386.47	
Coal - Pressureed Fuddand Bed			245 76	200 50	275 40	51 DHC	205.74			000	500.00	500.00	000.00	500,000	
Monate geniler turbine			500 80	500.00	500 50	500.00		8	88	8	0.00	8	8	88	
Cles cogen intrust	8	8	0.00	8			8	8	0.00	000	18			5.	
Cool cooling turbing	8	8	8		8	81	8	0.00	8	8		8	- 1	1	
Nielle fuele cogen	8		88	88	8	8	8		38	88	81	0.00	8	8	
Wind Generators			81	8	8	0.00	8		88	0	8	8	8		
Fuel Call Cogenerater	88	82	8	0.00	0.00	8	38	88	8	8	0.00	8	8	B	
Tidal Power	8		8	000		38	88	8	0.00	0.00	0.90		8	8	
	8	0.0	8	88	88	8	980	0.00	0.00	0.00	2000	072 85	101.27	100.00	
	000	0.0		1116	1000 12	CE 1001	1039-48	1015 06				3152.44	3078.36	3008.04	
Natural Geo	1198 01			3013.05	3723 44	No SCOK	1550 42	3447.08	1243	12.14	11.85	11.54	11.30	1.0	
Case			1	14 98	13 67	13 25			Ma	840.15	603.75	027.41	C1.12		
	700 75			1160			8	8	0.00	0.00	0.90				
	8	00	8			117.13	114.3	111.00	100.07	100 50		NAK M	1481.8	14,945,13	=
Biome H	20-161		12012/1	12,200,13	12,995,33	12,793.44	12,012,55	12,270.70	11.00011	801 SK	M0 51	979.38	1010.54	1056.13	
Lora article				721.12	750 80	18.8	010.07								

Table C.4 - Total generation capacity (MW) by technology type - ADD

Total (GWh,	Blomass	Nuclear	Hydro	2		Natiral Gas			Solar PV	Parabolic-Tro	Tidal Power	Fuel Cell Cog	Wind Genera	Waste fuels o	Coal gasifier	Large Hydro	Gars cogen re	Biomass gas	Coal - Pressu	Biomass stea	Small hydro	Combined cy	Combined sy	Simple cycle	!		Total (GWh, I	Biomass	Nuclear			Natural Gas	N'N	NA	NA	Solar PV	Tidal Power	Fuel Cell Cog	Wind Generat	Weste fuels o	Coal gasifier t	Lanse Hydro	Gas coon ret	Codi - Pressu	Biomass stea	Small hydro	Combined cyc	Combined cyc	Simple cycle o	Simple cycle o		Generat	
Millions)							_			ugh Solar		enerator		ŏ	untine		TON	fier turbin	rized Flut	3							Aillions)								_			enerator .	3	ogen	urbine			iar turbin			6 9.		9			ing Sta	
58,822	40.9	00	3683 1	4.7	30491 5	454 5			00		00		00	0	0.0	00	2417.6	0.0	0.0	0.0	1138.8	0.0	16436.8	3/34.2	GWD	2011	43,104	59.9	0.0	43257	3/3/1.1	1134.5	00	0.0	0	0.0		00	00	0.0	0.0	0.0	0.0		00	0.0	0.0	0.0	000	0.0	awh	tistics -	
\$ 1,665	2.8	00	100 2	03	829 2	28.0			000			0.0	00		0	00	057	00	0.0	0.0	31.0	00	475.5	132.3	¢ uonim		3	21	0.0	5	1.701	33.0	0.0	0.0	0.0	00		0.0	000	00	00	0	0.0	000		0.0	0.0	00	0	0.0	Million S	Alberta	
60,000	40.0	00	3646 3	40	20774.8	427.8					200		200	00			2581.1	00	0.0	00	1138.8	00	18401.5	0.0		2012	43,850	58.5	0.0	4282.4	3/084.0	1503.4	00	00	00	00			200	0.0	00	0.0	12.5	0.0		774	00	21	0.0	21.2	GWh	I Model	
\$ 1,702	2	0	8	0	810	27	2 9										70	0	0	00	31.0	00	518.6				8 103	2.2	0			44.0	00	0.0	00	8			200	0.0	0.0	0.0	0.2	000		1.5	0.0	0.1	00	0.0	Million S	- BAU	
61,2	3	<u> </u>	3 380	<u>a</u>	2907	8											274		~	<u> </u>	1136		19970			2 8	44,73	57	~	4230		171430			~ _	-							174	0.0		1100		28	0	ž	GW .	ŝ	
2 5 1	9.0	0	9.8		52	0	5	5 6	32	52	52	52	52	ä		56	2	ŏ	ă	ö	5	5		ă : -				1	õ	āi			ā	ā	õ	õ	ōċ	5	ō	ā	õ	õ	•	0	5 6		ō	0	0	•		Annual G	
,740	28	0	9 8 5	03	703 1	267			2	2				0			14.9	00	00	0.0	31.1	0	62 3	000		\$ •	727	2.3	00	64.5		3 5	0	0	00	00				0.0	8	00	3.5	000		22.1	0	0.8	00	2	<u>,</u>	Wh prod	
62,428	38.1	00	3573.7	43	26391.9	374.0						8					2910.2	00	0.0	00	1138.8	00	21542.2	00	1127	2014	45,636	55.0	00	4197.2		37484 3		00	00	0			38	0	00	0.0	336.0	000		1138.8	00	515.4	0.0	000.3	GWh	uction and R	
1,794	2.8	00	98.5	0.0	782.2	26.1	00			2 2							8	0.0	0.0	0.0	31.4	00	612.1	0.0	1000			2.5	0.0	80.3		745 1		0.0	0.0	0.0				00	00	0.0	7.0	0		23.4	0.0	E.	0.0	17.5	Million \$	rvenue Rec	
63,678	37.2	0.0	3538.0	4 2	27724.7	348.4	00										3076 0	0.0	0.0	0.0	1138.0	0.0	23112.5	00		2016	100,04	54.4	00	4155 2	62	17411		00	0.0	00	88		38	88	00	00	499.5	8	88	1136 8	00	1187.9	0.0	885.7	GWh	alved for Elec	
8 1,834	28	00	97 8	03	765 0	3	8	8					38						0	00	31.4	0.0	050.4	00		Millon S	000,1	2.5	0.0	80	03	1255		0.0	00	8	000				00	00	10.8	0.0	000	24.7	200	330	0.0	26.3	Million \$	ctricky (millio	
CG'H	38		3502		27073.	330			2 9	2 9	2 5	2 2	2 9		2 9		3242.0				1138.0	20	24660.1	8	1979		001 14	53.2	0.0	4113.7		37223.6		0.0	0.0	0.0	0.0		0.0		0.0	0.0	050.1	00	00	0.0	0.0	2034.5	00	1144.4	GWh	3 5 2000	
1,0			5	-	~	2															4	-	70	_	171	Millio			_		_	Ĕ.			_	~	_		~ ~			_	=	_	~ ^	- 9	*~	. 8		8	Millon		
101	27	0	6.8	0.3	8.3	50	0	8	8	30									0.0	0	in	0.0	<u>=</u>	0	Ĩ	<u>.</u>	ł	1	0	1	33		2	10	6	0.0	8	50	00	50	5	5	5	5	<u>.</u>	5 6			ă	52		-	
1 10,00	355	00	3467.6	40	26437.0	322.5	00	00		00		3		33	200	5	2000.7	100	0.0	0.0	1138.8	0.0	26335.9	00	5162.1	GWA		51.9	8	40725	5.9	30012.3		38	00	00	00	0	88		00	00	820.2	00	88	1130.0	0.0	3019.8	0.0	1404.5	GWh	2001	
1,011	2.7		80	03	731.9	24.8	0	0		00								3		0.0	31.5	00	747.9	0.0	189.3	Million S		2.0	00	93.9	03	661.0		000	0	0.0	00		000		000	0.0	18.0	0.0	000			02.8	0.0	44.0	Million \$	_	
700110			3432.8	3	25815.7	314.9	0.0	8					0	2 2						0.0	1138.8	0.0	28106.1	8	5383.9	GWh		1.00	00	4031.8	5.8	36494.4			0.0	0.0	0.0	00	0.0		0.0	0.0	979.5	0.0	000	11.00.0		4109.7	0.0	1649.7	GWh	2002	

Table C.5 - Annual energy generation (GWh) by technology type - BAU
		243	77,647 8	2,385	76,123 \$	2,301	74,628 1	2,220	73,162	2,141	1 21'11	een'z	. 115'01		ane la c	
		14.2	410.7	2.6	30.0	2.6	30.8	27	31.5	2/			10.00	3005	68 930	
		00	0.0	00	00	00	0.0	0.0	00				300	2 0 0		2.7
		97.1	3100.7	9 8.6	3232.0	80	3264 7	05 5	3297.7	94.9	3.91.0		0 0000		000	00
-		0.3	3	0.3	34	03	35	6 0	36	203	37		2201			5.0
		003.4	21857.0	663 .8	22383 0	874 2	22921.6	679 6	23473 2	085 1	24036.1	7 000	24010 0	100.2	10	
		23.8	206.6	23.0	273 0	24 1	279.6	24 2	286 3	24 3	293 2	24 4	5 000	240	20000	7150
_				0	00	0	00	0.0	00	00	0.0	00	00		100	200
			0.0		00	0	00	00	00	00	00	00	00	00	00	
			5			00	00	00	00	00	00	00	00	00	0.0	0.0
		0.0	200					00	00	00	00	00	00	0.0	0.0	00
			00					000	00	00	00	00	00	00	00	00
		00	0.0	80			5	0		00	00	00	00	00	00	00
		00	0.0						00	00	00	00	00	00	00	00
		0.0	00						0	00	0	00	00	00	00	00
		0.0						0	0	00	00	00	00	00	00	00
							00	0.0	00	00	00	00	00	00	00	00
		101.7	3350.7	2	3330.1	2		00	00	00	00	00	00	00	00	00
_					7 005		3350 7	97 0	3350 7	8 5 5	3350 7	940	3350 7	93 1	3350 7	0 20
							00	00	0	00	00	00	00	00	00	00
							00	00	00	00	00	00	00	00	0.0	0.0
			1120.0	د د د د			00	00	00	00	00	00	00	00	00	00
					1134 8	33.5	1138.6	330	1138 8	32.5	1138.8	32 0	1138 8	31.0	1138.8	31.0
		1262.0	1.0000	1.7011	0 D	0.0	0.0	00	00	00	00	00	00	0.0	00	0.0
_		0.0				11117	37128.5	1042 0	35292.6	974.7	33475.0	8 BOB	31875.2	849 3	29685.9	798.4
		273.2	0501.4	2/1.7		0.0	0.0	000	00	0.0	00	00	00	0.0	0.0	00
		S troilin	GWh	C UCNIIIN	U ALLA	6 1041114	ACOA	2445	8 78C9	231.5	0002.0	218.0	5833 7	206.6	5007.4	198.9
		,	2025		2024			Million S	GWh	Million S	GWh	Million \$	GWh	Million \$	GWh	Million S
\$ 1,629	57,667	1,575	96,979	120,1	4 nev'bo	1,400	1		200		2021		2020		2019	
2.8	41.9	2.0	42.9	2.0			1		612.63	1.383	52.224	1,325	51,272 \$	9 1,279	80,321	\$ 1,225
0.0	0.0	0.0	0.0	00	0.0	200	100			30	47.2	2.0	48.3	2.9	49.5	2.0
101.0	3720.3	100.0	3757.0	100.2	3795.0	80.8	3634.2		30/30		0.0	000	00	00	00	0
0.3	4.0	0.3	4.0	0.3	5.0	0.3	5.1	20.3		30	1017 A	9.5	3051 6	200	3001.5	8
847.9	31225.3	856.0	31976.7	864.5	32730.7	872.0	33403.2	2010	1.14140			50.0	55	0.3	5.0	E.O.
26.7	481.8	29.4	509.0	30.6	550.7	32.0	500.8		7 860		24700 7	878 4	35400 7	879.9	35991.1	872.1
0.0	0.0	0.0	0.0	0.0	0.0	00	0.0				700 4		762 1	36.8	834.0	37.9
0.0	0.0	0.0	0.0	0.0	00	00	0.0							0.0	0	0
0.0	0.0	0.0	0.0	0.0	00	00	00							0.0		0.0
0.0	0.0	0.0	0.0	0.0	0.0	0.0	00	00	0.0					0.0	0	00
0.0	0.0	0.0	0.0	00	00	0.0	0.0	0.0							0	
00	00	0.0	00	8	0.0	0.0	0.0	0.0	00	00	00					
	0	00	00	0.0	00	0.0	0.0	0.0	00	00	0.0					
			00	00	00	0.0	0.0	0.0	00	0.0	00	0.0	200	0.0		
			0.0	0	0	00	0.0	0.0	0.0	0.0	0.0	00	00	0.0		
						00	0.0	0.0	0.0	0.0	00	0.0	00	0.0	0.0	
51.2	1.001	2 2			00	0	00	0.0	0.0	0.0	0.0	00	00	0.0	00	
0.0	. 0.0	0			1076.3		1774.8	41.4	1011.8	30.6	1447.2	31.0	1293 0	27.0	1137.1	23
0.0	0.0	0.0	0.0	0.0				00	000	0	0	0.0	0.0	0.0	0.0	0.0
0.0	0.0	00	0		0.0					0	0.0	00	0	0.0	00	00
30.9	1138.8	30.5	1130.0	30	1135.6	20.7	1138.0		1,20,0	0.0		00	00	0.0	00	00
0.0	0.0	0.0	0.0	00	00	0.0	0.0			, ,	1100	29.4	1138.8	27.6	1138.8	27.0
432.4	15275.8	384.6	13744.2	341.6	12202.9	200.0	10705.0	200.4	WZ41.1	212	1000.3		00	0.0	0.0	0
00	0.0	0.0	0.0	0.0	0.0	00	0.0	00						145.4	5281.4	113.0
123.4	3524.0	114.8	3307.9	105.0	3076.7	90.8	2835.3	87.9	2500.3	20	2302.1	202	2120.2	01.0		0
		Million S	GWh	Million S	GWh	Willion \$	GWh	Million \$	GWh	Million 1	GWh	S UDHIM	UVU	E LOURIN		102
	2010	_	2009		2008		2007		2006		2006		2004			
											•	_	-	_	-	

1	6	-	Ţ			Ļ	5	Ĩ	Ĩ	Ĩ					Ì	F	E	f	ĩ	Cen	ŝ	1	Î	Į	Γ			11000			ļ	2	Ē	Į	ł	Ţ	NY.	Solar			ŝ			1			ġ.	5	-	Bloma	Sil	1	Comb	Ĭ	Ĭ	1		00	2
ľ	•	_	The Gara			N	boto-Trough S	Power	Cel Cogenera	Generators	e lueve cogen						- Pressurined P		Ĩ	hand cycle of	And chose days			in cycle pin							-							2	oc reversion of	TOWN I	Con Cogeneration		and the second se			ł			Pressuried Pi			ned cycle of	ned cycle gen	even on (a)				Runn on	mailing
_	<u>.</u>	DACCZ	417	-											š				1130			COLOC		1024.0	Ś		ł				413			8.618	2	2											00	00		00						N. N.			
•		·~· 8	•	•				•					-				-		-				- -	10	The stand		-			_	i			•	-	-	-										•	0		-									
8		-	-		8					3 6			20	2			8	5					<u> </u>	5	ľ	i		Ē					<u> </u>	ĩ	ě	ĕ	ă	6	<u>.</u>	5	. c	5	5			ō	0	2	9									- 6	
00	Ĕ.	22542.7	391 2	0													0	1720 7					00	1 1202	k		Ĩ	1977.4			2024		Ĭ	1213	0	00	00		2	2	2	33	3	33		ò	3	:	0		12.5					ł			
																									1.111	F																																	
00			23 1	0												8	0	1	10				0	107 1		F				0	ŝ	2	128	5	0	9	0					2	2	2	8	00	2	2	0	: 3					2			_	
																																				-																							
0	Š.	21700	¥	2	2	2	22								7 0501	1830 2	8	1830 2	A BELL			2422M 7	00	2112							1224		10020 8	Ī	0	00							3	2	0	8	1200	Ī	00						į			į	
				-		-							-	-			-									F																															Ę		
	4		ž	0	0	_		2 9	2 4					-	ž	74	2	74				Ŧ							-	2	190	2	1370 7	55		2	: :					2	2		0	00	-		00	::					2:				
•	<u>.</u>				<u>e</u>	0.0		2.0	20		_				-	-			-		_	-	_		Ĩ	-		ĺ		-	-	_		-																						T		-	
	5	2005	×	_	_	_			_			_	_	_	22	Ē	_	Ī			_	Ì		5		2	2		Ē	_	ŧ		3510	1002						-					•		1510	22		. 5				1			2 :	ŧ	
8		2.0		00	0	8	66	34				5	8	¢	2	2	3	-			3	~	90			\$	ł		-	ö	2	Ξ	7	ă	ā	2						5		ĕ	ŏ	ă	ũ	ũ	ā	5			5 4				53		"
																										ļ																																	
•	Ē,		2	•	•	•		-		2		2	<u>e</u>	_	127	Ĩ	0	2			2	8							515	2	Ē	:	13474	¥				2				0	8	0	8	00	57.5	-				2:	2		0	157			
2	34			0	<u>e</u>	2			2		-	<u> </u>		~		_						-			Ī	-				-	-		_																									-	
•			2		•	•	•			•	•			•	DEC	ĩ	•	Ş		Ê.	•	27783				ŝ	ž		130	•	4195		Maat	1010						0		0	0	0	2	0	2000	5	-	. 2			_	2		5		Í	
ō	ō		• •	ō	ð	ō	0	0 (•	ō	•	0	0	•	-	•		•					•	•		∍ E	ā		1	•	~	~	•				•			0		•	•	•	P	Ģ											E	•	
	=	8			_	_	_		_	_	_	_	_	_	ĩ					•	_	11		. :					5	_	ź	_	1122	2						-		_			_		110				:	b .		2		2			
8				0	8	8	0			8			ä	<u> </u>		-					<u> </u>	-	2		-	2			2	ě	3		-							ā	0	ē	õ	õ	ö	ō						-		-	0	1	2	-	
	ž	į													ŭ	ž						3		1	2	۵	*		121		1		3324	2	5				_	_	_	_	_	_	_	_	5			. (_		_		9	¥	
0		-		0	ŝ	8	0			0	0	8	8	8	2	-					8	ï	: 2	3		≩	3		•	â		-	-				33	5	8	8	8	8	8	8	ä	ä	-			5			ŏ	ĩ	ð	=	§ : €	2	
	-														_				_	_		17	;										12	i	_																			~		_			
0	45.4			0	8	8	0	0	0 (0 (¢	0	\$	8	00	20.5	2				2	8	27.5				5		:		0		-			2		2		0	<u>0</u>	8	2	0	8	00						3	5	2	õ	2	3	5	_	
															ų	N	,	•	• •	_		H	2		2			1	L		*		Ħ														٤					=		<u>پ</u>			_		
00	167.0	-			0	8	0	0	0	0	00	0	0.0	00	No.	217 5					8				5	3	Ħ		140.0	0	5.21		2.0			3		8	2	0	8	8	8	8	0.0	0.0						Ĭ	0		8	20.4	3	Ĩ	
																										Į																																	
0	Ē	2			0	00	0	ŝ	8	0	0	00	0	0	Ĭ	1				5	0				Ĭ	5			-	0	2					3		0	8	8	8	8	8	8	0		1		;;		2	1	8	ĩ	8	3 2			
0	3432.8					0	8	8	ŝ	8	8	00	0	00	1000						8	A 7415			7037.0	3	2018		1219.7		-						ċ	0	8	8	8	8	0	0	0					8			8	5255.2	8	8	3	ž	
																										Ş																															Į		
K e	Ē	0	Į	5	g	2	2	2	2	2	8	0	0	2	140.1	Ē				2	8	1.1		8	ī.				47.5		1911	5			5	8		8	8	8	8	8	00	00	0.0			ŝţ		8	X	11	00	200	0.0	65.7	Dunn		

Table C.6 - Annual energy generation (GWh) by technology type - EA

138



Figure C.1 - Investment in Sustainables Capacity (MW) - all scenarios Capacity [MW]

Alberta - Market Penetration of Sustainables



Figure C.2 - Generation by Sustainables (GWh) - all scenarios



Figure C.3 - Peak wholesale electricity rates (\$1995/kWh) - all scenarios



Figure C.4 - Baseload wholesale electricity rates (\$1995/kWh) - all scenarios



Figure C.5 - Total cost of electricity (million \$1995) - all scenarios



TEST TARGET (QA-3)











O 1993, Applied Image, Inc., All Rights Reserved

